CONTROL OF RUBBER TREE PATHOGENS BY ENDOPHTIC FUNGI

SAKUNTALA SIRI-UDOM

DOCTOR OF PHILOSOPHY IN APPLIED MICROBIOLOGY

ลิ<mark>ปสิทธิ์มหาวิทยาลัยเชียงใหม่</mark> Copyright[©] by Chiang Mai University All rights reserved

> GRADUATE SCHOOL CHIANG MAI UNIVERSITY DECEMBER 2016

CONTROL OF RUBBER TREE PATHOGENS BY ENDOPHTIC FUNGI



THESIS SUBMITTED TO CHIANG MAI UNIVERSITY IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN APPLIED MICROBIOLOGY

GRADUATE SCHOOL, CHIANG MAI UNIVERSITY DECEMBER 2016

CONTROL OF RUBBER TREE PATHOGENS BY ENDOPHTIC FUNGI

SAKUNTALA SIRI-UDOM

THIS THESIS HAS BEEN APPROVED TO BE A PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN APPLIED MICROBIOLOGY

Examination Committee: Soplan Boealue...Chairman (Assoc. Prof. Dr. Sophon Boonlue) Saisaman Munyang...Member (Prof. Dr. Saisamorn Lumyong) Member (Asst. Prof. Dr. Yingmanee Tragoolpua) A Algebra piece

(Asst. Prof. Dr. Angsana Akarapisan)

ンゴル (Asst. Prof. Dr. Maslin Osathanunkul) Advisory Committee: Saisanon Lumyong Advisor (Prof. Dr. Saisamorn Lumyong)

(Asst. Prof. Dr. Yingmanee Tragoolpua) Boonson Bussaban.....Co-advisor

(Dr. Boonsom Bussaban)

28 December 2016

Copyright © by Chiang Mai University

ACKNOWLEDGEMENT

I would like to express my gratitude to all those who gave me the possibility to complete this thesis. I am deeply grateful to Prof. Dr. Saisamorn Lumyong for her superior supervision, worthy suggestions, patience and giving me an important goal for making this work. I am appreciate to Dr. Nakarin Suwannarach for giving me isolation technique of *Muscodor* and suggestions on molecular techniques; Dr. Boonsom Bussabun for worthy suggestions on phylogenetic tree assay; Dr. Nareeluk Nakaew for giving me the phytopathogen for this study; Dr. Rungroch Sungthong comments and English proofreading on manuscripts; Prof. Dr. Kozo Azano for his encouragement and suggestions on GC/MS assay.

I would like to acknowledge the financial support from the Graduate School of Chiang Mai University, The Higher Education Commission, Thailand, under the National Research University (A1) program, Chiang Mai University, Thailand Research Fund (RTA5580007), and Udon Thani Rajabhat University, Udon Thani, Thailand.

I would like to thank the facilities support from The Sustainable Development of Biological Resource Laboratory (SDBR), Department of Biology, Faculty of Science, Chaing Mai University and all colleagues in my laboratory are also thank for their kindly help and friendships.

Finally, I extend my warmest appreciation to my family for their encouragement, faith in me, endless patience and their support during my studies.

All rights reserved

Sakuntala Siri-Udom

หัวข้อดุษฎีนิพนธ์ ผู้เขียน ปริญญา คณะกรรมการที่ปรึกษา การควบคุมเชื้อก่อ โรคในยางพารา โดยราเอน โคไฟท์

นางสาวศกุนตลา ศิริอุคม

วิทยาศาสตรดุษฎีบัณฑิต (จุลชีววิทยาประยุกต์)

ศ. คร. สายสมร ลำยองอาจารย์ที่ปรึกษาหลักผศ. คร. ยิ่งมณี ตระกูลพัวอาจารย์ที่ปรึกษาร่วมอ. คร. บุญสม บุษบรรณ์อาจารย์ที่ปรึกษาร่วม

บทคัดย่อ

กัดแยกเชื้อราเอนโคไฟท์จำนวน 271 ไอโซเลตจากพืชสมุนไพรจำนวน 13 ชนิดและยางพารา (Hevea brasiliensis Müll.Arg.) ในภาคเหนือและภาคอีสานของประเทศไทย ราเอนโคไฟท์ส่วน ใหญ่คัดแยกได้มาจากพญายอ *(Clinacanthus nutans* (Burm.f.) Lindau, 24.4%), ทองพันชั่ง (Rhinacanthus nasutus (Linn.) Kurz, 10.3%) และลิ้นงูเห่า (C. siamensis Bremek, 8.9%) ตามลำดับ โดยพบราเอนโดไฟท์ในใบ (53.1%) มากกว่าในก้าน/ลำต้นและราก ราเอนโดไฟท์ ไอโซเลต AL1T1 และ AL1T2 ซึ่งคัดแยกจากลำต้นของว่านหางจระเข้ (Aloe vera L.) มีถทธิ์ยับยั้ง เชื้อราได้หลายชนิด โดยยับยั้งการเจริญของ Colletotrichum gloeosporioides, Fusarium oxysporum f. sp. vasinfectum, Phellinus noxius, Rigidoporus microporus une Rhizoctonia solani ใค้มากกว่า 50% เมื่อทคสอบด้วยเทคนิค dual culture สารสกัดหยาบของไอโซเลต AL1T1 และ AL1T2 มีถุทธิ์ยับยั้งเชื้อราก่อโรคในพืชเมื่อทุดสอบเปรียบเทียบกับยาฆ่าเชื้อรา tridemorph และ carbendazim สารสกัดหยาบของไอโซเลต AL1T1 สามารถยับยั้งเชื้อรา P. noxius อย่างมี นัยสำคัญ (P<0.05) โคยมีค่า MIC เท่ากับ 0.31 มก./มล. ซึ่งน้อยกว่าค่า MIC ของยาฆ่าเชื้อรา tridemorph (0.62 มก./มล.) ขณะที่สารสกัดหยาบของไอโซเลต AL1T2 สามารถยับยั้งเชื้อรา F. oxysporum f. sp. vasinfectum อย่างมีนัยสำคัญ (P<0.05) โดยมีค่า MIC เท่ากับ 0.16 มก./มล. ซึ่งน้อยกว่าค่า MIC ของยาฆ่าเชื้อรา carbendazim (0.31 มก./มล.) การตรวจสอบฤทธิ์ยับยั้งเชื้อรา ของสารสกัคหยาบด้วยวิธี TLC-autobiography พบว่า สามารถแยกสารประกอบที่มีฤทธิ์ยับยั้งเชื้อ ราของใอโซเลต AL1T1 ได้ เมื่อใช้ตัวทำละลายเคลื่อนที่คือ dichloromethane: methanol, 95:5 (v/v) โดยมีค่า Rf อยู่ในช่วง 0.22-0.33 ส่วนใอโซเลต AL1T2 พบว่า สร้างสารประกอบที่มีฤทธิ์ยับยั้ง

เชื้อราที่มีค่า Rf อยู่ในช่วง 0.20-0.33 โดยสามารถยับยั้งการเจริญของเส้นใยและการงอกของสปอร์เชื้อ รา *F. oxysporum* f. sp. *vasinfectum* ได้ นอกจากนี้ราเอนโคไฟท์ไอโซเลต BS1B201, C4IV301 และ PS1IV102 สามารถสร้าง indole acetic acid (IAA) ได้ในปริมาณ 40.7±0.5 ไมโครกรัม/มล., 23.4±0.1 ไมโครกรัม/มล. และ 16.8±0.1 ไมโครกรัม/มล. ตามลำดับ สามารถจำแนกราเอนโคไฟท์ ไอโซเลต AL1T1 และ AL1T2 โดยอาศัยลักษณะสัณฐานวิทยาและลักษณะทางพันธุกรรมอยู่ใน ไฟลัม Ascomycota ในอันดับ Sordariales โดยมีความเหมือนของลำดับเบสของยีนบริเวณ ITS rRNA เท่ากับ 99-100%

จากการทดลองสามารถกัดแขกราเอนโดไฟท์ที่สร้างไอระเหยอินทรีย์จากใบของขางพาราได้ จำนวน 3 ไอโซเลต ซึ่งราเอนโดไฟท์ทั้งหมดจัดจำแนกอยู่ในสกุล Muscodor วงศ์ Xylariaceae โดย อาศัยลักษณะทางสัณฐานวิทยาและลักษณะทางพันธุกรรม เชื้อราทุกไอโซเลตสามารถสร้าง ใอระเหยอินทรีย์ (VOCs) ที่มีฤทธิ์ยับยั้งจุลินทรีย์ได้หลายชนิด เช่น แบกทีเรีย, ยีสต์ และราเส้นสาย เชื้อราไอโซเลต RTM5IV3 มีความเหมือนกับลำดับเบสของยีน ITS rRNA กับเชื้อรา Muscodor สปีชีส์อื่นน้อยกว่า 86% ดังนั้นจึงถูกตั้งเป็นสปีชีส์ใหม่ โดยให้ชื่อว่า Muscodor heveae sp. nov. ซึ่ง สร้างไอระเหยอินทรีย์ที่มีสาร 3-methylbutan-1-ol เป็นสารหลักและ 3-methylbutyl acetate และ อนุพันธ์ของ azulene เป็นสารรอง VOCs ที่สร้างโดยเชื้อรา Muscodor มีศักยภาพในการควบคุม เชื้อจุลินทรีย์ก่อโรคทั้งแบคทีเรีย, ยีสต์ และราเส้นสาย ยิ่งกว่านั้นไอระเหยอินทรีย์ของ M. heveae สามารถยับยั้งเชื้อราก่อโรค P. noxius และ R. microporus ซึ่งเป็นสาเหตุของโรคในระบบรากของ ยางพารา

การรมด้วย ใอระเหยอินทรีย์ (biofumigation) เป็นการตรวจสอบประสิทธิภาพของ *M. heveae* ในการควบคุมโรครากขาวในยางพารา โดย ใอระเหยอินทรีย์ของ *M. heveae* สามารถ ยับยั้งการเจริญของเชื้อรา *R. microporus* ในระดับห้องปฏิบัติการด้วยเปอร์เซ็นต์ยับยั้งการเจริญ เท่ากับ 100 ส่วนการทดลองในระดับ โรงเรือน โดย ใช้หัวเชื้อ *M. heveae* รมดินที่มีเชื้อรา *R. microporus* เจริญอยู่ ทำการบันทึกค่าการเกิดโรค (disease score) เมื่อพืชอายุครบ 150 วัน และ วิเคราะห์ผลทางสถิติ การทดลองประกอบด้วย 7 กรรมวิธี การรมด้วย ไอระเหยอินทรีย์ของ *M. heveae* สามารถยับยั้งการเกิดโรค ได้เป็นอย่างดี โดยกรรมวิธี RMH40 และ RMH80 ซึ่งมี หัวเชื้อ *M. heveae* ปริมาณ 40 ก./1 กก. และ 80 ก./1 กก. ตามลำดับ ให้ผลไม่แตกต่างกันอย่างมี นัยสำคัญ (P<0.05) เมื่อเปรียบเทียบกับกรรมวิธี RT ที่ใส่ยาฆ่าเชื้อรา tridemorph และชุดควบคุมที่ ไม่ได้ไส่เชื้อราก่อโรคพืช กรรมวิธี RMH40 และ RMH80 ให้ผลแตกต่างกันอย่างมีนัยสำคัญ (P<0.05) กับกรรมวิธี R ที่ใส่เฉพาะเชื้อราก่อโรค โดยกรรมวิธี RMH40 และ RMH80 มีก่าการเกิด โรคเท่ากับ 1.3 ±0.5 และมีอัตราการอยู่รอดของขางพาราเท่ากับ 100% ขณะที่กรรมวิธี R มีค่าการ เกิดโรคเท่ากับ 4.8±0.5 และมีอัตราการอยู่รอดของขางพาราเท่ากับ 25 % อาการสำคัญที่แสดงการ เกิดโรคคือ พบ rhizomorphs สีขาวของเชื้อ *R. microporus* บนรากพืชที่มีการติดเชื้อ การทดสอบ ฤทธิ์ยับยั้งของสารสังเคราะห์ไอระเหยอินทรีย์ 3-methylbutan-1-ol, 3-methylbutyl acetate และ 2-methylpropanoic acid พบว่า สามารถยับยั้งการเจริญของเชื้อรา การเจริญของรากและต้นของ พืชทดสอบได้ แต่ไม่ส่งผลต่อยางพาราเมื่อสังเกตจากการทดลองการควบคุมโรคด้วยไอระเหยอินทรีย์ ของ *M. heveae* ในระดับโรงเรือน ดังนั้นไอระเหยอินทรีย์ที่มีฤทธิ์ทางชีวภาพจากเชื้อรา *M. heveae* สามารถใช้เป็นวิธีทางเลือกสำหรับการควบคุมโรครากขาวในยางพารา และสามารถนำไปประยุกต์ใช้ ทางการเกษตรและอุตสาหกรรมต่อไป

<mark>คำสำคัญ</mark> ราเอนโคไฟท์, ฤทธิ์ยับยั้งเชื้อรา, indole acetic acid, *Muscodor*, ยางพารา, ไอระเหย อินทรีย์ที่มีฤทธิ์ยับยั้งจุลินทรีย์



Dissertation Title	Control of Rubber Tree Pathogens by End	lophytic Fungi
Author	Ms.Sakuntala Siri-Udom	
Degree	Doctor of Philosophy (Applied Microbiol	ogy)
Advisory Committee	Prof. Dr. Saisamorn Lumyong	Advisor
	Asst. Prof. Dr. Yingmanee Tragoolpua	Co-advisor
	Dr. Boonsom Bussaban	Co-advisor

ABSTRACT

A total of 271 endophytic fungi were isolated from 13 medicinal plants and the rubber tree (Hevea brasiliensis Müll.Arg.) in northern and northeastern of Thailand. Most endophyte were isolated from *Clinacanthus nutans* (Burm.f.) Lindau (24.4%), followed by Rhinacanthus nasutus (Linn.) Kurz (10.3%) and C. siamensis Bremek (8.9%), respectively. Endophytic fungi were more prevalent in the leaves (53.1%) than the branches/stems and the roots. Isolate AL1T1 and AL1T2 from stem of Aloe vera L. exhibited board range of antifungal activity above 50% growth inhibition by dual culture technique toward to Colletotrichum gloeosporioides, Fusarium oxysporum f. sp. vasinfectum, Phellinus noxius, Rigidoporus microporus and Rhizoctonia solani AG-2. Compared with the positive control tridemorph and carbendazim, crude extract of AL1T1 and AL1T2 showed a strong antifungal activity on the above phytopathogens. Crude extract of AL1T1 significantly inhibited P. noxius with MIC 0.31 mg/ml (P<0.05) and less than MIC value of tridemorph (0.62 mg/ml). While, AL1T2 crude extract significantly inhibited F. oxysporum f. sp. vasinfectum with MIC 0.16 mg/ml (P<0.05) and less than MIC value of carbendazim (0.31 mg/ml). The antifungal activity of AL1T1 and AL1T2 crude extract were examined by TLC-bioautography. The results showed the separation of active compounds and Rf values ranged from 0.22-0.33 of AL1T1 (separated using 95:5 (v/v) dichloromethane: methanol) was found to inhibit mycelial growth of P. noxius. The active compounds of AL1T2 showed Rf values ranged from 0.20-0.33 and inhibited mycelial growth and spore germination of F. oxysporum f. sp. vasinfectum. Isolate BS1B201, C4IV301 and PS1IV102 produced plant growth hormone,

indole acetic acid (IAA) at the concentration of $40.7\pm0.5 \ \mu g/ml$, $23.4\pm0.1 \ \mu g/ml$ and $16.8\pm0.1 \ \mu g/ml$ respectively. Endophytic fungi isolates AL1T1 and AL1T2 belong to phylum Ascomycota in the order Sordariales based on morphological characteristic and the sequence similarities of ITS rRNA gene (99-100% sequence similarity).

Volatile metabolite-producing endophytic fungi were isolated from leaves of rubber tree and their antimicrobial competence were studied. Three isolates was obtained, and their phenotypic and phylogenetic relationship with the genus *Muscodor* in the family *Xylariaceae* was studied. All isolates could produce volatile metabolites with apparent antimicrobial activity against diverse tested microbes (bacteria, yeast and filamentous fungi). An isolate, RTM5IV3, with <86% similarity with ITS rRNA gene compared to other species of the genus *Muscodor*, was proposed as a novel species with the name, *Muscodor heveae* sp. nov. Its bioactive volatile metabolites included 3-methylbutan-1-ol as a major component, followed by 3-methylbutyl acetate and azulene derivatives. The volatile organic compounds (VOCs) produced by the *Muscodor* isolates have potential for biological control of pathogenic microorganisms such as bacteria, yeast and filamentous fungi. Furthermore, the VOCs of *M. heveae* were active against the pathogenic fungi, *P. noxius* and *R. microporus* causing root disease of rubber tree.

The bioactive compounds of *M. heveae* were examined by the process of biofumigation for the purposes of controling white root rot disease in the rubber trees. VOCs of *M. heveae* possess antimicrobial activity against *R. microporus in vitro* with 100% growth inhibition. *In vivo* tests were carried out under greenhouse conditions using *M. heveae* inoculum fumigated soil that had been inoculated with *R. microporus* inoculum. After which, all 7 treatments were compared. Significant differences were observed with a disease score at 150 days after treatment. Biofumigation by *M. heveae* showed great suppression of the disease. Biocontrol treatments; RMH40 (40 g 1 kg⁻¹ *M. heveae* inoculum) and RMH80 (80 g 1 kg⁻¹ *M. heveae* inoculum) were not found to be significantly difference when compared with fungicide treatment (RT) and the non-infested control, but results were found to be significantly difference from those of the untreated inoculated control (R) (P<0.05). RMH40 and RMH80 revealed a low disease score of 1.3 ±0.5 with a high survival rate of rubber tree seedling (100%), while R showed

the highest disease score of 4.8±0.5 with 25% survival rate of rubber tree seedling. A sign of disease is the presence of rhizomorphs of *R. microporus* on the infected roots, appearing as a white color. The artificial volatile compounds test confirmed that the major component, 3-methylbutan-1-ol, and the minor compounds, 3-methylbutyl acetate and 2-methylpropanoic acid, inhibited mycelial growth of fungal pathogens and root and shoot elongation in the tested plant but had no effect on the rubber tree when fumigation was demonstrated and observed *in vivo*. The bioactive VOCs of *M. heveae* would be an alternative method for the control of white root rot disease in rubber trees, while there is potential to expand these findings toward further agricultural and industrial applications.

Keywords Endophytic fungi, antifungal activity, indole acetic acid, Muscodor, rubber



CONTENTS

Acknowledgement	iii
Abstract in Thai	iv
Abstract in English	vii
List of Tables	xvii
List of Figures	xix
Statement of Originally in Thai	XXV
Statement of Originally in English	xxvi
Chapter 1 General introduction and thesis outline	1
Chapter 2 Literature review	5
2.1 Introduction of endophytic fungi	5
2.2 Natural products from endophytic fungi	8
2.3 Volatile producing-endophytic fungi	10
2.4 Plant growth promoting endophytic fungi	15
2.5 Para rubber tree and natural rubber	18
2.6 Endophytic fungi in para rubber tree	20
2.7 rubber tree disease	21
2.7.1 Leaf disease	21
2.7.1.1 Leaf and seedling blight caused by	
Fusarium oxysporum f. sp. vasinfectum	21

2.7.1.2 Leaf and seedling blight caused by	
Rhizoctonia solani Kuhn.	22
2.7.1.3 Seedling blight caused by <i>Phytophthora</i> spp.	22
2.7.1.4 Anthracnose leaf spot and leaf blight	23
2.7.2 Stem and branches diseases	23
2.7.2.1 Die back disease	23
2.7.3 Root disease	24
2.7.3.1 White root rot disease	24
2.7.3.2 Brown root rot disease	26
2.7.3.3 Red root rot disease	27
2.8 Management of rubber tree disease	28
2.8.1 Chemical control	28
2.8.2 Biological Control	28
Chapter 3 Isolation and screening of endophytic fungi that capable of	
producing bioactive compounds and indole acetic acid	30
3.1 Introduction	30
3.2 Materials and methods	31
3.2.1 Preparation and surface sterilization of plant materials	31
3.2.2 Isolation of endophytic fungi	32
3.2.3 Isolation and pathogenicity test of phytopathogenic fungus	32

	3.2.4 Phenotypic and genotypic classification	33
	3.2.5 Screening for antagonism against rubber tree	
	pathogens by dual culture method	36
	3.2.6 Crude extract preparation	37
	3.2.7 Bioassay of crude extract by agar well diffusion method	37
	3.2.8 Determination of minimum inhibitory concentrations	
	(MICs) of crude extract	37
	3.2.9 Antifungal activities assay by TLC-bioautography	38
	3.2.10 Preliminary assay of hemolysis	39
	3.2.11 Screening for indole acetic acid (IAA) production	39
	3.2.12 Extraction of crude IAA and Thin layer chromatography	
	(TLC)	39
Re	esults	40
	3.3.1 Isolation of endophytic fungi	40
4		40
	causing leaf and seedling blight in the rubber tree	42
	3.3.3 Genotypic classification of Rigidoporus microporus	44
	3.3.4 Screening for antagonism against rubber tree pathogens	
	by dual culture method	45
	3.3.5 Bioassay of crude extract by agar well diffusion method	48

3.3

Page

3.3.6 Determination of minimum inhibitory concentrations (MICs)	
of crude extract	50
3.3.7 Antifungal activities assay by TLC-bioautography	52
3.3.8 Preliminary assay of hemolysis	53
3.3.9 Screening for indole acetic acid (IAA) production	54
3.3.10 Extraction of crude IAA and thin layer	
chromatography (TLC)	55
3.3.11 Phenotypic and genotypic classification of active	
endophytic fungi	56
3.4 Discussion	59
Chapter 4 Existence of Muscodor in leaves of the rubber tree	63
4.1 Introduction	63
4.2 Materials and methods	64
4.2.1 Preparation and surface sterilization of plant materials	64
4.2.2 Isolation of volatile metabolite-producing endophytic fungi	65
4.2.3 Phenotypic classification	67
4.2.4 Genotypic classification	67
4.2.5 Profiling and comparison of volatile metabolites	68
4.3 Results	68

Page

4.3.1 Volatile metabolites-producing endophytic fungi of rubber	
tree leave	68
4.3.2 Volatile analysis of Muscodor spp.	69
4.3.3 Taxonomic description of a novel species	81
4.4 Discussion	87
Chapter 5 Antimicrobial activity of Muscodor spp. and relevant alleopathy effects	90
5.1 Introduction	90
5.2 Materials and methods	91
5.2.1 In vitro antimicrobial assay of VOCs	91
5.2.2 Effect of VOCs from Muscodor heveae on root and	
shoot growth	92
5.2.3 Statistical analysis	93
5.3 Results	95
5.3.1 Antimicrobial activity of volatile metabolites produced	
by Muscodor isolates	95
5.3.2 Effect of VOCs from <i>M. heveae</i> on root and shoot growth	100
5.4 Discussion	104
Chapter 6 Mycofumigation with Muscodor heveae for controlling of	
White root rot disease	107
6.1 Introduction	107

6.2 Materials and methods	108
6.2.1 Fungal cultures	108
6.2.2 Preparation of Muscodor heveae and	109
Rigidoorus microporus inocula	
6.2.3 Effect of VOCs from Muscodor heveae	
on Rigidoporus microporus in vivo	111
6.2.4 Soil analysis	112
6.2.5 Statistical analysis	112
6.3 Results	112
6.3.1 Effect of VOCs from Muscodor heveae on	
Rigidoporus microporus in vivo	112
6.3.2 Soil analysis	114
6.4 Discussion	116
Chapter 7 General discussion and conclusion	118
7.1 Endophytic fungi in medicinal plants and the rubber tree	118
7.2 Antifungal activity and plant growth promoting property of	
endophytic fungi	119
7.3 Molecular and morphological identification of genus Muscodor	120
7.4 Antagonistic volatile compounds of Muscodor heveae and revelant	
alleopathy	121

Page

7.5 Controlling of white root rot disease in rubber tree using volatile	
compounds from Muscodor heveae	121
References	123
List of publications	146
Appendix	
Appendix A	147
Appendix B	149
Appendix C	151
Appendix D	153
Appendix E	154
Appendix F	156
Appendix G	158
Appendix H ASUM STATISTICS HIS SOLVED	160
Curriculum Vitae yright [©] by Chiang Mai University	162
All rights reserved	

LIST OF TABLES

Table 2.1 Symbiotic criteria used to characterize endophytic fungi classes	6
Table 2.2 Comparison of characteristics of C-endophytes and NC-endophytes	7
Table 2.3 Structure, odors and functions of VOCs produced by fungi	12
Table 2.4 List of non-systemic endophytic fungal taxa that produce VOCs	13
Table 2.5 Major volatile compounds and bioactivity compare with related	
Muscodor spp.	16
Table 3.1 Collection details of selected plants in this study	35
Table 3.2 The list of pathogenic fungi used in this study.	36
Table 3.3 Number of endophytic fungi isolated from medicinal plants and	
the rubber tree	41
Table 3.4 Antifungal activity of endophytic fungi from medicinal plant	46
Table 3.5 Percentage of inhibition of mycelial growth after treated with	
crude extract	48
Table 3.6 Determination of minimum inhibitory concentrations (MICs)	
of crude extract and commercial fungicide	51
Table 3.7 Screening for IAA production of endophytic fungi	55
Table 3.8 Closest related taxa of AL1T1 and AL1T2 based on the	
sequence similarities of ITS1 rRNA gene	57
Table 4.1 Closest related species of Muscodor isolates based on the sequence	
similarities of ITS rRNA gene	70

LIST OF TABLES (CONTINUED)

Table 4.2 Composition of the volatile metabolites produced by	
Muscodor spp. by GC/MS	73
Table 4.3 Growing condition of Muscodor spp. and GC-MS system for	
analysis of their volatile metabolites	80
Table 4.4 Morphology and bioactivity of <i>Muscodor heveae</i> and its closely related	
phylogenetic species	86
Table 5.1 The list of tested microbes used in this study	94
Table 5.2 Antimicrobial activity of volatile metabolites-producing Muscodor	
against human and plant pathogens	97
Table 5.3 Effect of artificial volatile compounds on the hyphal growth	
of fungal pathogens	99
Table 5.4 Effect of VOCs from Muscodor heveae on root and shoot growth	101
Table 5.5 Effect of artificial volatile compounds on root and shoot growth of	
tested plant	103
Table 6.1 Survival rate (%) of rubber tree after biofumigation and chemical	
analysis of soil used in the pot culture	115

LIST OF FIGURES

Figure 1.1 Schematic presentation of the relationships between	
chapters of the thesis	4
Figure 2.1 Endophyte asexual life cycle	6
Figure 2.2 The world's first billion dollar anticancer drug, taxol	8
Figure 2.3 Rubber tree plantation, leaves and seeds of Hevea brasiliensis	18
Figure 2.4 Colony of Fusarium oxysporum f. sp. vasinfectum on PDA,	
symptom of seedling blight	22
Figure 2.5 The symptom of Leaf and seedling blight caused by	
Rhizoctonia solani Kuhn.	22
Figure 2.6 Anthracnose leaf spot disease on rubber tree leaves	23
Figure 2.7 Early stage of white root rot disease	25
Figure 2.8 White root rot disease cycle	26
Figure 2.9 Colony of <i>Phellinus noxius</i> on PDA, staghorn-like hyphae and	
arthrospores	27
Figure 3.1 Colleting sites (star marks) of selected plants in Thailand where used	
in this study	32
Figure 3.2 Plant samples in this study	34
Figure 3.3 Antifungal activities assay by TLC-bioautography	38

LIST OF FIGURES (CONTINUED)

Figure 3.4 Fungal isolate JP1B1-1 and JP1B1-2 grow out from	
branch segment of Jatropha podagrica	41
Figure 3.5 Colony of Fusarium oxysporum f. sp. vasinfectum, symptom of	
seedling blight, macroconidia and microconidia	42
Figure 3.6 Neighbor-joining (NJ) phylogenetic tree of Fusarium spp.	
based on ITS rRNA sequence alignment of 13 sequences	43
Figure 3.7 Pathogenicity test of <i>Fusarium oxysporum</i> f. sp. vasinfectum (PRT3-1)	
on leaf and branch of rubber tree seedling	43
Figure 3.8 Characteristic of Rigidoporus microporus	44
Figure 3.9 Neighbor-joining (NJ) phylogenetic tree of Rigidoporus microporus	
based on ITS rRNA sequence alignment of 13 sequences	45
Figure 3.10 Preliminary testing for antifungal activity of endophytic fungi (isolate	
AL1T1 and AL1T2) against plant pathogenic fungi by dual culture technique	47
Figure 3.11 Antifungal activity of endophytic fungus (isolate JA1B1-2)	47
Figure 3.12 Bioassay of crude extract by agar well diffusion method	49
Figure 3.13 Mycelial growth of <i>Phellinus noxius</i> under light microscope	50
Figure 3.14 Minimum inhibitory concentrations (MICs) value of AL1T1	
and AL1T2 crude extract	52
Figure 3.15 TLC-bioautograph of AL1T1 crude extract against mycelial	
growth of Phellinus noxius	53

LIST OF FIGURES (CONTINUED)

Page

Figure 3.16 TLC-bioautograph of AL1T2 crude extract against mycelial growth	
and spore germination of Fusarium oxysporum f. sp. vasinfectum	53
Figure 3.17 Preliminary assay of hemolysis	54
Figure 3.18 Thin layer chromatography (TLC) of IAA crude extract	56
Figure 3.19 Morphological characteristics of isolate AL1T1	57
Figure 3.20 Morphological characteristics of isolate AL1T2	58
Figure 3.21 Neighbor-joining (NJ) phylogenetic tree of isolate AL1T1 and	
AL1T2 based on ITS rRNA sequence alignment of 14 sequences	59
Figure 4.1 Colleting sites (star marks) of rubber tree in Nong Bua Lamphu	
province, Thailand	65
Figure 4.2 Muscodor grew out from the plant segment on half strength PDA	66
Figure 4.3 Neighbor-joining (NJ) phylogenetic tree of Muscodor spp. based on	
ITS rRNA sequence alignment of 25 sequences	71
Figure 4.4 SPME-GC/MS profile of 3-methylbutan-1-ol, 3-methylbutyl	
acetate and 2-methylpropanoic acid produced by Muscodor heveae cultured	
on PDA	72
Figure 4.5 Morphological characteristics of Muscodor heveae	83
Figure 4.6 Morphological characteristics of Muscodor vitigenus	84
Figure 4.7 Morphological characteristic of Muscodor equiseti	85

LIST OF FIGURES (CONTINUED)

Figure 5.1 Antifungal ability test of VOCs from Muscodor heveae against	
Aspergillus flavus, A. niger, Rhizoctonia solani AG-2 and Phytophthora parasitica	96
Figure 5.2 Antifungal ability test of VOCs from Muscodor heveae against	
Ganoderma australe, Colletotrichum gloeosporioides, Phellinus noxius and	
Rigidoporus microporus	96
Figure 5.3 Effect of artificial volatile compounds; 2-methylpropanoic acid on the	
hyphal growth of Rigidoporus microporus	98
Figure 5.4 Effect of artificial volatile compounds; 3-methylbutyl acetate	
on the hyphal growth of Rigidoporus microporus and Phellinus noxius	98
Figure 5.5 Effect of VOCs from Muscodor heveae on the root length and	
shoot length	102
Figure 5.6 Effect of artificial volatile compounds, 3-methylbutan-1-ol on	
the root and shoot growth of ruzi grass	102
Figure 6.1 Preparation of Muscodor heveae inocula	110
Figure 6.2 Preparation of Rigidoporus microporus inocula	110
Figure 6.3 Inoculation method for investigation the effect of VOCs from	
Muscodor heveae on Rigidoporus microporus in vivo.	112
Figure 6.4 Comparison of disease score for different treatments	113
Figure 6.5 Biofumigation test for suppression of white root rot disease caused	
by Rigidoporus microporus after 150 days	114

LIST OF ABBREVIATION

μl	microlitre
cm	centimeter
g	gram
kg	kilogram
h	hour
ml	millilitre
mm	millimeter
min	minute
М (G	molar
mM	millimolar
PCR	polymerase chain reaction
RNA	ribonucleic acid
rpm	round per minute
sec	second
v/v	volume by volume
	MAI UNIVERS

LIST OF SYMBOLS





ข้อความแห่งการริเริ่ม

- วิทยานิพนธ์นี้ได้นำเสนอการคัดแยกราเอนโคไฟท์ในพืชสมุนไพรไทยและยางพารา เพื่อ ศึกษาถึงฤทธิ์ยับยั้งจุลินทรีย์ก่อโรคพืชของสารปฏิชีวนะและไอระเหยอินทรีย์ของราเอนโค ไฟท์ รวมทั้งสมบัติในการส่งเสริมการเจริญเติบโตของพืช เพื่อการนำไปใช้ประโยชน์ทาง การเกษตรและอุตสาหกรรมต่อไป
- เพื่อนำเสนอวิธีทางชีวภาพในการควบคุมโรคในยางพาราโดยใช้ประโยชน์จากราเอนโดไฟท์ ซึ่งการรมด้วยไอระเหยอินทรีย์จากราเอนโดไฟท์ที่มีฤทธิ์ยับยั้งจุลินทรีย์ก่อโรคพืชถูกนำมา ประยุกต์ใช้



STATEMENT OF ORIGINALITY

- 1. This study provides the isolation and screening endophytic fungi from Thai medicinal plants and the rubber tree that capable of producing antibiotics, volatile metabolites and plant growth promoting metabolites and expand these findings toward further agricultural and industrial applications.
- 2. Discovery of alternative technique to control plant pathogens by biofumigation using antagonistic volatile metabolites from endophytic fungi.



CHAPTER 1

General introduction and thesis outline

1.1 Introduction

Rubber tree (Hevea brasiliensis Müll.Arg.) is an economic crop that produces natural latex as a major raw material for rubber manufacturers. Most of plantation areas of rubber tree are located in South East Asia such as Malaysia, Indonesia and Thailand (van Beilen and Poirier, 2007). Especially, in Thailand is optimally located to supply the natural rubber demands of the Asia Pacific region. Although recent farming practices for rubber tree are well developed to earn high yield of latex, the plant diseases that cause reduction of latex yield and lethal effect of rubber tree are still often found (Jayasuriya and Deacon, 1995; Evueh and Ogbebor, 2008). One of the main pathogens in rubber plantation is Rigidoporus sp., which causes white root disease. This is the most destructive root disease in rubber plantations which kill the tree irrespective of age or health status and results in economic losses to the latex industry (Guyot and Flari, 2002; Ogbebor et al., 2015). Although, many chemicals such as metalazyl and tridermorph were used to control fungal disease but they have a negative effect on human health, cause environmental pollution, leave residues in agricultural soil (Harman et al., 2004; Jayasinghe, 2010) and induce chemical resistance of several plant pathogenic fungi (Kim and Hwang, 2007). by Chiang Mai University

Fungal endophytes are determined by their in *planta* life without causing apparent disease on the host plants (Chaverri and Gazis, 2011; Prakash, 2015). Along with the symbiotic life, they have been recognized for their advantages in sustaining plant growth and plant defensive system (Strobel and Daisy, 2003; Prakash, 2015). With such benefits, they become one of the promising microbial resources as for biocontrol agents applied in agricultura purposes. A mode of action of endophytes in prevention and/or suppress of plant diseases caused by phytopathogens is to form antimicrobial substances. Moreover, some endophytic fungi that belong to the families *Xylariaceae* of the phylum Ascomycota

have been reported their capacity to form volatile metabolites with antimicrobial activity. Among of endophytes, *Muscodor albus* is the first known fungal endophyte isolated from *Cinnamonum zeylanicum* which produces bioactive volatile metabolites (Strobel *et al.*, 2001; Ezra and Strobel, 2004).

Thus, alternative methods with emphasis on biological control using microorganism have been suggested as the most sustainable long-term solution. Natural products from microorganism are the most important source for new and potential pharmaceutical agents. Specifically, fungal endophytes are expected to be a potential source of new natural bioactive agents that potential kill a wide variety of harmful disease-causing agents including, but not limited to, phytopathogens, as well as bacteria, fungi, viruses and protozoans that affect humans and animals (Strobel and Daisy, 2003). The efficiency of biological control depend on interactions between biological control agent and pathogen (Viterbo et al., 2007) such as physical contact, synthesis of hydrolytic enzyme, toxic antibiotic compound, competition, and induction of resistance in plant host (Benítez et al., 2004). Furthermore, volatile metabolites-producing endophytes may be an alternative biological approach as biofumigation in control of plant diseases. It is known that sapwood and leaf fragments of rubber tree are a rich source of fungal endophytes (Evueh and Ogbebor, 2008; Rocha, 2010; Gazis, 2012), but only few studies revealed antimicrobial activity of them (Evueh and Ogbebor, 2008; Rocha, 2010). Moreover, no volatile metabolites-producing endophyte has yet been reported from rubber tree host.

1.2 Thesis objectives

1.2.1 To isolate and identify endophytic fungi from Thai medicinal plants and the rubber tree

1.2.2 To screen the effective fungi that capable of producing bioactive agents, plant growth promoting metabolites and volatile metabolites against rubber tree pathogens

1.2.3 To purify and characterize the effective bioactive agents and volatile metabolites

1.2.4 To investigate biological control of rubber tree pathogens *in vivo* by using endophytic fungi

1.3 Usefulness of the thesis

1.3.1 The possibility to discover the effective/novel endophytic fungi for controlling rubber tree pathogens or other pathogenic microorganisms.

1.3.2 Possible discovery of new species of endophytic fungi in the rubber tree.

1.3.3 Discovery of modify method or technique to control pathogens in rubber plantation.

ามยนต

1.4 Plan of the thesis

This thesis was initiated to collect 13 kinds of medicinal plants and the rubber tree from various locations in northern and northeastern of Thailand. Endophytic fungi was isolated, identified and screened for their antagonism. Setting the background to this work, in Chapter 2 reviews the knowledge about endophytic fungi, their ability to produce bioactive compounds and volatile organic compounds (VOCs), including the diversity of endophytic fungi in the rubber tree.

Chapter 3 describes the isolation, identification and screening endophytic fungi from Thai medicinal plants that capable of producing antimicrobial substances and plant growth promoting metabolite.

Chapter 4 describes the isolation and identification base on morphology, molecular, and volatile composition in genus *Muscodor* isolated from the rubber tree.

Chapter 5 describes the antimicrobial activity and phytotoxic activity of VOCs from *Muscodor*.

Chapter 6 describes the use of VOCs from *Muscodor* to control white root disease cause by *Rigidoporus microporus*.

Chapter 7 describes the summarization and discussion of data in this thesis.



Figure 1.1 Schematic presentation of the relationships between chapters of the thesis.



CHAPTER 2

Literature review

2.1 Introduction of endophytic fungi

Endophytic fungi are the fungi that colonize living in plant tissue without causing any symptoms or negative effects (Huang et al., 2001). They have been found in every plant species and may live in roots, stems and/or leaves, emerging to sporulate at plant tissue (Stone et al., 2004) (Figure 2.1). Two endophytic groups, Clavicipitaceous (Cendophytes) and nonclavicipitaceous (NC-endophytes) have been recognized based on phylogenetic analysis, plant hosts and ecological functions (Table 2.1). Although the systemic, C-endophytes and NC-endophytes are considered endophytes but they differ in important ways (Table 2.2). C-endophytes were lived in some grasses, while NCendophytes are highly diverse fungi. They can be found in asymptomatic tissues of nonvascular plants, fern and allies, conifers, and angiosperms (Rodriguez et al., 2009). C-endophytes, class 1 represent the phylogenetic characteristic related clavicipitaceous species which limited to some cold and warm season grass (Rodriguez et al., 2009). Endophytes of nongrass hosts in class 2-4 represent a broader range taxa from several orders and families of Ascomycetes and some Basidiomycetes families (Stone et al., 2004). The unique ability of Class 2 is their ability to confer habitat-adapted stress tolerance to host plant (Rodriguez et al., 2008; Rodriguez et al., 2009). Transmission of Class 1 endophytes is primary vertical, with maternal plants transfer fungi to offspring by seed infection. Class 1 endophyte frequently increase plant biomass, drought tolerance and produce active compounds that protect plants from herbivores (Clay, 1988).



The endophyte is concentrated in the base of plant

Figure 2.1 Endophyte asexual life cycle.

Table 2.1 Symbiotic criteria used to characterize endophytic fungi classes (Rodriguez et al., 2009).

Characteristic	Clavicipitaceous	Nonc	lavicipitaceou	S
Criteria	Class 1	Class 2	Class 3	Class 4
Host range	Narrow	Broad	Broad	Broad
Tissue(s) colonized	Shoot and Rhizome	Shoot, root and rhizome	Shoot	Root
	au a	1. /		11
In planta colonization	Extensive	Extensive	Limited	Extensive
In planta biodiversity	Low by Chia	Low	High	Unknown
Transmission	Vertical and horizontal	Vertical and horizontal	Horizontal	Horizontal
Fitness benefits*	NHA	NHA and HA	NHA	NHA

*Nonhabitat-adapted (NHA) benefits such as drought tolerance and growth enhancement are common among endophytes regardless of the habitat of origin. Habitat-adapted (HA) benefits result from habitat-specific selective pressures such as pH, temperature and salinity. All three classes in NC-endophytes have broad host ranges. Class 2 endophytes may live in both above- and below-ground tissues. Class 3 and 4 endophytes are limited to above-ground tissue and roots. The colonization of host tissues also differs, Class 2 and 4 endophytes are ability of extensive plant tissue colonization while Class 3 endophytes form highly localized infections (Rodriguez *et al.*, 2009).

Table 2.2 Comparison of characteristics of C-endophytes and NC-endophytes (Stone *et al.*, 2004).

C- endophytes	NC-endophytes
1. Few species, only Clavicipitaceae	1. Many species, taxonomically diverse
2. Extensive internal colonization	2. Restricted internal colonization
3. Occurring in several host species	3. Most species with limited host species
4. Systemic, seed transmitted	4. Nonsystemic, spore transmitted
5. Host colonized by only one species	5. Hosts infected by several species

Recently, it shows that endophytes are not host specific (Cohen, 2006) and single endophytic fungi can infect a wide host range. Some strains of the same endophytic fungi isolated from different part of the same host showed different ability to utilized different substances. So, endophytic fungi can be isolated from different plants belongs to the different families (Jalgaonwala *et al.*, 2011). C-endophytes show hyphae colonization in all grass tissues and is found both in the seed coat and the embryo (Stone *et al.*, 2004). While, NC-endophytes are limited infection in grass tissues but there are two genera *Phialophora* and *Gliocladium* can be seed-borne. However, there also are seed-borne endophyte in non-grass hosts. The dominant endophyte colonization in healthy tissue often is limited to no more than few cells. Endophytic fungi are viewed as outstanding source of secondary metabolite bioactive compounds and used as biocontrol agents applied in agriculture. A mode of action of endophytic fungi in prevention and/or suppression of plant diseases caused by phytopathogens is the production of antimicrobial substances. Moreover, endophytic fungi from medicinal plants could be rich source of active compounds (Sappapan *et al.*, 2008). It was found that symbiotic plant stimulates defense system more quickly than non-symbiotic plants after pathogen infection (Jalgaonwala *et al.*, 2011).

2.2 Natural products from endophytic fungi

Endophytic fungi are source of bioactive compounds with antimicrobial activities and also protect against insect, pest and plant pathogens (Jalgaonwala *et al.*, 2011). The world's first billion dollar anticancer drug, taxol is a highly functionalized diterpenoid (Stierle *et al.*, 1993) (Figure 2.2). It is an antitumor agents but it is used to treat other human tissue-proferating disease as well (Jalgaonwala *et al.*, 2011). *Taxomyces andreanae* is an endophytic fungus of the class Hyphomycete. It is the first report of an endophytic fungus from *Taxus brevifolia* that produce taxol (Stierle *et al.*, 1993). Recently, several endophytic fungi show ability to produce taxol such as *Fusarium redolens* isolated from *T. wallichiana* (Garyali *et al.*, 2013) and the 3 taxol-producing fungi; *Guignardia mangiferae, Fusarium proliferatum* and *Colletotrichum gloeosporioides* isolated from *T. media* (Xiong *et al.*, 2013).



Figure 2.2 The world's first billion dollar anticancer drug, taxol (Jalgaonwala *et al.*, 2011)

Antioxidant secondary metabolites are often produced by endophytic fungi. The previous report showed antioxidant capacity of endophytic fungi isolated from wheat (*Triticum durum*) with antimicrobial activity against bacteria, yeast and phytopathogenic fungi (Sadrati *et al.*, 2013). Moreover, many bioactive compounds with antimicrobial properties have been successfully discovered from several endophytic fungi such as *Phomopsis* sp. isolated from the leaves of *Vitex negundo* L. showed significant antimicrobial activity against human pathogenic bacteria (Desale and Bodhankar, 2013). Another isolate from mangrove cloud produce cytosporone B and cytosporone C with

antifungal activities against *Candida albicans* and *F. oxysporum* (Huang *et al.*, 2008). Furthermore, P. cassia isolated from Cassia spectabilis produced ethyl 2, 4-dihydroxy-5, 6-dimethylbenzoate and phomopsilactone displayed antifungal activity against fungal pathogens; Cladosporium cladosporioides and C. sphaerospermum (Silva et al., 2005). An endophytic fungi, *Pestalotiopsis adusta* produced 2 active compounds, pesralachlorides A $(C_{21}H_{21}Cl_2NO_5)$ and pesralachlorides B $(C_{20}H_{18}Cl_2O_5)$. These compounds showed significant antifungal activities against F. culmorum, Gibberella zeae and Verticillium alboatrum (Li et al., 2008). Aspergillus fumigatus LN-4 isolated from stem bark of Melia azedarach produced 2 new alkaloids, 12β-hydroxy-13αmethoxyverruculogen TR-2 and 3-hydroxyfumiquinazoline A with potent antifungal activities against several plant pathogens such as Alternaria alternata, Al. solani, Botrytis cinerea, C. gloeosporioides, G. saubinettii, F. oxysporum f. sp. niveum, F. oxysporum f. sp. vasinfectum and F. solani (Li et al., 2012). While, Aspergillus sp. isolated from Panax notoginseng produced averythrin, versicolorin B and averantin with antifungal activities against F. solani. Moreover, averythrin and averantin also showed antibacterial activities against Bacillus subtilis (Liu et al., 2014). A new α-tetralone derivative, (3S)-3, 6, 7trihydroxy- α -tetralone, together with cercosporamide, β -sitosterol and trichodermin were produced by *Phoma* sp. isolated from *Arisaema erubescens*. These compounds were obtained from *Phoma* species for the first time except trichodermin and displayed antimicrobial activities (Wang et al., 2012). Chaetomium globosum isolated from Ginkgo biloba produced 6 active compounds, chaetoglobosin A, C, D, E, G and R with inhibitory activity against 2 phytogenic fungi Rhizopus stolonifer and Coniothyrium diplodiella (Zhang et al., 2013). Three know compounds 6, 8, 1'-tri-O-methyl averantin, aversin and 6, 8-di-O-methyl versiconol were produced by Penicillium purpurogenum. These compounds displayed antifungal activities against B. cinerea which cause grey mold disease (Li et al., 2014a).

Some antiviral agents are reported from endophytic fungi such as cytonic acid A and B have been isolated from *Cytonaema* sp. with antiviral activities to inhibit human cytomegalovirus (hCMV) protease (Guo *et al.*, 2000). Isoindolones derivatives compounds, emerimidine A and B produced by *Emericella* sp. (HK-ZJ) isolated from *Aegiceras corniculatum* showed anti-influenza A viral (H_1N_1) activities (Zhang *et al.*, 2011). Endophytic fungi from Thai medicinal plants have been reported their antiviral
activities against Herpes simplex virus type 1 including anticancer activities (Wiyakrutta et al., 2004). Other natural product were produce by endophytic fungi, nodulisporic acid was isolated from Nodulipotium sp., Bontia and Daphnoides with insecticidal activities against the larvae of the blowfly (Verma et al., 2007). Nematicidal or insecticidal compounds, 1, 3 oxazinane derivatives from *Geotrichum* sp. AL4, isolated from leaves of Adenocarpus indica showed effective nematicidal activity against nematodes (Jalgaonwala et al., 2011). Furthermore, anti-parasitic agent, cercosporin has been produced by endophytic fungus Mycosphaerella sp. associated with Psychotria horizontalis in Panama (Moreno et al., 2011). 2102423

2.3 Volatile producing-endophytic fungi

Volatile organic compounds (VOCs) are carbon-based solids and liquids which enter the gas phase by vaporizing at 0.01 kPa at 20 °C (Pagans et al., 2006). Fungi produced VOCs in both primary and secondary metabolism pathways (Korpi et al., 2009). Previous reports showed the mixture of fungal volatile compounds that have been identified were aldehydes, alcohols, benzene derivatives, cyclohexanes, ketones, hydrocabons, heterocycles, phenol, thioalcohols, thioesters and their derivatives (Mercier et al., 2007; Morath et al., 2012). Fungal VOCs have special odors and macrofungi such as mushrooms could produce distinctive odors with mixture of different VOCs while the microscopic fungi produced musty odors (Morath et al., 2012) (Table 2.3). In agriculture, fungal VOCs may be used as part of biological control strategies to control plant disease in the termed "mycofumigation". Moreover, these VOCs were interested in the plantgrowth promoting properties and used to prevent post-harvest fungal contamination. VOCs from *Phoma* sp. significantly enhanced the growth of tobacco seedlings (Naznin et al., 2014). In addition, volatile producing endophytic fungi able to detect the plant host and induce their colonization process by producing canonical plant growth regulating substance such as auxins and/or cytokinins (Ortiz-Castro et al., 2009). Thus VOCs play a role in plant morphogenetic process (Kanchiswamy et al., 2015). The VOCs produced by endophytic fungi have not been commonly reported. But some endophytic fungi that belong to the families Diaporthaceae, Hypocreaceae, and Xylariaceae of the phylum Ascomycota are notable for their capacity to form volatile metabolites with antimicrobial activity (Stinson et al., 2003b; Suwannarach et al., 2013b). Endophytic fungus in family *Stachybotriaceae, Myrothecium inunduatum* cloud produce bioactive VOCs with antifungal activity against plant pathogenic fungi. Moreover, in microaerophilic conditions, this fungus produced fuel related hydrocarbons such as octane and 1, 4-cyclohexadiene, 1-methyl- and cyclohexane, (1-ethylpropyl) (Banerjee *et al.*, 2010). *Phomopsis* sp. isolated from *Odontoglossum* sp. produced unique VOCs with antifungal properties against phytopathogenic fungi including sabinene, the compound that previously known in higher plant (Singh *et al.*, 2011) (Table 2.4).



ลิขสิทธิมหาวิทยาลัยเชียงไหม Copyright[©] by Chiang Mai University All rights reserved

Volatile compounds	Structure	Odor and functions
1-butanol-3-,methyl-, acetate	HO	banana odor, antifungal
6-pentyl-a-pyrone		coconut odor, antibiotic
β-caryophyllene		woody-spicy odor,
5.		plant-growth promoting
isobutyric acid	Ļ	rancid cheese-like odor,
-333-	СН	antifungal
benzyl aldehyde		almond odor, antimicrobial
1,8-cineole		camphor-like odor, antifungal
l'a	4	SIT
2-methyl-1-propanol	HO. L	mild alcohol odor, fungivore
S S S		attractant
2-heptanone	nun sina 8	cheese odor, unknown
3-methyl-butanol		component of truffle odor, unknown

Table 2.3 Structure, odors and functions of VOCs produced by fungi (Morath *et al.*,2012).

Taxon	Classification	Host	Major compounds	
Acremonium sp.	Hypocreales	Brachiaria brizantha	Unidentified antibiotic compounds	
Aspergillus niger	Trichocomaceae	Rosa damacaena	2-phenylethanol	
Botrytis sp. BTF21	Sclerotiniaceae	Musa spp.	Butane 2-methyl; β -butyrolactone, 2-butenedinitrile	
Cladosporium sp. MIF01	Capnodiales	Mimosa pudica	Butane 2-methyl, 1-propanol 2-methyl	
Fusarium sp. CID124	Nectriaceae	Centaurea stoebe	Unidentified sesquiterpenes	
Gliocladium roseum,	Hypocreales	Unknown	Hydrocarbons (benzene, heptane, 1-octene, octane,	
G. catenulatum	335	Charles ??	m-xylene, 3-methylnonane, dodecane, tridecane,	
			hexadecane and nonadecane)	
Meliniomyces variabilis	Helotiaceae	Pinus sylvestris	Ethanol; acet-aldehyde	
Myrothecium inunduatum	Hypocreales	Acalypha indica	3-octanone; 3-octanol; 7-octen-4-ol	
Nodulisporium sp.1	Xylariaceae	Cinnamomum loureirii	<i>i</i> β -elemene; β -selinene; α -selinene; 1-methyl-1,	
		AI INIVER	cyclohexadiene	
Nodulisporium sp.2	Xylariaceae	Olearia argophylla	1,3,8-p-Menthatriene; caryophyllene; eucalyptol	
Phialocephala fortinii	Helotiaceae	Pinus sylvestris	Ethanol, acet-aldehyde, toluene	
Phoma sp.	Pleosporales	Larrea tridentata	alphahumulene (sesquiterpene); alcohols; reduced	
	Copyright	[©] by Chiang Ma	naphthalene derivatives; trans-caryophyllene	
Phomopsis sp.	Diaporthales	Odontoglossum sp.	Sabinene (monoterpene); 1-butanol, 3-methyl;	
		0	benzeneethanol; 1-propanol, 2-methyl; 2-propanone	

Table 2.4 List of non-systemic endophytic fungal taxa that produce VOCs (Zhi-Lin *et al.*, 2012).

The genus *Muscodor* is classified into the family *Xylariaceae*. The first species, Muscodor albus was isolated from Cinnamomum zevlanicum in Honduras, Central America which produces bioactive volatile compounds (Worapong et al., 2001; Strobel et al., 2001; Ezra et al., 2004). The low-molecular weight volatile compounds have broad antimicrobial property, inhibiting or killing fungi and bacteria (Strobel et al., 2001). This fungus lacked spores and spore-producing structures (Ezra et al., 2004). Recently, 18 species of the genus Muscodor were isolated from diverse plants in different countries (Table 2.5). *Muscodor albus* produced strong bioactive VOCs with antimicrobial activity. These volatile compounds could be used for non-agricultural and agricultural biofumigation such as they were investigated to control common building molds (Mercier and Jiménez, 2007) and control Pythium ultimum both in vitro and in vivo (Worapong and Strobel, 2009). Furthermore, the effectiveness of *M. albus* volatile to control pathogens was supported by the composition of the substrates (Ezra and Strobel, 2003). Desiccated rye grain culture was found similar to fresh culture for control of *Penicillium expansum* at 21°C for 24 h in postharvest fumigation and isobutyric acid was the fungicidal agent (Mercier et al., 2007). Biofumigation by volatiles metabolites from Muscodor was developed as a postharvest treatment. The first demonstration of biofumigation to control fungal decay was placed fresh rye culture of M. albus in closed plastic boxes with fruit samples (Mercier and Jiménez, 2004). It showed significant results to control blue mold and gray mold of apple, including brown rot of peaches (Mercier and Jiménez, 2004). Furthermore, M. albus volatile was used to control barley seed contamination with Ustilago hordei (Pers.) Lagerh. It showed 100% growth inhibition of U. hordei in the plant grown from the seed that fumigated with M. albus for 4 days (Strobel et al., 2001).

Soil-borne disease affects crop production and it often used fungicide to control pathogens. Volatile-producing endophytic fungi may be an alternative method as biofumigation in control of soil-borne disease. For example, solid-state culture of *M. albus* infested artificially to soil or potting mix for control *Verticillium* wilt of eggplants (Stinson *et al.*, 2003a), *Phytophthora* root rot of pepper (Mercier and Manker, 2005). Including, another species of *Muscodor*, *M. cinnamomi* displayed potential biofumigation to control *Rh. solani* AG-2 damping-off disease (Suwannarach *et al.*, 2012).

2.4 Plant growth promoting endophytic fungi

The fungi are considered to promote plant growth directly or indirectly. The common characteristic include production of plant growth promoting secondary metabolites such as auxin, gibberellin, cytokinin, ethylene, including siderophore (Ahmad et al., 2005; Hamayun et al., 2010). The ability to promote plant growth of fungal endophytes may be due to capacity to produce higher amounts of growth promoting regulators (Hamayun et al., 2010). The one of the most physiologically active auxins is indole acetic acid (IAA). It is a common product of L-tryptophan metabolism by plant growth promoting endophytic fungi. In addition, endophytes could enhanced plant growth by uptake of nutritional elements such as N and P (Ryan et al., 2008; Kumar et al., 2011). Several endophytic fungi were produced plant growth promoting metaboites such as *Heteroconium chaetospira*, the root endophyte significantly increased biomass of Chinese cabbage due to nitrogen transfer (Usuki and Narisawa, 2007). Phoma glomerata and *Penicillium* sp., which isolated from roots of cucumber could produce gibberellins and IAA to significantly promote the growth of Gas-deficient rice (Waqas et al. 2012). Endophytic fungus, Phoma spp. from Tinospora cordifolia and Calotropis procera could promote the growth of Zea mays seedlings in pot experiments (Kedar et al., 2014). Absidia and Cylindrocladium, endophyes from rice showed significant increases in rice height, fresh and dry weight (Atugala and Deshappriya, 2015).

> ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

Taxon	Host	Collection site	Major compounds	Bioactivity
<i>M. albus</i> $cz620^1$	Cinnamomum zeylanicumin	Honduras	2-methylpropanoic acid;	antifungal, antibacterial
		20 00	naphthalene and azulene	
	5		derivative	
M. vitigenus ²	Paullinia paullinoides	Lake Sandoval, Peru	naphthalene	anti-insect
$M. roseus^3$	Grevillea pteridifolia	North of Australian	ethyl 2-butenoate and 1,2,4-	antifungal
	204	1200	trimethylbenzene	
<i>M. crispans</i> ⁴	Ananas ananassoides	Bolivian Amazon	2-methylpropanoic acid	antifungal, antibacterial
		Basin		
<i>M. yucatanensis</i> ⁵	Bursera simaruba	Yucatan Peninsula,	1 <i>R</i> ,4 <i>S</i> ,7 <i>S</i> ,11 <i>R</i> -2,2,4,8-	phytoinhibitory
	E	Mexico	tetramethyltricyclo	activity
		G, Good	[5.3.1.0(4,11)]undec-8-ene	
M. fengyangensis ⁶	Actinidia chinensis	Southeast of China	2-methylpropanoic acid	antifungal, antibacterial
M. cinnamomi ⁷	C. bejolghota	Chiang Mai, Thailand	ethyl 2-methylpropanoate	antifungal, antibacterial
M. sutara ⁸	Prestonia trifidi	Columbian tropical	2-Methylpropanoic acid	antifungal
	adalib	Pacific rainforest	1010001111	
M. kashayum ⁹	Aegle marmelos	West Ghats of India	3-Cyclohexen-1-ol, 1-(1,5-	antifungal, antibacterial
	All r	ights r	dimethyl-4-hexenyl)-4-methyl-	
<i>M. equiseti</i> ¹⁰	Equisetum debile	0	2-methylpropanoic acid	antifungal, antibacterial

Table 2.5 Major volatile compounds and bioactivity compare with related *Muscodor* spp.

 Table 2.5 (continue)

Taxon	Host	Collection site	Major compounds	Bioactivity
$M. musae^{10}$	Musa acuminata	Chiang Mai, Thailand	2-methylpropanoic acid	antifungal, antibacterial
$M. \ oryzae^{10}$	Oryza rufipogon	Chiang Mai, Thailand	3-methyl-butan-1-ol	antifungal, antibacterial
M. suthepensis ¹⁰	C. bejolghota	Chiang Mai, Thailand	2-methylpropanoic acid	antifungal, antibacterial
<i>M. strobelii</i> ¹¹	C. zeylanicum	South of India	4-octadecylmorpholine	antifungal, antibacterial
<i>M. darjeelingensis</i> ¹²	C. camphora	Northeast of	2, 6-Bis (1, 1-dimethylethyl)-4-	antifungal, antibacterial
	225	Himalayas, India	(1-oxopropyl) phenol	
<i>M. tigerii</i> ¹³	C. camphora	Tiger Hill area of	4-octadecylmorpholine	antifungal, antibacterial
		Darjeeling, India		
M. ghoomensis ¹⁴	C. camphora	Darjeeling, West	4-octadecylmorpholine	antifungal, antibacterial
	1 E	Bengal, India		
M. indicus ¹⁴	C. camphora	Darjeeling, West	1, 6-dioxacyclododecane-7,	antifungal, antibacterial
		Bengal, India	12-dione and 4-octadecyl morpholine	

Worapong *et al.*, 2001¹; Daisy *et al.*, 2002²; Worapong *et al.*, 2002³; Mitchell *et al.*,2008⁴; Gonzalez *et al.*, 2009⁵; Zhang *et al.*, 2010⁶; Suwannarach *et al.*, 2010⁷; Kudalkar *et al.*, 2011⁸; Meshram *et al.*, 2013⁹; Suwannarach *et al.*, 2013a¹⁰; Meshram *et al.*, 2014¹¹; Saxena *et al.*, 2014¹²; Saxena *et al.*, 2015¹³; Meshram *et al.*, 2015¹⁴

Copyright[©] by Chiang Mai University All rights reserved

17

2.5 Para rubber tree and natural rubber

The rubber tree or pará rubber tree (*Hevea brasiliensis* Müll.Arg.) is an economic crop that produces natural rubber as a major raw material for rubber manufacture. It is a medium to large-sized plant and native plant in rainforests of the Amazon region of South America including Brazil, Venezuela, Ecuador, Columbia, Peru and Bolivia. Recently, most of the world' s plantation areas of rubber trees are located in countries in South East Asia such as Malaysia, Indonesia and Thailand (van Beilen and Poirier 2007). The genus *Hevea* is belong to the family *Euphorbiaceae*. There are 3 species, *H. brasiliensis*, *H. guianensis* and *H. benthamiana* could produce natural rubber. But, *H. brasiliensis* is the only species planted commercially viable quantities of high-quality rubber (30–50% by volume of fresh latex) (Ko *et al.*, 2003).

Description: *H. brasiliensis* is a deciduous tree, the plantation tree usually 15-25 m tall with a leafy crown and typically 30-40 tall in the wild (Figure 2.3A). The leaves are in spirals with 3 leaflets. The leaf stalk (petioles) usually 7.5-10 cm long and leaf shape is obovate (Figure 2.3B). The flowers are small with no petals. They are either male or female flower but both are found in the same tree (monoecious). The flower color is bright or cream-yellow. The bark is pale to dark brown with a smooth surface and inner bark is pale brown with white or cream colored latex. The fruit is an exploding 3-lobed capsule, usually 5-6 cm in diameter with gray-brown color (Figure 2.3B).



Figure 2.3 Rubber tree plantation (A) Leaves and seeds of *Hevea brasiliensis* (B).

The rubber seeds propagated and released explosively up to 15 m from the parent plant and can reach a height of over 44 m in the wild (Dickson *et al.*, 2011). Rubble tree in plantation is small for 2 reasons. First, it grow more slowly when latex was tapped from the rubber tree. Second, latex production of rubber tree declines after 30 years then the trees are cut. The original crop of rubber tree is in tropical and humid climate areas. However, it can be found in deeply flooded land, acidic boggy sites and drained upland (Gazis, 2012). For optimum growth and high productivity, it required temperature ranging from 26-30 °C and a fairly-distributed rainfall \geq 1,250 ml per year. Rubber tree can grown on a board range of soil type but the soil should contained the major nutrients such as organic matter (N) > 4.5%, phosphorous (P) > 20 ppm and potassium (K) > 150 ppm. Moreover, it requires a fairly deep surface soil of at least 1 m and pH 5-7. Flat or level lands are good for rubber planting. Rubber tree plantation spans from South American to Southeast Asia that is rubber tree has an exceptional adaptability to a variety of an environmental condition. However, clones adaptation do not always present the same performance in different regions (Martins and Zieri, 2003).

Seedling production of rubber tree concerns rootstock growth in nurseries, stalk formation and grafting operations. Seeds are utilized mainly for rootstocks production (Cardinal et al., 2007). High quality seeds produce healthy rubber tree seedlings. In addition, rootstocks from hybrid cultivars seeds are very healthy than collected seeds in native areas. In rootstock formation, the seeds are germinated to small seedlings that call "spider paw" then they are transferred to the nurseries or plastic bags contained artificial substrate for seedling productions. In stalk formation, clonal gardens have the objective to obtain plant buds for grafting. They have time period estimate 5 years with an objective obtain "brown stalk" buds and 12 years to obtain "green stalk" buds. In grafting operation, the selection of "green stalk" bud is essential for success in grafting because bud formation is rapidity and longevity in the clonal garden. Then, vigorous grafted seedlings are ready to transplant in the field (Cícero and Marcos-Filho, 2010). The previous study showed there were no significant differences in the anatomy and morphology of the leaf from rubber tree clone, GT1 compare with RRIM 600 clone which grafted on the same rootstock and same condition. However, the clone GT1 presented cell thickness in palisade parenchyma and amount of sclerenchyma fibers greater than the RRIM 600 clone (Martins and Zieri, 2003).

The natural rubber (cis-1, 4-polyisoprene) is a secondary metabolite which the producing plant uses natural rubber to protect it from insect and disease (Kaewchai, 2013). It is produced in special cells called laticifers and located inside the phloem. When the bark is tapped, the cytoplasm of laticifer cells are explelled in the form of latex. It contains 30-50% (w/w) of natural rubber which is a raw material in commercial defense and transportation industries (Ko *et al.*, 2003). The primary product of rubber are used for the automotive, shipping, agriculture, chemicals, pharmaceutical and consumer industries. Secondary product is lumber and latex products such as globes, vehicle tires, medical tubing and footwear. In addition, the seeds were used as food by Amazonian people and used as an alternative to petroleum fuel (Gazis, 2012). Latex extraction and harvesting is possible from 6-7 years after crop and it is the primary source of natural rubber. Rubber tree produce latex until 25-30 years.

In 2005, the large-scale plantation areas located in Southeast Asia and Africa (Gazis and Chaverri, 2010). About 90% of natural rubber in foreign markets is produced in Southeast Asia. Thailand is the leader of the world to produce and export the rubber products with product capacity of 3.1-3.2 million tons per year. In 2009, rubber plantations in Thailand covered 2.7 million ha. South of Thailand is the largest rubber plantation areas cover 2.1 million ha (77.8%) and the remaining 0.60 million ha (22.8%) are plated in the areas in Northeast, North and Central region, respectively (Saengruksawong *et al.*, 2012). In 2012, rubber plantation areas in Thailand increased to 22.2 million ha and plantation area in South of Thailand increased to 13.9 million ha.

1112001

2.6 Endophytic fungi in para rubber tree

Natural rubber is synthesized by many plant species belonging to 300 genera but *H. brasiliensis* produce the best quality with 90% of the world's natural rubber production (Gazis and Chaverri, 2010). Previous report showed diversity of endophytic fungi isolated from leaves and sapwood of *H. brasiliensis* grown in Peru, one native planted area. The most frequently isolated were the genera belong to phylum Ascomycota, *Penicillium*, *Pestaloiopsis* and *Trichoderma* (Gazis and Chaverri, 2010). Different endophyte species were dominated in different parts of the plant. Leaf samples display the great number of endophyte species because they have less infection barriers when compared with sapwood (Arnold and Lutzoni, 2007). Commonly, the genera that often found in *H*.

brasiliensis were Alternaria, Botryosphaeria, *Colletotrichum*, Fusarium, Pestalotiopsis/Pestalotia and Xylaria (Cannon and Simmons 2002; Gamboa et al. 2002; Santamaria and Bayman, 2005). But, Gazis and Chaverri (2010) found the other genera that are not commonly isolated from *H. brasiliensis* such as *Corallomycetella*, Fimetariella, Perisporiopsis, Rubrinectria and Trichoderma. Moreover, the new endophyte spececies, T. amzonicum was found in sapwood and leaves of H. brasiliensis and H. guianensis (Chaverri and Gazis, 2011) while Trichoderma species such as T. cacao and T. gileri have been isolated from sapwoods and T. harzianum was isolated from leaves and sapwoods (Gazis and Chaverri, 2010). However, a diversity study of the genus Muscodor that lives in association with rubber trees has not been reported previously.

2.7 Rubber tree disease

Although recent farming practices for the rubber tree have achieved a high yield of latex, plant diseases that cause reduction of latex yield and mortality in rubber trees are still common (Jayasuriya and Deacon 1995; Evueh and Ogbebor 2008). Early detection of disease helps prevent their spread and damage in nurseries and plantations.

2.7.1 Leaf disease

2.7.1.1 Leaf and seedling blight caused by *Fusarium oxysporum* f. sp. *vasinfectum*.

The fungus usually cause vascular wilt of cotton (Hering *et al.*, 1999) but also infects to another plant leaf and seedling including the rubber tree. The pathogen can survive for many years in the form of chlamydospore in plant debris or the soil (Moricca *et al.*, 1998). Culture characteristic is cottony whitish to purple (Figure 2.4A), conidiophores simple and branching. Macroconidia, several-celled slightly curved or bent at the pointed ends, multiseptate, typically canoeshaped, hyaline. Microconidia, single-cell in chains and clusters. It infects both seedling and trees in plantations. After infection, the symptom shows irregular brown lesions on leaf surface or seedling stem (Figure 2.4B) and branch with a pinkish, shiny, cottony growth on the infected area.



Figure 2.4 Colony of *Fusarium oxysporum* f. sp. *vasinfectum* on PDA (A) symptom of seedling blight (B) (Tangonan *et al.*, 2008).

2.7.1.2 Leaf and seedling blight caused by Rhizoctonia solani Kuhn.

The fungus infects start from the tip of the leaf with a chlorotic lesion. It is easily seen mycelial growth at the lower part of the leaves. The symptom shows discoloration and rotting of vascular tissue (Figure 2.5). In addition, seedling blight of rubber caused by *Rh. solani* associated with the presence of nematodes that infected stem of budded rubber seedlings.



Figure 2.5 The symptom of Leaf and seedling blight caused by *Rhizoctonia solani* Kuhn (Tangonan *et al.*, 2008).

2.7.1.3 Seedling blight caused by *Phytophthora* spp.

The fungus infects the rubber in nurseries. Symptom shows that lower portion of the plant have discolored tissue around the base. Brown strips that extend to the last bud are observed when cut. The pit shrink, then plants dies.

2.7.1.4 Anthracnose leaf spot and leaf blight

This disease is caused by *Colletotrichum gloeosporioides*. The fungus produces 10-15 μ m x 5-7 μ m conidia with hyaline, one-celled, ovoid to oblong, slightly curved or dumb-bell shape. The spore masses showed pink or orange colored. The infection start from the edge of the leaf and spread to the center. Inside infected tissue appear the waxy acervuili, typically with setae and simple, short, erect conidiophores. The disease affects mainly young budded plants, seedling and young trees. The symptom shows brown and irregular-shaped lesions on leaf surface. Moreover, the infected lesions merge to form concentric lines and then malformation occurs (Figure 2.6).



Figure 2.6 Anthracnose leaf spot disease on rubber tree leaves (Tangonan et al., 2008).

2.7.2 Stem and branches diseases
 2.7.2.1 Die back disease
 Various species of fungi such as *Phytophtora*,

Various species of fungi such as *Phytophtora*, *Colletotrichum* and *Botrydiplodia* are caused this disease. The progressive death of branches start from the terminal bud to the main stem or trunk, then the whole tree is dead. It is observed by leafless branches at the upper part of the canopy (Tangonan *et al.*, 2008).

2.7.3 Root disease

2.7.3.1 White root rot disease

It is the most serious problem of rubber tree plantation and causing slow death in all stages from seedling to mature trees (Tangonan *et al.*, 2008). This disease is present in Congo, Cote d'Ivorire, India, Malasia, Nigeria, Sri Lanka and Thailand (Rajalakshmy and Jayarathnam, 2000; Omorusi, 2012; Kaewchai, 2013). White root rot is responsible for yield losses of up to 50% in rubber plantations in West Africa (Ogbebor *et al.*, 2013). In Thailand, previous report showed that 55% disease trees in rubber plantations located in Phangnga province are white root rot disease (Kaewchai, 2013). *Rigidoporus microporus* (Syn. *R. lignosus, Fomes lignosus*) is major causing agent of this disease by decay lignin and polysaccharide in root tissues (Geiger *et al.*, 1986). The process of disease infection are 3 stage; penetration, colonization and degradation. The fungus penetrates and colonizes root tissues. After degrade the host cell, the mycelium repeatedly carry out penetration and colonization of the host cell wall. Fungal mycelium can observed as intra and inter interactions of cell wall. The pathogen produces range of cell wall degrading enzymes such as CM-cellulase, pectinase and laccase, which correspond to the diverse polymers in host plant cell wall (Omorusi, 2012).

The disease detection is difficult at the early stage because symptoms of root disease in the above ground level are different from the below ground level. In early stage, the symptom showed slightly off-green leaves, then the green leaves in unaffected branches turn to yellow and the whole canopy is destroyed, it is possible indication of root disease (Tangonan *et al.*, 2008) (Figure 2.7A). Typical symptom of the disease is the present of rhizomorphs on infected roots with white color but becomes yellow or reddish brown as they grow old. The rhizomorphs are flattened mycelium with 1-2 mm thick which grow firmly attached to the root or tree trunk surface (Omorusi, 2012) (Figure 2.7B). In addition, fungal rhizomorphs with whitish or yellowish colored grown on top or upper layers of the soil and decaying leaves and branches around the base of tree trunks that are sources of fungal inoculum.



Figure 2.7 Early stage of white root rot disease (A), typical symptom of the disease is the present of rhizomorphs (B), fruiting body of *Rigidoporus microporus* (C) (Kaewchai, 2013) and mycelial growth on potato dextrose agar (PDA) (D).

Fruiting body commonly grow at the collar or near the disease tree base and on decaying roots or stumps (Figure 2.7C). It looks like bracket fungus attached to host. The growth and development of *R. microporus* often occur during rainy season and can survive both in living hosts and nonliving substrate. In the disease cycle, the infection and colonization of *R. microporus* occur in root system of young rubber trees which grow from stumps of infected trees referred to as primary inoculums. The other aspect is the progress of the *R. microporus* along roots from infected trees as secondary inoculums toward healthy rubber trees (Figure 2.8). Disease development and the death of infected trees are most rapid between the first few years after planting.



2.7.3.2 Brown root rot disease

Phellinus noxius, a fungus which require acidic, high temperature and humid conditions. It can infect root in several trees of all ages and health conditions and has a variety of host plant worldwide (Ann *et al.*, 2002). The fungus produces a white colony which turns brown on PDA at the optimal temperature, 30 °C (Figure 2.9A). Arthrospores and staghorn-like hyphae are also observed in culture

(Figure 2.9B-C). The vegetative hyphae have no clamp connections. *P. noxius* is one of species that are strong virulence while another species act as saprobe in nature or weak pathogens on trees.

The rhizomorphs in the infected roots is tawny brown, becoming dark color with age. The infected roots develop a very rough and irregular surface. The symptom showed pale brown root rot in early stage, then brown zigzag line display in the wood and brown line network can also be seen on the wood surface. The disease tree exhibit reduced growth, discoloration and wilting of leaves and dieback of branches, then whole plant eventually dies (Sahashi, 2013). However, leaf and root symptoms are unique and different from the other root rot pathogen, *R. microporus* and it degrades the polysaccharide fraction but not lignin (Kirk, 1971). The fruiting body is rarely observed on disease trees. It is dark brown bracket fungus while the underside is dark gray.



Figure 2.9 Colony of *Phellinus noxius* on PDA (A), staghorn-like hyphae (B) and arthrospores (C).

2.7.3.3 Red root rot disease

The disease is caused by *Ganoderma pseudoferreum*. It produce red shiny mycelium cover infected root. The growing edge of the fungus is commonly white while the red colors is formed only a few centimeter. The rot is pale brown in the early stage, then advance stage, the fungus produces fruiting body. It is hard woody bracket with dark reddish brown but the upper surface is wrinkle with ash white under the bark of the root.

2.8 Management of rubber tree disease

2.8.1 Chemical control

The application of chemical to control the disease was done by spraying for leaf disease and soil drenching for root rot disease. However, it is unworkable and costly due to the tree height and the wide area. Chemical control affect to environment, animal and human health. Moreover, several phytopathogenic fungi have developed resistance to chemical fungicides (Kim and Hwang, 2007). The chemicals such as mancozeb, hexaconazole, difenoconazole and tridemorph (calixin) were applied to control root rot disease (Tangonan et al., 2008). The application dosage follow by the manufactures but it should be applied to the root every 3 months. When detected early stage, dig the soil around the base of the tree, then scrape the fungal rhizomorphs. Drench with chemicals for the rubber diseases. Build canals around the effected trees to protect spreading the disease to healthy trees. The fungal rhizomorphs was inhibited and the treated trees showed complete recovery. However, the infected tree at advanced stage of R. microporus did not recover. It also added the powdered sulphur in the plating hole, when it is known to promote fungal growth antagonistic to inhibit fungal root disease. However, clearing methods are used to reduce root rot disease such as uprooting and poisoning of old trees. Phytophthora black strip was control by 4 effective chemicals, mancozeb, metalaxyl, oxadixyl and benomyl.

2.8.2 Biological Control

Biological control is the using of resistant varieties and the manipulation of antagonists and biological competitors (Tangonan *et al.*, 2008). It is the method avoids problems from chemical control such as the development of the chemical resistance of pathogens. Although, it can't completely control the pathogens as rapidly as chemical control but it can be an important component in the development of agriculture management system. Basidiomycete fungi was used to control the growth of *R. lignosus* under nursery and plantation. Likewise, *T. harzinum* was applied to control the develop of fungal rhizomorph on rubber plant. It gave an effect in suppressing *R. lignosus* and found to have different results with the chemical control. *Trichoderma harzianum* showed highly antagonistic by completely overgrowing the pathogen (Tangonan *et al.*, 2008). In 2010, Kaewchai and Soytong reported that the crude extract and rotiorinol, a bioactive compound produced from *Cheatomium cupreum* could inhibit mycelial growth of *R. microporus* and the application of *Ch. cupreum* in poweder and oil form showed significantly inhibit the growth of *R. microporus* in the rubber tree. In addition, *T. hamatum* STN07 and *T. harzianum* STN01 were rapidly grown over the pathogen colony. However, the application of antagonistic fungi in the field needs to develop for effective results and sustainable (Kaewchai, 2013). In 2015, Ogbebor *et al.* showed antifungal activity of *Hypocrea jecorina* to control *R. lignosus* with 86.83% growth inhibition. Moreover, *H. virens, H. jecorina, H. lixii, T. spirale* and *Trichoderma* sp. were also effective on *R. lignosus. In vivo*, seedling in *H. jecorina* treatment at 60 days and *H. virens* treatment at 150 days after inoculations, respectively had highest length of stem and length of tap rot, lowest plant death, and least foliar symtomp compared to the control.



ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

CHAPTER 3

Isolation and screening of endophytic fungi that capable of producing bioactive compounds and indole acetic acid

3.1 Introduction

At least one million species of endophytic fungi are found in the plant tissue (Dreyfus and Chapela, 1994). Natural select the evolution of beneficial endophyte strains and several endophyte are found to produce bioactive compounds that protect plant from insect pests and pathogens (Saikkonen *et al.*, 2004). They play an important role in plant defense including the function as growth promoter and enable the host survival under extreme conditions (Rosa *et al.*, 2012). Recently, the development and spread of drug-resistant pathogens are still a global problem and there is a need to search for new active agents with antimicrobial activity (Espinel *et al.*, 2001). Thus, endophytic fungi are source of novel or bioactive metabolites for pharmacological and agricultural applications (Idris *et al.*, 2013). Natural products from fungal endophyte showed antagonistic activity to inhibit several pathogenetic organism such as bacteria, fungi, viruses and protozoans. Furthermore, they could be plant growth regulators such as indole acetic acid (IAA), one of the most physiologically active auxins. IAA is a product of L-tryptophan metabolism by various microorganism including Plant growth promoting rhizobacteria (PGPR) (Ahmad *et al.*, 2005).

Plants that produce effective natural products had associated with endophytic fungi that produced the same compounds (Tan and Zou, 2001). Valuable plant products might be produced by their endophytic fungi, which were easier and more economical for large-scale production than plant sources (Strobel and Daisy, 2003). Thus, searching for active fungal endophyte, medicinal plants were a great source of endophyte based on their pharmaceutical properties.

Endophytic fungi colonized plant tissues without causing any symptoms. They were found in all terrestrial plants and healthy plants were interesting source for endophyte isolation. The large number of endophytic fungi played an important role on ecosystem with greatest biodiversity (Strobel, 2003). The biodiversity of plant species were reported in Thailand and it had a long traditional of herbal medicine (Panthong *et al.*, 1991). Thai medicinal plant might be another source of effective endophytic fungi with biological activity or produced novel bioactive compounds. The previous report showed that *Colletotrichum* sp. E5T9, an endophytic fungus from wild medicinal plants in Thailand produced asparagenase with cancer inhibitory property (Theantana *et al.*, 2009). Furthermore, *Fusarium* spp. and *Acremonium* spp. were the predominate species of endophytic fungi from *Sesbania grandiflora* (L.) Pers. with antimicrobial activity against bacteria, yeast and molds (Powthong *et al.*, 2013).

3.2 Materials and methods

3.2.1 Preparation and surface sterilization of plant materials

Fourteen healthy medicinal plant species and rubber tree were used as source for isolation of endophytic fungi. Three plants such as *Clinacanthus nutans* (Burm.f.) Lindau, *C. siamensis* Bremek and *Rhinacanthus nasutus* (Linn.) Kurz were collected from Medicinal Plant Garden, Doi Suthep-Pui National Park (18°48'22"N, 98°54'51"E, altitude 1,076 m), located in Chiang Mai Province. *Aloe vera* L., *Hippeastrum johnsonii* Bury., *Muchlenbeckia platyclada* Meissn., *Pereskia grandiflora* Haw., *Jatropha podagrica* Hook., *Schefflera leucantha* R. Vig. and *Citrus hystrix* DC. were collected in different areas from Chiang Mai Province. *Bauhinia strychnifolia* Craib and *Betula alnoides* Buch.-Ham. ex G.Don were collected from Udon Thani province (17°24'54"N, 102°47'12"E). *Pseuderanthemum palatiferum* (Nees) Radlk. was collected from Nakhon Ratchasima province (14°58 '16"N, 102°5'59"E) and *Hevea brasiliensis* Müll.Arg. was collected in planting areas from Nong Bua Lamphu province (17°12'14"N, 102°26'26"E).

Plant materials were cut randomly into small segments (leaf, 5 mm x 5 mm; branch 5 mm long). All segments were sterilized by soaking in 75% ethanol for 30 s, 2% sodium hypochlorite for 3 min, and 95% ethanol for 30 s under a laminar flow hood (Suwannarach *et al.*, 2010).

3.2.2 Isolation of endophytic fungi

Each plant materials were placed on 2% malt extract agar containing rose bengal (0.033 g/l) and chloramphenicol (50 mg/l) (Theantana *et al.*, 2007). Petri dishes were sealed with Parafilm® M (Bemis company, Inc., USA) and incubated at 25 ± 2 °C for 2 weeks. The hyphal tips were cut when the fungi growing out from the samples and transferred to potato dextrose agar (PDA). The medium was also be used for subculture and stock culture.



Figure 3.1 Colleting sites (star marks) of selected plants in Thailand where used in this study.

3.2.3 Isolation and pathogenicity test of phytopathogenic fungus

Leaf and seedling blight disease of the rubber tree were sampled from rubber tree planting area in Nong Bua Lamphu province. Stem containing lesions were cut into small pieces (5 mm long) and surface sterilized according to the method in section 3.2.1, then placed on PDA and incubated at room temperature $(25\pm2 \text{ °C})$ for 2 weeks. The hyphal tips were cut when the fungi grew out from the samples and transferred to PDA. The PDA medium was used for subculture and stock culture.

Pathogenicity tests was carried out in plastic boxes. Healthy leaf and stem of the rubber tree seedlings were surface sterilized with 70 % ethanol. Each mycelial plug of fungal isolates (6 mm diameter) were placed on leaf and stem for pathogenicity test with moistened cotton wool and incubated at room temperature (25 ± 2 °C) for 2 weeks. Then, disease symptom on leaf and stem were recorded.

3.2.4 Phenotypic and genotypic classification

Identification of endophytic fungi was carried out on the basis of morphological and molecular characteristics (Huang et al., 2008). The fungal isolates were grown on potato dextrose agar (PDA) and incubated at room temperature to observe morphological characteristic such as color, shape and size of spores. For molecular identification, the effective fungal endophytes were grown on PDA at 25±2 °C for 10 day. The aerial mycelium of each isolate was scraped from the PDA surface. This fungal biomass was then freeze-dried and ground into a fine powder with a pestle and mortar. A modified SDS-CTAB method (Suwannarach et al., 2013a) was used for the DNA extraction. The internal transcribed spacer regions 1 and 2 including 5.8S rDNA (ITS1-5.8-ITS2 rDNA) was a targeted nucleotide sequence for the amplification using a pair of universal primers (ITS4 and ITS5). The amplification was carried out using a GeneAmp 9700 thermal cycler (Applied Biosystems) with the following polymerase chain reaction (PCR) conditions: initial denaturization at 95 °C (2 min), 30 cycles of reaction [denaturization at 95 °C (30 s), annealing at 50 °C (30 s), and extension at 72 °C (1 min)], and final extension at 72 °C (10 min). The reaction mixture (25 µ L) contained 1 µ L of the DNA template, 0.2 µ M dNTP, 0.2 µ L of FastTaq (Applied Biosystems), 0.2 µ M each primer, 2.5 µ L of MgCl 2 buffer, and sterile water to bring the final volume to 25 µ L. The PCR product was confirmed by gel electrophoresis and purified with a PCR cleanup Gel Extraction NucleoSpin® Extract II Purification Kit (Macherey-Nagel, Germany) following the manufacturer's manual. The purified PCR product was then sequenced by 1st Base, Malaysia. All DNA sequences obtained were aligned and compared to the available sequences in the GenBank database using the BLASTN facility within the NCBI website (http://www.ncbi.nlm.nih.gov/BLAST/). Phylogenetic analysis of all sequences was conducted by the neighbor-joining method using MEGA6 software (Tamura *et al.*, 2013).



Figure 3.2 Plant samples in this study; A, Aloe vera; B, Clinacanthus nutans; C, Clinacanthus siamensis; D, Hevea brasiliensis; E, Citrus hystrix; F, Jatropha podagrica; G, Muchlenbeckia platyclada; H, Hippeastrum johnsonii; I, Pereskia grandiflora; J, Rhinacanthus nasutus.

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

Plant host	Collection site	Collection time
Aloe vera	Chiang Mai Province	June, 2013
Bauhinia strychnifolia	Udon Thani Province	September, 2012
Betula alnoides	Udon Thani Province	September, 2012
Clinacanthus nutans	Medicinal Plant Garden,	June, 2011
	Chiang Mai Province	
Clinacanthus siamensis	Medicinal Plant Garden,	August, 2011
S. /	Chiang Mai Province	
Hevea brasiliensis	Rubber tree planting area,	May, 2011-
204	Nong Bua Lamphu Province	December, 2012
785	and Payao Province	-785-
Citrus hystrix	Chiang Mai Province	September, 2013
Jatropha podagrica	Chiang Mai Province	August, 2013
Muchlenbeckia platyclada	Chiang Mai Province	August, 2013
Hippeastrum johnsonii	Chiang Mai Province	August, 2013
Pereskia grandiflora	Chiang Mai Province	August, 2013
Pseuderanthemum palatiferum	Nakhon Ratchasima	June, 2012
Aİİrig	Province Province	rved
Rhinacanthus nasutus	Medicinal Plant Garden,	June, 2011
	Chiang Mai Province	
Schefflera leucantha	Chiang Mai Province	June, 2013

 Table 3.1 Collection details of selected plants in this study.

3.2.5 Screening for antagonism against rubber tree pathogens by dual culture method

Antagonistic activity of all fungal endophytes was tested against a set of pathogenic fungi listed in Table 3.2. The antagonistic analysis was observed from the interaction between fungal endophytes and the pathogens using dual culture technique. An agar plug (6 mm diameter) 4-day-old fungal endophytes growing on PDA was inoculated on PDA part of the petri dish, then an agar plug of fungal pathogens was inoculated opposite side of the petri dish. The control plates were inoculated with either pathogens. All Petri dishes were wrapped with Parafilm® M (Bemis company, Inc., USA) and incubated at room temperature (25 ± 2 °C) for 5 days. The percentage of inhibition of fungal growth after the dual culture test was calculated with the following equation: [(R1-R2)×100] ÷ R1 , where R 1 was the average colony radius of each tested fungi measured in the control plates (without the tested fungi), and R 2 was the average colony radius that calculated from the tested plates.

Pathogenic fungi	Source		
Colletotrichum gloeosporioides	The Sustainable Development of		
Fusarium oxysporum f. sp. vasinfectum	Biological Resources (SDBR) Lab,		
Rhizoctonia solani AG-2	Department of Biology, Faculty of		
	Science, Chiang Mai University, Chiang		
ລິມສິກຊິ້າເຮດດິກ	Mai, Thailand		
Phellinus noxius BCC26237	The BIOTEC (National Center for		
Phytophthora parasitica BCC15560	Genetic Engineering and Biotechnology)		
All rights	Culture Collection (BCC), Bangkok,		
	Thailand.		
Rigidoporus microporus	Department of Microbiology and		
	Parasitology, Faculty of Science,		
	Naresuan University, Phitsanulok.		

Table 3.2 The list of pathogenic fungi used in this study.

3.2.6 Crude extract preparation

The crude extract of effective endophytic fungi was prepared according to the modified solid state fermentation method of Sharmin, *et al.* (2013). Fungal endophytes were grown on PDA at room temperature $(25\pm2 \text{ °C})$ for 5 days. Three pieces $(0.5x0.5 \text{ cm}^2)$ of mycelial plugs were inoculated on PDA at room temperature $(25\pm2 \text{ °C})$ for 7 days. The medium from all the petri dishes were cut into small pieces and collected in beakers containing ethyl acetate (1:1 v/v) for 3 days. The filtrates was extracted three times with ethyl acetate and evaporated to dryness using a rotary vacuum evaporator. Then, the extract was resuspended with 50% methanol (MeOH).

3.2.7 Bioassay of crude extract by Agar well diffusion method

The crude extracts were dissolved in 50% methanol at the concentration of 5 and 10 mg/ml. An agar plug of pathogenic fungi was placed on the center of PDA plates. Then, 35 μ l of each resuspended fungal supernatant was separately loaded into 0.6 mm diameter wells cut around the mycelial plug of pathogenic fungus. The plates were incubated at room temperature (25±2 °C) for 7 days and the results were recorded and calculated the percentage of inhibition of fungal growth according to the method in section 3.2.4. Tridemorph (2, 6-Dimethyl-4-tridecylmorpholine) and carbendazim (Methyl *1H*-benzimidazol-2-ylcarbamate) were used as fungicide to control *R. microporus* and *P. noxius*, and *F. oxysporum* f. sp. *vasinfectum*, respectively. In addition, 50% methanol was used as negative control.

3.2.8 Determination of minimum inhibitory concentrations (MICs) of crude extract

The crude extracts were determined their MIC by using 2-fold serially diluted crude extract from 0.04 to 5.00 μ g/ml with 50% methanol. Then, 35 μ l of each concentration was loaded into 0.6 mm diameter wells that were cut around the mycelial plug of pathogenic fungus. The plates were incubated at room temperature (25±2 °C) for 7 days. The lowest concentration that inhibited mycelial growth was recorded as the MIC. The tests were conducted in three replicates and SPSS program version 17.0 for Windows was used to analyze the experimental data with one-way analysis of variance (ANOVA).

3.2.9 Antifungal activities assay by TLC-bioautography

Each 40 μ l of crude extract in 50% methanol (5 mg/ml) was spotted on TLC sheet (TLC aluminium sheets, 10 cm x10 cm, silica gel 60GF254, Merck, Germany). Then, it was placed in a glass tank with solvent system using 95:5 (v/v) dichloromethane: methanol as mobile phase. After running the chromatogram, TLC plate were dried and viewed under UV light at different wave length of 254 nm and 365 nm (Prakash *et al.*, 2013). The Rf values of TLC plate were calculated.

Each active compound was examined for antifungal activity by performing bioautography coupled with thin-layer chromatography (TLC-bioautography). The modified bioautography method was performed according to the method that described by Bhavya *et al.* (2011). Six mycelial plugs (6 mm diameter) of 4-day-old *P. noxius*, and *F. oxysporum* f. sp. *vasinfectum* were placed on PDA plates and incubated at room temperature (25 ± 2 °C) for 1 day. While, spore suspension of *F. oxysporum* f. sp. *vasinfectum* (1×10^6 spore/ml) was swab on PDA plate to detect the inhibition of spore germination. The chromatograms were developed, as described and were placed on PDA plate between the colonies of *P. noxius* and *F. oxysporum* f. sp. *vasinfectum*, respectively (Figure 3.3). After 1 hour diffusion process, TLC sheets were removed and incubated at room temperature (25 ± 2 °C) for 3-7 days. The inhibition zone was observed and recorded.



Figure 3.3 Antifungal activities assay by TLC-bioautography. The developed chromatogram were placed on PDA plate between the colonies of *P. noxius* (A) and *F. oxysporum* f. sp. *vasinfectum* (B), respectively.

3.2.10 Preliminary assay of hemolysis

An agar plug (6 mm diameter) of 4-day-old fungal endophytes growing on PDA was inoculated on 5% Sheep blood agar. All Petri dishes were wrapped with Parafilm® M (Bemis company, Inc., USA) and incubated at room temperature $(25\pm2 \ ^{\circ}C)$ for 7-15 days. The crude extracts were dissolved in 50% methanol at the concentration of 5 and 10 mg/ml. Then, 35 µl of each resuspended fungal supernatant was separately loaded into 0.6 mm diameter wells on 5% Sheep blood agar. The plates were incubated at room temperature $(25\pm2 \ ^{\circ}C)$ for 7 days. Type of hemolysis on blood agar was observed and recorded.

3.2.11 Screening for indole acetic acid (IAA) production

All isolates of endophytic fungi were inoculated in 5 ml Potato dextrose broth (PDB) with L-tryptophan (2 mg/ml) and incubated in the dark at room temperature $(25\pm2 \ ^{\circ}C)$ with shaking at 150 rpm on a shaker for 5 days. The broth cultures were filtrated by two layers of gauze cloth to separate the broth cultures and mycelia. A modified method described by Admad *et al.*, 2005 was used for screening IAA production. All filtrates (1 ml) were mixed with 2 ml of Salkowski's reagent (1 ml of 0.5 M FeCl₃; 50 ml of 35% perchloric acid (HClO₄)) and incubated in the dark for 30 min. The development of pink color indicated IAA production and the absorbance at 530 nm was measured. The level of IAA production was estimated by standard IAA graph (Appendix B).

3.2.12 Extraction of crude IAA and Thin layer chromatography (TLC)

The broth cultures of endophytic fungi were prepared according to the method in section 3.2.9 and filtrated by two layers of gauze cloth. Then, the filtrates were acidified to pH 2.5 to 3.0 with 1 N HCl and extracted twice with ethyl acetate at double volume of the filtrates. The ethyl acetate fraction was evaporated to dryness using a rotatory evaporator. The crude extracts were dissolved in 1 ml of methanol and spotted on TLC sheets (TLC aluminium sheets, silica gel 60GF254, Merck, Germany) with standard IAA and standard IBA (indole-3-butyric acid). Then, TLC sheets were developed in chloroform:methanol:water (84:14:1, v/v/v). Spots with Rf values identical to authentic IAA were identified under UV light at wave length of 254 nm by spraying

with Ehmann's reagent (mixed with 1 volume of 50 ml of 0.2 % (w/v) p-dimethylaminobenzaldehyde in conc. HCl added with 50 ml of absolute ethanol and 3 volumes of 2.03 g of FeCl_{3.6}H₂O in 500 ml of water and 300 ml of conc. H₂SO₄) and Salkowski's reagent (Ehmann, 1977).

3.3 Results

3.3.1 Isolation of endophytic fungi

A total of 271 endophytic fungi were isolated from 14 plant species, the most of them were isolated from *C. nutans* (24.4%), followed by *R. nasutus* (10.3%) and *C. siamensis* (8.9%), respectively (Table 3.3). Most of them colonized in leaf (53.1%) and only 10 isolates colonized in root of plant sample (Table 3.3). Some of them did not produce spore and belonged to mycelia sterilia group. While, endophytic fungi isolate JP1B1-1 and JP1B1-2 were isolated from branch of *J. podagrica* with whitish colony and fast-growing mycelium (Figure 3.4).



Plant species	Leaf		Stem/Branch	Root	Total
	Intervein	Vein	_		
Aloe vera	3*	3*	7	1	11
Bauhinia strychnifolia	3	1	5	0	9
Betula alnoides	1	2	3	0	6
Clinacanthus nutans	17	14	29	6	66
Clinacanthus siamensis	4	9	11	0	24
Hevea brasiliensis	0	3	20	0	23
Citrus hystrix	15	103-	~2	0	19
Jatropha podagrica	12	2	8	0	22
Muchlenbeckia platyclada	8	33	0	0	11
Hippeastrum johnsonii	12	6	2	0	20
Pereskia grandiflora	3	6	0	0	9
Pseuderanthemum	3	5	2	0	10
palatiferum	0	X	1 2 3		20
Rhinacanthus nasutus	0		25	3	28
Schefflera leucantha	0	9 6	4	0	13
(*) = vein absent					

Table 3.3 Number of endophytic fungi isolated from medicinal plants and the rubber tree.



Figure 3.4 Fungal isolate JP1B1-1 and JP1B1-2 grew out from branch segment of Jatropha podagrica.

3.3.2 Isolation and pathogenicity test of pathogenic fungus causing leaf and seedling blight in the rubber tree

Fusarium oxysporum f. sp. *vasinfectum* (PRT3-1) was isolated from rubber tree seeding with observation of the disease symptom. Culture characteristic is cottony whitish to purple (Figure 3.5A), conidiophores simple and branching. Macroconidia, several-celled slightly curved or bent at the pointed ends, multiseptate, typically canoeshaped, hyaline (Figure 3.5C). Microconidia, single-cell in chains and clusters (Figure 3.5D). The fungal isolate was classified and identified base on morphological and molecular characteristic. It shared relatively sequence similarity of ITS rRNA gene (100%) with *Fusarium oxysporum* f. sp. *vasinfectum* (Figure 3.6).

Phytopathogenicity test showed that *Fusarium oxysporum* f. sp. *vasinfectum* (PRT3-1) infected leaf and branch of rubber tree seedling. The symptoms showed irregular brown lesions on leaf and branch surface with a cottony growth on the infected area (Figure 3.7).



Figure 3.5 Colony of *Fusarium oxysporum* f. sp. *vasinfectum* (A), symptom of seedling blight (B), macroconidia (C) and microconidia (D).



Figure 3.6 Neighbor-joining (NJ) phylogenetic tree of *Fusarium* spp. based on ITS rRNA sequence alignment of 13 sequences. *C. gloeosporioides* was used as an out group. The isolate PRT3-1 was proposed as *F. oxysporum* f. sp. *vasinfectum*. The codes indicated in parentheses and refered to the accession numbers of the sequences available in the GenBank database.



Figure 3.7 Pathogenicity test of *Fusarium oxysporum* f. sp. *vasinfectum* (PRT3-1) on leaf and branch of rubber tree seedling. The symptom showed irregular brown lesions on leaf (A) and branch surface (B).

3.3.3 Genotypic classification of Rigidoporus microporus

The white root rot pathogen, *Rigidoporus* spp. (Nakaew *et al.*, 2015), was classified and identified using morphological and molecular characteristics (Figure 3.8-3.9). It belongs to the Polyporales and the mycelial morphology on PDA was found to be a whitish colony with radially growing flattened mycelial growths. It also produced a slightly yellowish undersurface. This isolate (KR076793) showed 100% sequence similarity of the ITS rRNA gene to *R. microporus*. However, it has been associated with three distinctive clades corresponding to Africa (clade I), South East Asia (Clade II) and South America (Clade III) (Figure 3.9). The African clade, South East Asian clade and South American clade consisted of isolates from Nigeria, Malaysia and Peru, respectively. The phylogenetic tree dendrogram supported the node separating the isolate (KR076793) in the South East Asian clade from the African clade with 99% bootstrap support (Figure 3.9).



Figure 3.8 Characteristic of *Rigidoporus microporus*, colony on PDA (A) and hyphae (B).



Figure 3.9 Neighbor-joining (NJ) phylogenetic tree of *Rigidoporus microporus* based on ITS rRNA sequence alignment of 13 sequences. *Tyromyces canadensis* was used as an out-group. The codes indicated in parentheses and refered to the accession numbers of the sequences available in the GenBank database.

3.3.4 Screening for antagonism against rubber tree pathogens by dual culture method

Based on the results, 7 isolates of endophytic fungi could inhibit the mycelial growth of fungal pathogens by showing of inhibition zone. Only 3 isolates showed the percentage of inhibition above 50%. Isolate AL1T1 showed the greatest percentage of inhibition of mycelial growth with *C. gloeosporioides* (60.0 %), followed by *P. noxius* (59.4%) and *R. microporus* (58.3%), respectively (Figure 3.10 and Table 3.4). Isolate AL1T2 showed the greatest percentage of inhibition of mycelial growth with *F. oxysporum* f. sp. *vasinfectum* (65.0 %), followed by *P. noxius* (60.7%) and *R. microporus* (54.4%), respectively (Figure 3.9 and Table 3.4). In addition, isolate JA1B1-2 showed highest antifungal activity by overgrowth of all tested fungal pathogens (Figure 3.11 and Table 3.4).
Plant	Isolate		Р	ercentage of inhibi	tion of microbial g	rowth	
	number	R. microporus	P. noxius	F. oxysporum f. sp. vasinfectum	C. gloeosporioides	Ph. parasitica	Rh. solani
Aloe vera	AL1T1	58.3	59.4	-	60.0	20.4	51.0
	AL1T2	54.4	60.7	65.0	44.0	42.0	40.5
	AL1R2	53.3	52.4	25.0	48.0	32.5	44.5
Bauhinia strychnifolia	BS1B201	32.6	-	-	-	-	-
Clinacanthus nutans	C4V202-1	20.0	-	-	19.0	-	-
	C4R301	20.0	-	-	14.5	-	-
Schefflera leucantha	SC1V1	57.8	54.7	40.0	56.4	40.0	60.0
Hippeastrum johnsonii	HI1V3	70.2*	78.3*	67.2*	50.0*	63.8*	44.5*
	HI5IV1-1	65.5*	32.4*	56.4*	45.0*	57.5*	55.3*
	HI5IV1-2	63.1*	32.4*	55.3*	45.4*	56.7*	45.6*
Pseuderanthemum palatiferum	PS1IV102	37.1	-	-	-	-	-
Jatropha podagrica	JA1B1-1	51.8*	61.1*	28.6*	47.4*	41.5*	44.2*
	JA1B1-2	82.1*	81.5*	71.4*	60.4*	80.8*	65.9*
	JA1B1-6	53.6*	55.6*	28.6*	42.1*	53.7*	53.8*
(*) = Inhibition of pat	hogen by overg	rowth of pathoge	n.	ing mai ui	iiversity		

Table 3.4 Antifungal activity of endophytic fungi from medicinal pla	int.
--	------

(-) = No visible sign of inhibition of pathogenic fungi.



Figure 3.10 Preliminary testing for antifungal activity of endophytic fungi (isolate AL1T1 and AL1T2) against plant pathogenic fungi by dual culture technique.



Figure 3.11 Antifungal activity of endophytic fungus (isolate JA1B1-2) against *Rigidoporus microporus, Phellinus noxius, Phytophthora parasitica* and *Rhizoctonia solani* AG-2 by overgrowth of all tested fungal pathogens.

3.3.5 Bioassay of crude extract by agar well diffusion method

Two isolates of endophytic fungi, AL1T1 and AL1T2 showed antagonistic activity with high percentage of inhibition against almost tested fungi especially, fungal pathogens caused of root rot disease (*R. microporus* and *P. noxius*) and leaf and seedling blight disease (*F. oxysporum* f. sp. *vasinfectum*) in the rubber tree. They were cultured and extracted for their active metabolites. The crude extract of AL1T1 at the concentration of 10 mg/ml and 5 mg/ml inhibited mycelial growth of *P. noxius* as $65.8\pm5.7\%$ and 63.3 ± 3.8 , respectively (Figure 3.12 and Table 3.5). It inhibited mycelial growth of *R. microporus* as $57.7\pm1.8\%$ and $55.8\pm3.0\%$, respectively. While, the crude extract of AL1T2 at the concentration of 10 mg/ml and $51.8\pm3.0\%$, respectively. While, the crude extract of AL1T2 at the concentration of 10 mg/ml inhibited mycelial growth of *F. oxysporum* f. sp. *vasinfectum*, *P. noxius* and *R. microporus* as $63.5\pm1.8\%$, $57.5\pm4.3\%$ and $54.7\pm2.3\%$, respectively (Figure 3.12 and Table 3.5). In addition, mycelial growth of *P. noxius* under light microscope showed many swollen mycelia and staghorn-like hyphae (trichocysts). These structures were occurred abundantly at colony edge (Figure 3.13).

Table 3.5 Percentage of inhibition	of mycelial	growth	after	treated	with	crude	extract of
AL1T1 and AL1T2.		K I		10	°/		

Fungal pathogens	Percentage of inhibition of mycelial growth						
-	AL1	IT1	AL1T2				
_	10 mg/ml	0 mg/ml 5 mg/ml 10 mg/ml 5 mg/m 7.7±1.8 55.8±3.0 54.7±2.3 50.7±					
R. microporus	57.7±1.8	55.8±3.0	54.7±2.3	50.7±1.8			
P. noxius	65.8±5.7	63.3±3.8	57.5±4.3	55.0±4.3			
F. oxysporum f. sp.	-	-	63.5±1.8	58.4±1.8			
vasinfectum							

(-) = No visible sign of inhibition of pathogenic fungi.



Figure 3.12 Bioassay of crude extract by agar well diffusion method. Mycelial growth of *Phellinus noxius* and *Fusarium oxysporum* f. sp. *vasinfectum* after treatment with crude extracts of AL1T1 and AL1T2 at different concentration, respectively (A-B). Mycelial growth of *P. noxius* and *F. oxysporum* f. sp. *vasinfectum* after treatment with tridemorph and carbendazim, the fungicides used as the control *P. noxius* and *F. oxysporum* f. sp. *vasinfectum*, respectively (C-D).

 AI UNIVERSI

 âdânấuk 10

 hái UNIVERSI

 âdânấuk 10

 Copyright[©] by Chiang Mai University

 A I I rights reserved



Figure 3.13 Mycelial growth of *Phellinus noxius* under light microscope. Mycelial growth on PDA in normal growth condition (A). Mycelial growth on PDA after treatment by crude extract of AL1T2; many swollen mycelia were occurred at colony edge (B) and staghorn-like hyphae (trichocysts) were occurred abundantly at colony edge (C-D).

3.3.6 Determination of minimum inhibitory concentrations (MICs) of crude extract

The active crude extracts were determined for their MIC. The MIC values of AL1T1 and AL1T2 ranged from 0.16-0.62 mg/ml. Crude extract from AL1T1 significantly inhibited *P. noxius* with MIC value of 0.31 mg/ml (P<0.05) (Figure 3.14 and Table 3.6). While, crude extract from AL1T2 significantly inhibited mycelial growth of *P. noxius* and *F. oxysporum* f. sp. *vasinfectum* with MIC value of 0.62 mg/ml and 0.16 mg/ml, respectively (P<0.05) (Figure 3.14 and Table 3.6). In addition, tridemorph and carbendazim showed MIC value at 0.62 mg/ml and 0.31 mg/ml, respectively (Table 3.6).

Concentration		_					
of crude extract and fungicide		P. noxiu	5	F. oxysporum f. sp. vasinfectum			
(mg/ml)	AL1T1	AL1T2	Tridemorph ¹	AL1T2	Carbendazim ²		
Control	4.0±0.1a	4.0±0.1a	4.5a	3.9±0.1a	3.9±0.1a		
5	1.9±0.1b	1.9±0.3b	1.5b	1.4±0.2b	1.7±0.1b		
2.50	2.0±0.1b	1.9±0.1b	2.0±0.1c	1.6±0.3b	2.0c		
1.25	2.0±0.1b	2.0±0.5b	3.2±0.1d	1.8±0.1b	2.1±0.1c		
0.62	2.1±0.1b	2.3±0.4b	3.1±0.1d	1.9±0.1b	2.8±0.1d		
0.31	2.8±0.3b	3.2±0.4a	4.4±0.1a	2.1±0.1b	3.2e		
0.16	3.7±0.2a	3.5±0.2a	4.4±0.1a	2.3±0.2b	3.8±0.1a		
0.08	3.8±0.1a	3.8±0.1a	4.5a	3.5±0.6a	3.8±0.1a		
0.04	3.8±0.1a	3.6±0.1a	4.4±0.1a	3.7±0.6a	3.9±0.1a		

Table 3.6 Determination of minimum inhibitory concentrations (MICs) of crude extract

 and commercial fungicide.

All tests were performed in triplicate, the radial growth is shown with mean \pm SD. Means with different letters are significantly different according to Tukey's Test (P < 0.05). Two commercial fungicides, tridemorh¹ and carbendazim² for controling *P. noxius* and *F. oxysporum* f. sp. *vasinfectum*, respectively, were used as positive control and 50% methanol was used as negative control.



Figure 3.14 Minimum inhibitory concentrations (MICs) value of crude extracts ranged from 0.16-0.62 mg/ml. AL1T1 and AL1T2 showed antifungal activity against *Phellinus noxius* (A) and *Fusarium oxysporum* f. sp. *vasinfectum* (B), respectively.

3.3.7 Antifungal activities assay by TLC-bioautography

Among 2 wave length of UV light (254 nm and 365 nm), various spots were observed and Rf value of AL1T1 and AL1T2 crude extracts were measured at 0.22 to 0.83 and 0.20 to 0.78, respectively. In bioautography of AL1T1, 4 spots at Rf 0.22, 0.28, 0.30 and 0.33 showed antifungal activity against *P. noxius* (Figure 3.15). While, in AL1T2 bioautography, 3 spots at Rf 0.20, 0.26 and 0.33 inhibited mycelial growth and spore germination of *F. oxysporum* f. sp. *vasinfectum* (Figure 3.16).

Copyright[©] by Chiang Mai University All rights reserved



Figure 3.15 TLC-bioautograph of AL1T1 crude extract against mycelial growth of *Phellinus noxius*.



Figure 3.16 TLC-bioautograph of AL1T2 crude extract against mycelial growth (A) and spore germination (B) of *Fusarium oxysporum* f. sp. *vasinfectum*.

3.3.8 Preliminary assay of hemolysis

Isolate AL1T1 grown on 5% Sheep blood agar with beta-hemolysis. The clear zone was 1.9-2.1 cm (diameter). However, no change of the medium under and surrounding crude extract of AL1T1 after tested toxicity on 5% Sheep blood agar (Figure 3.17). While, isolate AL1T2 showed non-hemolysis after grown on 5% Sheep blood agar and tested with crude extract (Figure 3.17).



Figure 3.17 Preliminary assay of hemolysis. Isolate AL1T1 showed beta-hemolysis after grown on 5% Sheep blood agar for 15 days (A). Non-hemolysis on 5% Sheep blood agar after tested with crude extract of AL1T1 and AL1T2 (B).

3.3.9 Screening for indole acetic acid (IAA) production

Twelve isolates of endophytic fungi showed their ability to produce IAA in preliminary test. The range of IAA production was $9.4\pm1.2 \mu g/ml$ to $40.7\pm0.5 \mu g/ml$. Endophytic fungus, isolate BS1B201 from branch of *Bauhinia strychnifolia* produced maximum concentration of IAA, following by isolate SC1V1 and C4V202-1, respectively (Table 3.7).

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

Plant	Isolate number	IAA (µg/ml)
Aloe vera	AL1T1	22.6±6.0
Bauhinia strychnifolia	BS1B201	40.7±0.5
Clinacanthus nutans	C4V202-1	29.4±0.2
	C4IV301	23.4±0.1
ab 9	C4R301	19.6±1.1
Jatropha podagrica	JA1B1-1	19.9±2.0
a c	JA1B1-2	14.7±2.4
1.58%2	JA1B1-6	9.4±1.2
Pseuderanthemum palatiferum	PS1IV102	16.8±0.1
Schefflera leucantha	SCIV1	33.6±5.0
Hevea brasiliensis	RTM5IV1	13.9±0.2
MI	RTM5IV3	18.8±0.2

Table 3.7 Screening for IAA production of endophytic fungi.

3.3.10 Extraction of crude IAA and thin layer chromatography (TLC)

The chromatogram showed the compounds of isolate C4IV301, C4V202, RTM5IV1, PS1IV102 and BS1B201 closed to Rf value of standard IAA and standard IBA (Appendix C). Then, they were confirmed by TLC again with different mobile phase according to the method in section 3.2.10. The culture spots, standard IAA and standard IBA showed the same result after spray with Ehmann's reagent and Salkowski's reagent. Only 3 isolates of endophytic fungi, C4IV301, PS1IV102 and BS1B201 showed the same Rf value with standard IAA and standard IBA (Figure 3.18).



Salkowski's reagent Ehmann's reagent

Figure 3.18 Thin layer chromatography (TLC) of IAA crude extract such as IAA standard, IBA standard, 1 = C4IV301, 2 = PS1IV102, 3 = BS1B201, 4 = C4V202-1, 5 = RTM5IV1 and PDB (negative control).

3.3.11 Phenotypic and genotypic classification of active endophytic fungi

Endophytic fungi isolate AL1T1 and AL1T2 colonized in stem of *Aloe vera* based on DNA analysis of ITS rRNA gene, it is suggested that both AL1T1 and AL1T2 belong to the phylum Ascomycota, order Sordariales (Figure 3.21 and Table 3.8). Colony color of isolate AL1T1 on PDA was whitish in the young age and becomes grayish in the old age (Figure 3.19A). The hyphae with septum (Figure 3.19B). Mycelium forms swollen cells (8.2-10.0 µm diameter, Figure 3.19C). Hyphae (1.0-1.3 µm thick) with coils (11.9-21.3 µm diameter, Figure 3.19D).

Colony color of isolate AL1T2 on PDA was yellowish. It produces watersoluble pigment and excretes around the colony with reddish color (Figure 3.20A). Mycelium forms swollen cell (Figure 3.20B) and produces chain of conidia (2-6 conidiospore on single conidiophore, Figure 3.20C). Spore shape is ovate ($1.3x2.6 \mu m$). Spore color is hyaline. Hyphae ($1.0-1.8 \mu m$ thick) with coils ($6.1-8.5 \mu m$ diameter, Figure 3.20D).

Isolate	Sequence	Closest related taxa in the GenBank	% Sequence
number	length (nt)	database	similarity
AI 1T1	/81	Sordariales en IX2/3802	100
ALIII	401	Solutiones sp. JA243072	100
		Sordariales sp. JX243747	99
		Sordariales sp. JX244061	99
		Sordariales sp. JX244012	99
AL1T2	465	Sordariales sp. JX243892	100
	12	Sordariales sp. JX243747	99
	12	Sordariales sp. JX244061	99

Table 3.8 Closest related taxa of AL1T1 and AL1T2 based on the sequence similarities of ITS rRNA gene.



Figure 3.19 Morphological characteristics of isolate AL1T1. Colony morphology was observed after growing on PDA for 7 days (A). The hyphal morphology of isolate AL1T1 was observed further with light microscopes. The hyphae with septum (B), the swollen hypha (C) and formation of coiling hypha (D).



Figure 3.20 Morphological characteristics of isolate AL1T2. Colony morphology was observed after growing on PDA for 7 days (A). The hyphal morphology of isolate AL1T2 was observed further with light microscopes. The swollen hypha (B), Chain of conidia (C) and formation of coiling hypha (D).

ลับสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved



Figure 3.21 Neighbor-joining (NJ) phylogenetic tree of isolate AL1T1 and AL1T2 based on ITS rRNA sequence alignment of 14 sequences. *Blumeria graminis* f. sp. *agropyri* was used as an out group. The isolate AL1T1 and AL1T2 belonged to order Sordaiales. The codes indicated in parentheses refer to the accession numbers of the sequences available in the GenBank database.

3.4 Discussion

Endophytic fungi colonized internal plant tissue without any symptom. The research about endophytic fungi expanded in recent year from isolation and identification of fungal endophyte, examining the nature of endophyte/plant interaction, including studying endophytic fungi from medicinal plant in order to discover active and novel compounds (Alvin *et al.*, 2014). Endophytic fungi are presented in most of plant section, especially healthy leaf tissue (Karunai and Balagengatharathilagam, 2014) according to this study, most endophytic fungi were more prevalent in the leaves (53.1%) than the branches/stems and the roots. However, the endophytes population was affected by the environmental conditions under which the host is growing and the endophyte profile may be is more diversified in tropical areas (Karunai and Balagengatharathilagam, 2014). Isolate AL1T1 and AL1T2 were proposed as the effective fungi with antagonistic activity against a broad range of fungi. They were isolated from a very short stem of *Aloe vera*. This medicinal plant was widely known to use in traditional medicine as a skin treatment to relive minor burns and sunburn and help heal wounds. It served as a good model plants

for studying the effects of endophytic fungi colonization on secondary metabolism (Yadav *et al.*, 2015). The previous studies showed that various fungi from leaves, stem and root of *A. vera* could produce extracellular enzyme and antagonistic activity towards microorganisms (Jalgaonwala *et al.*, 2010; Yadav *et al.*, 2015). In addition, the genus *Phoma* was isolated from *A. vera* and could produce anti-cancer agent, taxol (Rebecca *et al.*, 2011).

Root rot disease is the most serious problem of rubber tree plantation and present in many countries, including Thailand. In this study, the antagonistic test showed that the causing pathogen, *R. microporus* was susceptible to the extract of AL1T1 with the highest percentage of inhibition. While, the causing agent of brown root rot disease, *P. noxius* was susceptible to the extract of AL1T2 with the highest percentage of inhibition. However, AL1T2 showed greatest antagonistic activity to control *F. oxysporum* f. sp. *vasinfectum* according to the study of Yadav *et al.* (2015). Under the light microscope, there is swollen mycelia and staghorn-like hyphae (trichocysts) of *P. noxius* occurred abundantly at colony edge after treated by crude extract of AL1T2. It is still unknown about the function of trichocyst but it is the specialized hyphae of *P. noxius*, including arthospores and forming in pure culture (Ann *et al.*, 2002).

The root rot disease spread to the healthy tree through root-to-root contact. After the tree removal, the pathogens can survive in colonized roots remaining in the soils for many years and also remain viable as mycelium in the rhizosphere for several months (Ann *et al.*, 2002; Omorusi, 2012). Thus, the antagonistic activity of fungi to inhibit mycelial growth of the root rot pathogens was considered to control root rot disease in the field according to the previous studies such as the use of *Trichoderma hazianum* (Jayasuriya and Thennakoon, 2007) and *Chaetomium cupreum* to control *R. microporus in vivo* (Kaewchai and Soytong, 2010). Comparing to the positive control, tridemorph and carbendazim, crude extract of AL1T1 and AL1T2 showed a strong antifungal activity on the above phytopathogens, *R. micoporus*, *P. noxius* and *F. oxysporum* f. sp. *vasinfectum*. Crude extract of AL1T1 significantly inhibited *P. noxius* with MIC 0.31 mg/ml (P<0.05) which was less than MIC value of tridemorph (0.62 mg/ml). While, AL1T2 crude extract significantly inhibited *F. oxysporum* f. sp. *vasinfectum* with MIC 0.16 mg/ml (P<0.05) which was less than MIC value of carbendazim (0.31 mg/ml). The TLC-autobiography is one of the more convenient method for testing crude extract of microorganisms, plant and purified bioactive compounds against pathogenic bacteria and fungi (Prakash *et al.*, 2013). Isolate AL1T1 and AL1T2 were examined for their antifungal activity by TLC-bioautography. The results showed the separation of active compounds and Rf values that ranged from 0.22-0.33 of AL1T1 (separated using 95:5 (v/v) dichloromethane: methanol) and was found to inhibit mycelial growth of *P. noxius*. The active compounds of AL1T2 showed Rf values ranged from 0.20-0.33 and inhibited mycelial growth and spore germination of *F. oxysporum* f. sp. *vasinfectum*.

It was supported by the study of Li *et al.* (2012) who found that endophytic fungi from *Melia azedarch* had ability to produce active compounds to inhibit broad range of pathogenic fungi, including *F. solani*, *F. oxysporum* f. sp. *niveum* and *F. oxysporum* f. sp. *vasinfectum*. From TLC-autobiography assay, the extracellular fractions of both isolate were effective, which indicates that antifungal compounds from fungal endophytes are secreted extracellularly and may contribute in part to the plant protection systems against the pathogens (Abdel-Motaal *et al.*, 2010).

016191

In preliminary test of hemolysis, AL1T1 grown on 5% Sheep blood agar and produced enzyme or a substance that causes beta-hemolysis to breakdown of red blood cells. However, the crude extract of AL1T1showed non-hemolysis and should determine the general toxicity by another method to confirm the result such as the brine shrimp lethality assay, which is an effective and rapid assay to screen potential cytotoxic activity (Lu *et al.*, 2012).

Twelve isolates of endophytic fungi (4.4%) showed ability to produce IAA in preliminary test but only 3 isolates exhibited the same Rf value with standard IAA and IBA after TLC tested. Isolate BS1B201 produced maximum IAA concentration. This result was supported by the previous study of Khan *et al.* (2016) who found that endophytic fungi, *Biporalis* and *Phoma* from medicinal plant produced IAA and increased rice seed germination and growth. Furthermore, in this study, all of them showed antagonistic activity to inhibit *R. microporus* but percentage of growth inhibition was less than 50%. Endophytic fungi isolates AL1T1 and AL1T2 belonged to phylum Ascomycota in the order Sordariales based on morphological characteristic and the sequence similarities of ITS rRNA gene (99-100% sequence similarity). The

phylogenetic tree showed that isolate AL1T1 and AL1T2 closed to Sordariales sp. JX243892 with 63 % bootstrap. However, both AL1T1 and AL1T2 could not be identified at genus and species level only ITS rRNA gene was used in DNA analysis for this work. It is not enough to indicate the species of both isolates, thus other regions such as large-subunit rRNA (LSU) and small-subunit rRNA (SSU) should be included in molecular analysis. In addition, investigation of morphology characters of both isolates showed differences, further analyses would be useful for the identification.

Sordariales is a large group of microfungi that occur as the degraders of dung, wood, plant debris, and soil (Cannon and Kirk, 2007). However, these fungi have been isolated as endophytuc fungi such as *Chaetomium* (Beena *et al.*, 2000; Li *et al.*, 2014b) and *Trichoderma* (Bailey *et al.*, 2009; Mulaw *et al.*, 2013; Hosseyni-Moghaddam and Soltani, 2014). They produce bioactive compounds with antimicrobial activity. The synthetic chemicals are costly and have harmful effect on human health and environment, including may induce pathogen resistance. Thus, biological control by using a cooperation of microorganisms such as endophytic fungi isolate AL1T1 or AL1T2 and isolate BS1B201 to prevent disease and promote plant growth offer an attractive alternative method for disease management without the negative impact of the chemical control.

> ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

CHAPTER 4

Existence of *Muscodor* in leaves of the rubber tree

4.1 Introduction

Microbial endophytes are bacteria or fungi which live within plants without causing apparent disease (Huang *et al.*, 2001; Chaverri and Gazis, 2011). As symbionts most endophytes enhanced plant growth and a plant's defensive system (Strobel and Daisy, 2003; Tejesvi *et al.*, 2007). Because of such benefits, endophytes are one of the promising microbial resources as biocontrol agents applied in agriculture. A mode of action of endophytes in prevention and/or suppression of plant diseases caused by phytopathogens is the production of antimicrobial substances. The bioactive compounds in the form of volatile metabolites produced by endophytes have not been commonly reported. Some fungi that belong to the families *Diaporthaceae*, *Hypocreaceae*, and *Xylariaceae* of the phylum Ascomycota are notable for their capacity to form volatile metabolites with antimicrobial activity (Stinson *et al.*, 2003b; Suwannarach *et al.*, 2013b). In the family *Xylariaceae*, *Muscodor albus* is the first known fungal endophyte isolated from *Cinnamomum zeylanicum*, which produces bioactive volatile metabolites (Strobel *et al.*, 2001; Ezra *et al.*, 2004).

The genus *Muscodor* does not produce spores and is classified into the family *Xylariaceae* based on the polyphasic approach (Kudalkar *et al.*, 2011). Although morphological characteristics can be used to classify the members of *Muscodor* at the generic level, a more appropriate genotypic classification method is needed for identification at the species level and to understand their phylogenetic lineage. In addition, profiling and comparison of unique volatile metabolites produced by members of the genus *Muscodor* are important criteria for its classification (Strobel, 2006; González *et al.*, 2009). The major volatile metabolite produced by the genus *Muscodor* is 2-methylpropanoic acid (Kudalkar *et al.*, 2011; Suwannarach *et al.*, 2013a), which differs from other endophytic genera in the same family, e.g. *Nodulisporium* spp. that produce

ເນດດັກແດລ໌ແມ

eucalyptol (Tomsheck *et al.*, 2010). Recently, many novel species of the genus *Muscodor* (*M. cinnamomi*, *M. equiseti*, *M. musae*, *M. oryzae*, and *M. suthepensis*) were isolated from diverse plant hosts growing in northern Thailand (Suwannarach *et al.*, 2010; 2013a). Other species of *Muscodor* have been isolated from tropical trees and vines in Australia, Central and South America, and Central, South and Southeast Asia (Meshram *et al.*, 2013; Suwannarach *et al.*, 2013a; Saxena *et al.*, 2015).

While a previous report showed that *Penicillium*, *Pestalotiopsis* and *Trichoderma* were the most frequently isolated fungi from rubber trees (Gazis and Chaverri, 2010). However, a diversity study of the genus *Muscodor* that lives in association with rubber trees has not been reported previously. Volatile metabolite-producing endophytes may be an alternative biological approach as biofumigation in control of plant diseases and fungal contamination of rubber tree seedlings and their latex products. It is known that sapwood and leaf fragments of rubber trees are a rich source of fungal endophytes (Evueh and Ogbebor, 2008; Rocha *et al.*, 2010; Gazis, 2012), but few studies have reported their antimicrobial activities (Evueh and Ogbebor, 2008; Gazis, 2012). Moreover, no volatile metabolite-producing endophytes have yet been reported from rubber trees.

4.2 Materials and methods

4.2.1 Preparation and surface sterilization of plant materials

Healthy leaves of rubber tree were used as a source for isolation of endophytic fungi. A total of 45 leaves were collected from rubber trees planted in Nongbualamphu province, northeastern Thailand (geographical location: $17^{\circ}12'14''$ N, $102^{\circ}26'26''$ E) (Figure 4.1). They were placed in plastic bags and stored in an icebox before transported to the laboratory within 24-48 h after sampling. The age of rubber trees was approximately 6 years, which was considered as the mature phase for latex harvest. The leaves were prepared and surface sterilized following the method described by Suwannarach *et al.* (2010). Briefly, plant leaves were rinsed with tap water for 15 min and randomly cut into small segments (25 mm²). All segments were sterilized by soaking in 75% ethanol for 30 s, 2% sodium hypochlorite for 3 min, and 95% ethanol for 30 s, respectively.



Figure 4.1 Colleting sites (star marks) of rubber tree in Nong Bua Lamphu province, Thailand.

4.2.2 Isolation of volatile metabolite-producing endophytic fungi

Endophytic fungi that form volatile metabolites were isolated using a parallel-growth isolation technique (Worapong *et al.*, 2001). Briefly, a part of a twocompartment petri dish was filled with potato dextrose agar (PDA, Himedia Laboratories, India), while the other side contained a half strength PDA supplemented with rose bengal and chloramphenicol at final concentrations of 0.033 g L⁻¹ and 50 mg L⁻¹, respectively. A strain producing volatile antibiotic metabolites, *M. cinnamomi* (MB518008) (Suwannarach *et al.*, 2010) was used as a reference strain for this isolation. An agar plug (6 mm diam) of *M. cinnamomi* grown previously on PDA at ambient temperature (25 ± 2 °C) for 7 days was placed on the PDA side of the prepared two-compartment Petri dish. This inoculation was incubated at 25 ± 2 °C for 4 days. Then, five surface sterilized plant segments were placed on the half strength PDA, where they were exposed to the volatile metabolites produced by *M. cinnamomi*. The Petri dish was sealed with Parafilm[®] M (Bemis company, Inc., USA), and incubated at 25 ± 2 °C for 2 weeks. The production of its VOCs facilitated selection pressure, allowing the growth of only fungal species that tolerated these volatiles (Figure 4.2). Then, the hyphal tips of endophytic fungi that grew out from the plant segments were aseptically transferred to a separated PDA plate. This technique could be used to select for other isolates of *Muscodor* (Worapong et al., 2001; Mitchell et al., 2008). The endophytic fungi were tested for antifungal activity following a protocol described by Strobel et al., 2001, if they produced volatile antibiotics. Spore production of endophytic fungi was studied on four different media including corn meal agar (CMA), malt agar (MA), PDA and water agar (WA). The microscopic structures like hyphal characteristics, cellular bodies and spore production were observed under a light microscope (Olympus SZ40, Japan). The colony texture, color and fruiting body were observed under a stereomicroscope (Olympus CH30, Japan). The pure cultures of endophytic fungi were subsequently stored in 20% (v/v) glycerol at -20 °C and deposited at the Sustainable Development of Biological Resources (SDBR) Lab, Department of Biology, Faculty of Science, Chiang Mai University, Chiang Mai, Thailand, and the BIOTEC Culture Collection (BCC), Bangkok, Thailand.



Figure 4.2 Muscodor grew out from the plant segment on half strength PDA.

4.2.3 Phenotypic classification

Colony and hyphal characteristics of all endophytic fungal isolates were initially observed with a light microscope (Olympus CH30, Japan). The morphological characteristics were observed also with a scanning electron microscope (JEOL JSM-5910LV, Japan), following a protocol described by Ezra *et al.* (2004) (Appendix D). All endophytic fungal isolates whose morphology was typical of the genus *Muscodor*, grouped and selected for further classification.

4.2.4 Genotypic classification

All Muscodor isolates were grown on PDA at 25 ± 2 °C for 10 d. The aerial mycelium of each isolate was scraped from the surface of PDA. This fungal biomass was then freeze-dried and ground into a fine powder with a pestle and mortar. A modified SDS-CTAB method (Suwannarach et al., 2013a) was used for the DNA extraction. The internal transcribed spacer regions 1 and 2, including 5.8S rDNA (ITS1-5.8-ITS2 rDNA) was a targeted nucleotide sequence for the amplification, using a pair of universal primers (ITS4 and ITS5). The amplification was carried out using GeneAmp 9700 thermal cycler (Applied Biosystems) with the following PCR conditions: initial denaturation at 95 °C (2 min), 30 cycles of reaction [denaturation at 95 °C (30 sec), annealing at 50 °C (30 sec), and extension at 72 °C (1 min)], and final extension at 72 °C (10 min). The reaction mixture (25 µL) contained 1 µL of the DNA template, 0.2 µM dNTP, 0.2 µL of FastTaq (Applied Biosystems), 0.2 µM each primer, 2.5 µL of MgCl₂ buffer, and sterile water to bring the volume to 25 µL. The PCR product was confirmed by gel electrophoresis and purified with PCR cleanup Gel Extraction NucleoSpin[®] Extract II Purification Kit (Macherey-Nagel, Germany) following the manufacturer's manual. The purified PCR product was then sequenced by 1st Base, Malaysia. All DNA sequences obtained were aligned and compared to the available sequences in GenBank database using the BLASTN facility within the NCBI website (http://www.ncbi.nlm.nih.gov/BLAST/). Phylogenetic analysis of all sequences was conducted by the neighbor-joining method using MEGA6 software (Tamura et al., 2013).

4.2.5 Profiling and comparison of volatile metabolites

Only the *Muscodor* isolate that was a possible novel species (having a <86% similarity to the ITS rRNA sequence of other members of the genus *Muscodor*), was analyzed for its profile of volatile metabolites. The mixture of volatile metabolites at headspace of the *Muscodor* isolate growing on PDA at 25 ± 2 °C for 10 days was identified by a modified gas chromatography-mass spectrometry (GC-MS) procedure with solid phase microextraction (SPME) (Strobel et al., 2001). The SPME equipped with a syringe consisting of gray hub fiber material made by divinylbenzene/carboxen (50/30) on polydimethylsiloxane affixed on a stable flex fiber, was used for trapping the volatile metabolites produced in the headspace of the Muscodor culture for 45 min. The syringe was inserted into the splitless injection port of an Agilent 7890A gas chromatograph equipped with mass spectrometer MSD 5975C (EI) mass selective detector. A DB-Wax capillary column (30 m \times 0.25 mm I.D.) with a film thickness of 0.25 μ m was used for separation of volatile metabolites. The column was initiated with a thermal program of 40 °C for 2 min, and increasing to 200 °C at a rate of 5 °C min⁻¹. Ultra-high purity helium was used as a carrier gas with an initial column head pressure of 60 kPa. Before trapping the volatile metabolites, the fiber material was conditioned at 250 °C for 34 min under a flow of helium gas. A 30-s injection time was used to introduce the adsorbed volatile metabolites into the GC. The volatile metabolites were identified through a library comparison with the NIST database by considering their quality match (80% or higher).

4.3 Results

4.3.1 Volatile metabolites-producing endophytic fungi of rubber tree leave

Chiang Mai University

A total three isolates of volatile metabolites-producing endophytic fungi was isolated from leaves of rubber trees, using the parallel-growth isolation technique with *M. cinnamomi* as a reference strain. All isolates (100%) were classified morphologically into the genus *Muscodor*. The phylogenetic analysis of ITS rRNA sequences derived from all *Muscodor* isolates revealed the different closest related species (Table 4.1). Isolates RTM5IV1 and RTM5IV2 were closely related to *M. vitigenus* and *M. equiseti*, respectively. This phylogenetic relation was supported by 100% similarity of the rRNA gene sequence and morphological characteristics. Isolate RTM5IV3 showed the lowest sequence similarity (86%) to *M. albus* cz620, *M. cinnamomi, M. musae*, and *M. oryzae*. The phylogenetic analysis of ITS rRNA sequences derived from all *Muscodor* isolates revealed the different closest related species and displayed three majorclades, A, B, and C, containing all known species of *Muscodor* and all isolates from this study with a high bootstrap support (91%) (Figure 4.3). Isolate RTM5IV3 showed the lowest sequence similarity (86%) to *M. albus* cz620, *M. cinnamomi, M. musae*, and *M. oryzae*. In clade A, it separated from *M. albus* group, *M. strobelii* and *M. suthepensis* with 93% bootstrap support (Figure 4.3). Isolates RTM5IV2 showed a maximum homology of 100% with *M. vitigenus* and *M. equiseti*, respectively and were classified into clade B with 99% bootstrap support (Figure 4.3). Morphological data and bioinformatics were insufficient to place the *Muscodor* isolates into exact linages. Nevertheless, we could suppose that the isolates giving a % sequence similarity lower than 100% could be novel members of the genus *Muscodor*. With this hypothesis, one of the *Muscodor* isolates (33.3%) found in this work was possibly a novel species isolated from rubber tree leaves.

4.3.2 Volatile analysis of *Muscodor* spp.

Isolate RTM5IV3 produced a mixture of 12 volatile metabolites which identified by GC/MS through a library comparison with the NIST database (Table 4.2 and Figure 4.4). Its volatile profile was compared with its closest phylogenetic species; M. albus cz620, M. cinnanomi, M. musae, and M. oryzae. The volatile compound produced by isolate RTM5IV3 with the highest percentage peak area (11.26) was 3methylbutan-1-ol, the same as produced by M. albus cz620 and M. oryzae. The minor volatile compounds were 3-methylbutyl acetate and 2-methylpropanoic acid. These volatile compounds are also produced by M. albus cz620. However, the volatile profile of isolate RTM5IV3 was different from other Muscodor spp. (Table 4.2). In addition, isolate RTM5IV3 produced two volatile compounds, 1,1,9-trimethyl-5methylidenespiro [5.5] undec-9-ene and methyl (Z)-N-hydroxybenzenecarboximidate, that were not found in other *Muscodor* spp. (Table 4.2).

Isolate	Sequence	Accession	Closest related taxa in % Seque				
number	length (nt)	number	the GenBank database	similarity			
RTM5IV1	584	KF850710	M. vitigenus AY100022	100			
RTM5IV2	611	KF850711	M. equiseti JX089322	100			
RTM5IV3	593	KF850712	M. albus cz620	86			
		1819	AF324336				
	12	10 01	M. cinnamomi GQ848369				
	5	S	<i>M. musae</i> JX089323				
	B	La contraction	<i>M. oryzae</i> JX089321				
	SHITLE ST		A STILL	3			
		MAIU	NIVERS				
ຄີ ເ A	เสิทธิ์ เ pyright [©] IIri	มหาวิท by Ch	<mark>ายาลัยเชียง</mark> niang Mai Unive s reserv	ใหม่ ersity e d			

Table 4.1 Closest related species of *Muscodor* isolates based on the sequence similaritiesof ITS rRNA gene.



Figure 4.3 Neighbor-joining (NJ) phylogenetic tree of *Muscodor* spp. based on ITS rRNA sequence alignment of 25 sequences. *Hypocrea lixii* was used as an out group. The isolate RTM5IV3 (KF850712) was proposed as a novel species, *M. heveae* (see text for the description of novel species). The isolate RTM5IV1 (KF850710) and RTM5IV2 (KF850711) were proposed as *M. vitigenus* and *M. equiseti*, respectively. The codes indicated in parentheses refer to the accession numbers of the sequences available in GenBank database.

Copyright[©] by Chiang Mai University All rights reserved



Figure 4.4 SPME-GC/MS profile of 3-methylbutan-1-ol, 3-methylbutyl acetate and 2methylpropanoic acid produced by *Muscodor heveae* cultured on PDA.



RT	Possible compound	Molecular	M/z		% Total area				
(min)	2 91	formula	2	Blank	MH ^a	MA ^b	MC ^c	MM ^d	MO ^d
5.41	cyclopentane	C5H10	7.69	0	0	0	7.69	0	0
5.56	ethyl acetate	$C_4H_8O_2$	88	0	0	7.63	0	0	0
5.60	2-methylpropan-1-ol	$C_4H_{10}O$	74	0	0.76/86*	0	0	0	0
6.03	3-methylbutyl acetate	C7H14O2	130	0 {	6.92/90*	0	0	0	0
6.38	methyl 2-methylbutanoate	$C_{6}H_{12}O_{2}$	116	0	0	0	14.90	0	0
6.51	methyl propanoate	$C_4H_8O_2$	102	0 0	0	0.31	0	0	0
7.07	2-methylpropan-1-ol	$C_4H_{10}O$	74	04	0	0	0	0	3.48
7.16	ethanol	C ₄ H ₆ O	ND	0	0	6.24	0	0	0
7.77	3-methylbutanoyl acetate	ND	130	0	0	0	0	20.41	4.70
8.03	ethyl 2-methylpropanoate	$C_6H_{12}O_2$	116	08	0	2.07	0	0	0
8.29	3-methylbutan-1-ol Copyright [©] by	C ₅ H ₁₂ O	88	Uni	11.26/83*	0	3.12	0	0
10.32	2-methylbutyl 2-methylpropanoate	$C_9H_{18}O_2$	158	0	veo	0	0	0.21	0

Table 4.2 Composition of the volatile metabolites produced by *Muscodor* spp. by GC/MS.

 Table 4.2 (continue).

RT	Possible compound	Molecular	M/z	% Total area					
(min)	\$ 918	formula	2	Blank	MH ^a	MA ^b	MC ^c	MM ^d	MO ^d
10.88	3-methylbutan-1-ol	C5H12O	88	0	0	0	0	13.34	32.69
11.45	2-methylpropyl 2-methylpropanoate	$C_8H_{16}O_2$	ND	0	0	0.58	0	0	0
12.05	2-methylpropan-1-ol	$C_4H_{10}O$	74	0	0	2.06	0	0	0
12.10	3-methyl-3-buten-1-ol	C5H10O	86	0	0	0	0	0	0.27
12.20	ethyl 2-hydroxy-2-methylpropanoate	$C_{6}H_{12}O_{3}$	132	0	0	0	0	0	0.88
12.50	3-methylbutyl acetate	$C_7H_{14}O_2$	ND	0 0	0	22.24	0	0	0
12.66	1,1,3,3,5,5,7,7,9,9,11,11-dodecamethyl-2,4,6,8,10,12-hexaoxa- 1,3,5,7,9,11-hexasilacyclododecane	$C_{12}H_{36}O_6Si_6$	445	0.67/91*	• 0	0	0	0	0
12.97	3-hydroxybutan-2-one	$C_4H_8O_2$	88	0	0	0	0	0.59	2.00
14.57	3-methylbutyl 2-methylpropanoate	$C_9H_{18}O_2$	ND	0	0	1.53	0	0	0
15.28	3-methylbutan-1-ol	C5H12O	88	1058	0 m	22.99	0	0	0
15.88	benzaldehyde Copyright by (C ₇ H ₆ O	106	4.61/96*	versity	0	0	0	0
16.08	2-pentylfuran	C ₉ H ₁₄ O	138	_o e r	y e o	0.29	0	0	0

 Table 4.2 (continue).

RT	Possible compound	Molecular	M/z		% Total area				
(min)	2 918	formula	9	Blank	MH ^a	MA ^b	MC ^c	MM ^d	MO ^d
16.19	2,4-dimethyl-1-heptene	C ₉ H ₁₈	126	0	0	0	0	0	0.13
16.31	1-isobutoxy-2-ethylhexane	$C_{12}H_{26}O$	186	0	0	0	0	0	0.31
17.13	2-methylpropanoic acid	$C_4H_8O_2$	88	0	4.73/91*	0	0	0	0
19.04	(1 <i>R</i> ,4 <i>E</i> ,9 <i>S</i>)-4,11,11-trimethyl-8-methylidenebicyclo[7.2.0] undec- 4-ene	C ₁₅ H ₂₄	204	0	0.41/93*	0	0	0	0
19.43	isopropyl-4-piperidone	C ₈ H ₁₅ NO	141	0	80	0	0	0.23	0
19.50	4,5-dimethyl-1,3-cyclopentanedione	$C_7 H_{10} O_2$	126	0	0	0	0	0	0.37
19.77	1,1,9-trimethyl-5-methylidenespiro[5.5]undec-9-ene	$C_{15}H_{24}$	204	0	0.51/95*	0	0	0	0
20.38	nonan-2-one	C9H18O	142	0	0	0.41	0	0	0
20.75	azulene,1,2,3,5,6,7,8,8 α -octahydro-1,4-dimethyl-7-(1- methylethenyl)-, [1 <i>S</i> (1. α .,7. α .,8 α . β .)]-	C ₁₅ H ₂₄	204	0- 100	2.03/99*	0	0	0	0
21.03	2-methylpropanoic acid	$C_4H_8O_2$	88	0 Jni	versity	0	0	33.17	15.41

 Table 4.2 (continue).

RT	Possible compound	Molecular	M/z			% Total area			
(min)	2 918	formula	91	Blank	MH ^a	MA ^b	MC ^c	MM ^d	MO ^d
21.07	8α -methyl-4-methylidene-6-propan-2-ylidene-2,3,4 α ,5,7,8- hexahydro-1 <i>H</i> -naphthalene	C ₁₅ H ₂₄	204	0	0	0.3	0	0	0
21.69	$[1S-(1, \alpha., 4, \alpha., 7, \alpha)]-1, 2, 3, 4, 5, 6, 7, 8$ -octahydro-1, 4-dimethyl-7-(1methylethenyl)-azulene	C ₁₅ H ₂₄	204	0	0	0	0	0.29	0
21.81	(4Z)-4,11,11-trimethyl-8-methylidenebicyclo[7.2.0]undec-4-ene	$C_{15}H_{24}$	204	0	0	0	0	0	2.19
21.92	methyl (Z)-N-hydroxybenzenecarboximidate	C ₈ H ₉ NO ₂	151	0	0.31/83*	0	0	0	0
22.54	azulene	$C_{10}H_8$	204	0	0	1.51	0	0	0
22.61	2-phenylethyl acetate	$C_{10}H_{12}O_2$	164	0	0.9/83*	0	0	0	0
23.16	1-methyl-4-[(2E)-6-methylhepta-2,5-dien-2-yl]cyclohexene	$C_{15}H_{24}$	204	0	0	0.94	0	0	0
23.76	2-methylbutanoic acid	$C_{15}H_{10}O_2$	102	0	0	0	0	1.10	0
24.72	2-phenylethanol	C ₈ H ₁₀ O	122	0	1.28/97*	0	0	0	0
24.75	azulene,1,2,3,5,6,7,8,8α-octahydro-1,4-dimethyl-7-(1- methylethenyl)-, [1S (1.α.,7.α.,8α.β.)]-	$C_{15}H_{24}$	204	Uni e r	versity versity	0	0	1.00	0.44

 Table 4.2 (continue).

RT	Possible compound	Molecular	M/z	z % Total area					
(min)	2 918	formula	9	Blank	MH ^a	MA ^b	MC ^c	MM ^d	MO ^d
25.20	1 <i>H</i> -3α,7-methanoazulene, 2,3,4,7,8,8α-hexahydro-3,6,8,8 tetramethyl-,[1 <i>R</i> -(1.α., 4α.α., 8α.α.)]	C ₁₅ H ₂₄	204	0	0	3.63	0	0	0
25.30	2-methylpropanoic acid	$C_4H_8O_2$	88	0	0	6.08	0	0	0
25.72	3,7-dimethyl-1,6-octadiene	$C_{10}H_{18}$	138	0	0	0	0	0.05	0
26.04	(1 <i>R</i> ,4 <i>E</i> ,9 <i>S</i>)-4,11,11-trimethyl-8-methylidenebicyclo[7.2.0]undec- 4-ene	C15H24	204	0	0	0.48	0	0	0
26.45	phenol	C ₆ H ₆ O	94	1.80/95*	· 0	0	0	0	0
26.91	2-methyl-propanamide	C ₄ H ₉ NO	87	0	0	0	0	0.07	0
27.05	2-phenylethyl acetate	$C_{10}H_{12}O_2$	164	0	0	0	0	3.28	1.46
27.27	6-nitro-2-picoline	$C_6H_6N_2O_2$	138	0	0	0	0	0.12	0
27.42	(1E,5E)-1,4,4-trimethyl-8-methylidenecycloundeca-1,5-diene	C ₁₅ H ₂₄	204	0	0	0	3.23	0	0
27.55	naphthalene,1,2,4 α ,5,6,8 α -hexahydro-4,7-dimethyl-1-(1- methylethyl)-,[1 <i>R</i> -(1. α ., 4 α . α ., 8 α . α .)]	C ₁₅ H ₂₄	204	Univ e r	versity vec	0.34	0	0	0

 Table 4.2 (continue).

RT	Possible compound	Molecular	M/z	% Total area					
(min)	2 9	formula	9	Blank	MH ^a	MA ^b	MC ^c	MM ^d	MO ^d
28.34	(<i>R</i>)-11-methylene-3,7,7-trimethylspiro[5.5]undec-2-ene	C15H24	204	0	0	0.36	0	0	0
28.50	azulene, 1, 2, 3, 5, 6, 7, 8, 8 α -octahydro-1, 4-dimethyl-7-(1- methylethenyl)-, [1 <i>S</i> (1. α ., 7. α ., 8 α . β .)]-	C ₁₅ H ₂₄	204	0	0	1.07	0	0	0
28.57	(2 <i>R</i>)-8,8,8α-trimethyl-2-prop-1-en-2-yl-1,2,3,4,6,7- hexahydronaphthalene	C ₁₅ H ₂₄	204	0	0	3.24	0	0	0
30.18	azulene,1,2,3,5,6,7,8,8α-octahydro-1,4-dimethyl-7-(1- methylethenyl)-, [1S (1.α.,7.α.,8α.β.)]-	C15H24	204	0	3.3/95*	0	0	0	0
30.89	azulene, 1, 2, 3, 5, 6, 7, 8, 8 α -octahydro-1, 4-dimethyl-7-(1- methylethenyl)-, [1 <i>S</i> (1. α ., 7. α ., 8 α . β .)]-	C ₁₅ H ₂₄	204	0	0	0	8.58	0	0
30.9	$(3R, \alpha S, \alpha R)$ - α -methyl-5-methylidene-3-prop-1-en-2-yl- 1,2,3,4, α ,6,7,8-octahydronaphthalene	C ₁₅ H ₂₄	107	0	0	0	7.32	0	0
31.12	2-phenylethyl acetate	$C_{10}H_{12}O_2$	ND	0	0	1.74	0	0	0
31.72	2-(2-methyl-2-propenyl)-2-cyclohexane-1-one		150		versity 0 v e d	0	0	0.25	0

 Table 4.2 (continue).

RT	Possible compound	Molecular	M/z			% To	tal area		
(min)		formula	-						
		. Mai 1961	91	Blank	MH ^a	МАь	MC ^c	MMª	MO ^d
			4	β / β					-
33.17	2-phenylethanol	C ₈ H ₁₀ O	122	0	0	1.06	0	0	0
			. \	- 3	11				

The volatile metabolites produced by *M. heveae* (MH) and its closest phylogenetic species; *M. albus* cz620 (MA), *M. cinnamomi* (MC), *M. musae* (MM), and *M. oryzae* (MO) were identified by GC-MS. The results were reported with retention time (RT), possible compound (with IUPAC name), molecular formula, mass to charge ratio (M/z), % total area and % quality*. The data obtained from ^athis study were compared with the GC/MS database of volatile compounds produced by *Muscodor* spp. which reported by ^bStrobel (2011), ^cSuwannarach *et al.* (2010), and ^dSuwannarach *et al.* (2013a). The red highlight refers to the major composition in each profile of the volatile metabolites. Compounds presented in a control PDA plate are not included in this Table. The analysis for *M. heveae* was done in triplicate.

79

AI UNIVERลิขสิทธิ์มหาวิทยาลัยเชียงใหม่Copyright[©] by Chiang Mai UniversityAII rights reserved

	M. heveae ^a	M. albus ^b	M. cinnamomi ^c	M. musae ^d	M. oryzae ^d
Growing condition		91818	านดี		
Media	PDA	PDA	PDA	PDA	PDA
Incubation time	10 days	14 days	10 days	7 days	7 days
Incubation temperature	25 ± 2 °C	23 °C	25 ± 2 °C	25 ± 2 °C	25±2 °C
GC-MS system	2	24 (4/2	a	326	
Equipment : GC	SPME fiber insert into an Agilent 7890A gas chromatograph	SPME fiber insert into a gas chromatograph (Hewlett Packard 5890 series II Plus)	An Agilent Technologis GC 7890	SPME fiber insert into Gas chromatography GC 2010 (Shimadzu, Japan)	SPME fiber insert into Gas chromatography GC 2010 (Shimadzu, Japan)
Equipment : MS	MSD 5975C (EI) mass spectrometer	VG 70E-HF double focusing magnetic mass spectrometer	MSD 5973 (EI) mass spectrometer	MS-QP2010 (Shimadzu, Japan)	MS-QP2010 (Shimadzu, Japan)
Temperature program of GC column	40 °C for 2 min, increased to 200 °C at 5 °C min ⁻¹	25 °C for 2 min, increased to 220 °C at 5 °C min ⁻¹	32 °C for 2 min, increased to 220 °C at 5 °C min ⁻¹	40 °C for 2 min, increased to 200 °C at 5 °C min ⁻¹	40 °C for 2 min, increased to 200 °C at 5 °C min ⁻¹
Condition of fiber	250 °C for 34 min under a flow of helium gas	240 °C for 20 min under a flow of helium gas	250 °C for 39.6 min under a flow of helium gas	250 °C for 57 min under a flow of helium gas	250 °C for 57 min under a flow of helium gas

Table 4.3 Growing condition of *Muscodor* spp. and GC-MS system for analysis of their volatile metabolites.

The data obtained from ^athis study or be taken from ^bStrobel (2011), ^cSuwannarach *et al.* (2010), and ^dSuwannarach *et al.* (2013a).

4.3.3 Taxonomic description of a novel species

Muscodor isolate RTM5IV3 did not produce spores on any medium, like the original *Muscodor* and had rope-like mycelium with coiled structures (Strobel, 2011). Although the isolate formed pale orange colonies when exposed to the natural light similar to *M. cinnamomi* and *M. oryzae*, its mycelial morphology was wavy and hair-like at the mycelial edge and produced a unique unidentified structure (Figure 4.5 and Table 4.4). The isolate RTM5IV3 was classified and identified with the polyphasic approach. Both phenotype and genotype revealed that the isolate belongs to the genus Muscodor. It shared relatively low sequence similarity of ITS rRNA gene (86%) with many species of the genus Muscodor. The phylogenetic tree showed isolate RTM5IV3 was separated from the other 12 Muscodor spp. in clade A with high bootstrap support (Figure 4.3). The phylogenetic dendrogram also supported the node separating isolate RTM5IV3 in clade A from five species of Muscodor in clade B and C with 88 and 91% bootstrap support, respectively. The isolate RTM5IV3 was able to form volatile metabolites with antimicrobial activity against diverse tested microbes, which is a typical phenotype of the genus Muscodor. However, the profile of volatile metabolites produced by the isolate was different than its closely related phylogenetic species, but the major component of the volatile metabolites was 3-methylbutan-1-ol, which was the same as the type species of the genus Muscodor, M. albus cz620 (Tables 4.2). Based on the phenotypic, genotypic and phylogenetic evidence, isolate RTM5IV3 represents a novel species of the genus Muscodor, for which the name Muscodor heveae sp. nov. was proposed. The taxonomic description includes the following:

Taxonomy right[©] by Chiang Mai University

Muscodor heveae S. Siri-Udom & S. Lumyong, sp. nov. (Figure 4.5) MycoBank: MB809310.

Diagnosis: Pale orange colonies on PDA in natural light, hyphae 1.2 - 3.7 μ m thick, coils 14.8 - 27.1 × 19.7 - 39.4 μ m diam, and producing a fruity odor.

Etymology: *heveae*, refers to the name of the host plant, *Hevea* brasiliensis.

Holotype: THAILAND, Nongbualamphu province, Chiang Mai, Chiang Mai University, from a healthy leaf of *Hevea brasiliensis* (*Euphorbiaceae*) during May,
2011-December, 2012. (ex-type living culture BCC 70461 and SDBR-CMU RTM5IV3), GenBank sequence KF850712.

Teleomorph: Unknown.

Description: In nature, the fungus is associated with *Hevea brasiliensis*. It is an ascomycete and does not produce spores. Colony color of the fungus on PDA in natural light is pale orange but becomes whitish in the dark (Figure 4.5A). The edge of colony was wavy mycelium (Figure 4.5B). Hyphae (1.2 - 3.7 μ m thick) with coils (14.8 - 27.1 × 19.7 - 39.4 μ m diam; Figure 4.5C), commonly intertwining of mycelium into twisted cable-like strand and rope like strand (4.9 - 7.4 μ m thick; Figure 4.5D, 4.5G). Mycelium forms unique unidentified structures (3.8 - 7.5 μ m diam; Figure 4.5E) and swollen cells at hyphal tips (2.5 - 4.9 μ m diam; Figure 4.5F). A culture on PDA produced a fruity odor, which contains 3-methylbutan-1-ol as a major component. Spores and other fruiting bodies did not appear under any of the tested conditions.

The other two isolates of *Muscodor* tested; isolate RTM5IV1 and RTM5IV2 were identified as *M. vitigenus* (Mycobank: MB373747) (Figure 4.6) and *M. equiseti* (Mycobank: MB800814) (Figure 4.7), respectively based on morphology and phylogenetic analysis. The description of *M. vitigenus* and *M. equiseti* were showed in previous publications of Daisy *et al.* (2002) and Suwannarach *et al.* (2013a), respectively.

MAI UNIVER



Figure 4.5 Morphological characteristics of *Muscodor heveae* (RTM5IV3) isolated from leaves of rubber tree. Difference in colony morphology was observed after growing on PDA in the dark (A) for 10 days. The hyphal morphology of *M. heveae* was observed further with light (B-D) and scanning electron (E-G) microscopes. The wavy mycelium from the colony edge (B), formation of coiling hypha (C), rope-like mycelium (D, G), unique unknown structure (E), and swollen like hyphal tips (F) were found as a typical morphology of *M. heveae*.



Figure 4.6 Morphological characteristics of *Muscodor vitigenus* (RTM5IV1) isolated from leaves of rubber tree. Difference in colony morphology was observed after growing on PDA in the dark (A) for 10 days. The hyphal morphology of *M. vitigenus* was observed further with light (B-C) and scanning electron (D) microscopes. The formation of rope-like mycelium (B, D), and coiling hypha were found as a typical morphology of *M. vitigenus*.



Figure 4.7 Morphological characteristic of *Muscodor equiseti* (RTM5IV2) isolated from leaves of rubber tree. Difference in colony morphology was observed after growing on PDA in the dark (A) for 10 days. The hyphal morphology of *M. equiseti* was observed further with light (B-E) and scanning electron (F) microscopes. The formation of coiling hyphae (B-C), triangular branching pattern (D), swollen cells (C-D), rope-like mycelium (E), and cottony-like mycelium (F) were found as a typical morphology of *M. equiseti*.

ลิ<mark>ขสิทธิ์มหาวิทยาลัยเชียงใหม่</mark> Copyright[©] by Chiang Mai University All rights reserved

MAI UNIVE

Morphology and	M. heveae	M. albus cz620 ¹	M. cinnamomi	M. musae	M. oryzae			
bioactivity		. 918	ยนติ					
Mycelial growth	Rope-like strand and	Rope-like with coils	Rope-like with	Rope-like with coils	Rope-like with coils			
	swollen cells with coils	structure	cauliflower-like bodies	structure	structure			
	structure	15.17		3				
Hyphal growth at	Wavy and hairy-like	Straight	Straight	Straight and hairy-like	Straight			
colony front	mycelium	304 (3-		mycelium				
Pigment production				7955				
in the light	Pale orange	Whitish	Pale orange	Whitish	Pale orange			
in the dark	Whitish	ND	Whitish	Whitish	Pale orange			
Host	Hevea brasiliensis	Cinnamomum zeylanicumin	Cinnamomum bejolghota	Musa acuminata	Oryza rufipogon			
Bioactivity	Antifungal/antibacterial	Antifungal/antibacterial	Antifungal/antibacterial	Antifungal/antibacterial	Antifungal/antibacterial			
Reference	In this work	Worapong et al. (2001)	Suwannarach et al. (2010)	Suwannarach et al.	Suwannarach et al.			
	ິດີເ	เสทธ์มหาวั	ทยาลยเชีย	(2013a)	(2013a)			
	Co	nvright [©] hv C	hiang Mai Un	iversity				
ND refers to not de	ND refers to not determined.							
	All rights reserved							

Table 4.4 Morphology and bioactivity of *Muscodor heveae* and its closely related phylogenetic species.

4.4 Discussion

It is known that ascomycetes including the following notable genera, Colletotrichum, Penicillium, Pestalotiopsis, and Trichoderma are dominant fungal endophytes of rubber tree (Gazis and Chaverri, 2010; Gazis, 2012). In addition, Arbuscular mycorrhizal fungi (AMF) were first described on the root of rubber tree but rubber tree plantation expressed lowest numbers of AMF (Pereira et al., 2014). Fungal communities in rubber tree were affected by the physical and chemical properties of trees, tree ages, biological interaction, substrate preference, host preferences and geographical characters (Kodsueb et al., 2008; Monkai et al., 2016). However, there is no report of volatile metabolites-producing endophytes that live associated with rubber tree. We were the first to find that the rubber tree can be a source for discovering volatile metaboliteproducing endophytes. The isolation technique used in this study was highly selective for the genus Muscodor. This was similar to a number of previous studies that used the same technique for isolation of volatile metabolite-producing endophytic fungi (Suwannarach et al., 2010; 2013a). Moreover, there was no contamination found during such isolations, suggesting that the volatile metabolites produced by either the reference strain and the emerging isolates exhibited strong and broad-spectrum antimicrobial activity. In addition, determining the distribution of volatile metabolites-producing endophytes in different plant tissues of rubber trees would be a further study in order to understand their spatial and temporal dynamics within the plant host.

In this study, we proposed a novel species, *M. heveae* that has a phenotypic and phylogenetic relationship with the genus *Muscodor*, but showing a distinct set of morphology, genotype, and profile of volatile metabolites produced. The phylogenetic analysis based on partial sequences of 18S rDNA, ITS1-5.8-ITS2 rDNA was required for accurate genetic classification. These sequences have been demonstrated to be highly conserved regions of DNA and very useful in the classification of fungi (von der Schulenburg *et al.*, 2001). The sequence data obtained on *Muscodor* indicated that it was a xylariaceous fungus with incomplete molecular identity to other fungi in family *Xylariacea* (Ezra *et al.*, 2004). In addition, a high percent similarity (99-100%) of ITS rRNA sequences among several emerging species of the genus *Muscodor* was often found (Suwannarach *et al.*, 2010; Kudalkar *et al.*, 2011). However, the authors are still confident

that their isolates could be proposed as the novel species. The most notable property of *Muscodor* was its ability to produce a mixture of volatile metabolites with antibiotic activity against variety of pathogenic fungi and bacteria (Ezra *et al.*, 2004).

Thus, detailed phenotypes of the isolates, volatile gas composition analysis and comparison with their closely related phylogenetic species were priority criteria for assigning their novelty. At early stage, colony and mycelial characteristics were the only available phenotypes to identified Muscodor species (Zhang et al., 2010). Subsequently, analysis and comparison of volatile metabolites produced by Muscodor isolates have been used as an important tool for their classification and novelty determination. Variation of volatile metabolites-profile is commonly found across member species of the genus Muscodor, but they often share the same major volatile component (Kudalkar et al., 2011; Suwannarach et al., 2013a). The volatile profiles (Table 4.2) of four Muscodor spp. have been compared with M. heveae. All species are prevalent in Thailand except M. albus cz620. However, the previous reports showed that there were many isolates of M. albus existed in local host plant in many countries (Ezra et al., 2004; Atmosukarto et al., 2005; Strobel et al., 2007; Banerjee et al., 2010; Banguela-Castillo et al., 2015). Including, M. albus MFC2 have been isolated from Myristica fragrans in Thailand and produced bulnesene, ledol, (-)-globulol and azulene derivatives as major volatiles (Sopalun et al., 2003). The variation profile of volatile metabolites may be caused by diverse factors, including growing media and environment, and enzyme activity for biosynthesis of such metabolites by the test microbes (Morath et al., 2012).

The bioactive volatile components for the genus *Muscodor* were 2methylpropanoic acid, 3-methylbutyl acetate and 3-methyl-1-butanol (3-methylbutan-1ol) (Strobel, 2006; Mitchell *et al.*, 2010), which were in agreement to the one detected in *M. heveae*. Volatile metabolites are formed as a mixture, and not all volatile metabolites exhibit antimicrobial activity. However, applying the mixture of volatile metabolites for antimicrobial activity has been suggested with more potent than using them as the individual compounds (Strobel *et al.*, 2001). The major volatile metabolites in the mixture are not always the bioactive ones, and this is different between either species or taxa. In the genus *Muscodor*, the major volatile antimicrobial metabolite was 2-methylpropanoic acid (Mitchell *et al.*, 2010). This bioactive volatile metabolite has been found to be formed by a yeast *Saccharomyces cerevisiae*, but the most active compounds of its volatile mixture were 2-methyl-1-butanol and 3-methyl-1-butanol (Fialho *et al.*, 2011).



CHAPTER 5

Antimicrobial activity of *Muscodor* spp. and relevant alleopathy effects

5.1 Introduction

Biological control is the control of destructive organisms by the use of other organisms. It suggests that endophytes can be used as biocontrol agents because they can help hosts tolerate harmful abiotic and biotic factors including pathogens. *Trichoderma* sp., one species of fungal endophyte was successfully used to control phytopathogens, *Colletotrichum gloeosporiodes* and *Rigidoporous lignosus* in rubber tree (Gazis and Chaverri, 2010). Endophytes could be used directly to treat seeds or transplants (Backman and Sikora, 2008), but the inoculation methods for endophyte associated with trees are limited. Syringes inoculation and soaking seeds in spore suspensions were used in the case of grasses and sprays of spore suspension have been used to introduce endophytes into beans and barley (Mejia *et al.*, 2003).

Strobel *et al.* (2001) recently reported the volatile antibiotic from an endophytes, *Muscodor albus* which effectively inhibited and killed certain pathogenic fungi and bacteria. Volatile organic compounds (VOCs) are low molecular mass and usually hydrophobic compounds with high vapor pressure, i.e. they easily evaporate at room temperature. They can diffuse a long way from their point of origin and migrate in soil and aerial environments as well as through porous wood materials (Hung *et al.*, 2013). Like plants and bacteria, fungi produce a large number of VOCs as mixtures of alcohols, ketones, esters, small alkenes, monoterpenes, sesquiterpenes, and derivatives (Korpi *et al.*, 2009). The mixtures of VOCs produced by fungal endophytes have both antibacterial and antifungal activities (Strobel *et al.*, 2001, 2006). The most effective group of inhibitory compounds was isoamyl acetate (Strobel and Daisy, 2003) and it benefit to use as mycofumigation to treat soil, seeds and plants. There are many *Muscodor* sp. that showed high activity in control pathogens such as *M. cinnamomi* exactly isolated from cinnamon in Thailand (Suwannarach *et al.*, 2010).

Allelochemistry is the production and release of toxic substances produced by one species, which affect a receiving susceptible species (Saraf *et al.*, 2014). Allelopathy of microorganism is often demonstrated by testing allelochemical effects on seed germination. The previous research showed that *Arabidopsis thaliana* was used as a model to study the plant growth effects of natural mixtures of volatiles emitted by biocontrol, plant pathogenic (Splivallo *et al.*, 2009). Moreover, *A. thaliana* grown in the presence of VOCs from *T. viride* were taller, bigger and had more lateral roots (Hung *et al.*, 2013). However, there are few studies that explore the herbicide activity of the volatile and semi-volatile compounds from endophyte (Guzmán-Trampe *et al.*, 2015).

5.2 Materials and methods

5.2.1 In vitro antimicrobial assay of VOCs

Antagonistic activity of all Muscodor isolates was tested against a set of pathogenic microbes listed in Table 5.1. The test microbes were comprised of bacteria, yeast and filamentous fungi. The parallel-growth isolation technique was adapted for the antagonism test. Briefly, a part of a two-compartment Petri dish was filled with PDA, while the other side contained nutrient agar for the test bacteria, yeast extract-malt extract agar for the test of yeast, or PDA for the test of filamentous fungi. An agar plug from the mycelial margin of each Muscodor isolate growing on PDA was inoculated on the PDA part of the Petri dish, and allowed to grow at room temperature $(25 \pm 2 \ ^{\circ}C)$ for 4 days. Then, each tested bacterium or yeast was individually streaked on its respective agar medium in the Petri dish. For the filamentous fungi test, an agar plug (6 mm diam) of their 4-day-old PDA culture was inoculated on PDA on opposite side of the Petri dish. All Petri dishes were wrapped with Parafilm® M (Bemis company, Inc., USA) and incubated at room temperature (25 ± 2 °C) for 6 days. The percentage of inhibition of fungal growth after the dual culture test was calculated with an equation: $[(R_1-R_2) \times 100]$ $\div R_1$, where R_1 is the average of colony radiuses of each test microbe measured in the control plates (without *Muscodor* isolate), and R_2 is the average of colony radiuses calculated from the test plates. In the case of bacteria and yeast, the microbes were checked for visible growth and viability. The viability of the test microbes was observed by transferring them from the test plates and re-growing in fresh media (Strobel et al., 2001).

Artificial volatile compounds including 3-methylbutan-1-ol (isoamy alcohol, Sigma-Aldrich, Germany), 3-methylbutyl acetate (isoamyl acetate, Merck, Germany) and 2-methylpropanoic acid (isobutyric, Sigma-Aldrich, Germany) were used to demonstrate the relative activity toward hyphal growth. They were found to be the most active volatile components produced by *M. heveae* according to the results in section 4.3.2, chapter 4. The artificial compounds were added individually at different concentrations ranging from 5-35 μ l L⁻¹ airspace into sterile filter paper in the compartments of the Petri dishes with an agar plug of *R. microporus* or *P. noxius* (Suwannarach *et al.*, 2016). The Petri dishes were then sealed with Parafilm® M (Bemis company, Inc., USA), incubated at room temperature (25 ± 2 °C) for 6 days and the radial growth was measured.

5.2.2 Effect of VOCs from Muscodor heveae on root and shoot growth

Four dicots, Arabidopsis thaliana Ceol-0, garden pea (Pisum sativum L.), mung bean (Vigna radiate L.) and tomatoe (Lycopersicon esculentum Mill.), and three monocots, jasmine rice (Oryza sativa L.), ruzi grass (Brachiaria ruziziensis) and sweet corn (Zea May L. var. saccharata) were selected for this study. All plants were selected for their rapid growth and frequency of germination because tropical plants including rubber tree seeds germinate poorly. All of the seeds were commercial products acquired from the Chia Tai Co., Ltd., except for ruzi grass which was purchased from a local market in Nongbualamphu Province, Thailand and A. thaiana col-0 seeds that had been obtained from Professor Kenji Matsui at the Department of Biological Chemistry, Faculty of Agriculture, Yamaguchi University, Japan. An agar plug of *M. heveae* (6 mm Ø) was placed on the PDA medium in one of the compartments of divided Petri dishes (9 cm) and was grown at room temperature (25 ± 2 °C) for 4 days. The seeds were surface sterilized by soaking in 1-2% sodium hypochlorite for 3 min, and washed in sterile distilled water 3 times, while the seeds of A. thaliana col-0 were surface sterilized by soaking in 70 % ethanol for 3 min, 10% sodium hypochlorite for 12-13 min and washed in sterile distilled water for 5-7 times. Then, seeds were placed into the other compartment that contained 1% water agar except for A. thaliana col-0, which were placed on Murashige and Skoog medium (MS medium). All Petri dishes were sealed with Parafilm[®] M (Bemis company, Inc., USA), and incubated at room temperature (25 ± 2) °C) for a photoperiod of 12:12h fluorescent light in a plant growth chamber (Sanyo MLR-

351H, Japan) for 7 and 14 days (Macías-Rubalcava *et al.*, 2010). The volatile assay with *A. thaliana* col-0 was incubated at room temperature $(25 \pm 2 \text{ °C})$ °C for a photoperiod of 16:8h fluorescent light for 14 days. Root and shoot length were measured and compared to the non-muscodor inoculated control with four replications per treatment.

Artificial volatile compounds including 3-methylbutan-1-ol (isoamy alcohol, Sigma-Aldrich, Germany), 3-methylbutyl acetate (isoamyl acetate, Merck, Germany) and 2-methylpropanoic acid (isobutyric, Sigma-Aldrich, Germany) were used to demonstrate the relative activity toward root and shoot growth. The artificial compounds were added individually at different concentrations ranging from 5-35 μ l L⁻¹ airspace into sterile filter paper in the compartments of the Petri dishes with seeds. The Petri dishes were then sealed with Parafilm® M (Bemis company, Inc., USA) incubated under suitable conditions and the root and shoot length were measured.

5.2.3 Statistical analysis

SPSS program version 17.0 for Windows was used to analyze the experimental data with T-test and one-way analysis of variance (ANOVA), along with Turkey's post hoc tests at different significant levels (P).

1 AI UNIVER

1 able 5.1 The list of tested microbes used in this	s study.
--	----------

Tested microbe	Source		
Bacteria			
Bacillus cereus	The Central and Diagnostic Laboratory,		
Methicillin-resistant	Maharaj Nakorn Chiang Mai Hospital,		
Staphylococcus aureus (MRSA)	Faculty of Medicine, Chiang Mai University,		
Micrococcus luteus	Chiang Mai, Thailand		
Enterobacter aerogenes	81918		
Enterococcus faecalis	2/2		
Escherichia coli ATCC 35218	20 %		
Klebsiella pneumoniae (ESBL+)	YR NGN		
Proteus mirabilis	3/2/2/		
Pseudomonas aeruginosa ATCC 27859	- A 136		
Salmonella sp. Group D			
Yeast	VWI Z		
Cryptococcus neoformans	AR SI		
Filamemtous fungi	BO A		
Aspergillus flavus	The Sustainable Development of Biological		
Aspergillus niger	Resources (SDBR) Lab, Department of		
Colletotrichum gloeosporioides	Biology, Faculty of Science, Chiang Mai		
Fusarium oxysporum f. sp. vasinfectum	University, Chiang Mai, Thailand		
Rhizoctonia solani AG-2	niang Mai University		
Ganoderma australe BCC22321	The BIOTEC (National Center for Genetic		
Phellinus noxius BCC26237	Engineering and Biotechnology) Culture		
Phytophthora parasitica BCC15560	Collection (BCC), Bangkok, Thailand.		
Rigidoporus microporus	Department of Microbiology and		
	Parasitology, Faculty of Science, Naresuan		
	University, Phitsanulok.		

5.3 Results

5.3.1 Antimicrobial activity of volatile metabolites produced by *Muscodor* isolates

The volatile metabolites produces by different *Muscodor* isolates were tested for their antimicrobial activity against tested microbes (Table 5.2). The volatile metabolites produced by *Muscodor* isolates RTM5IV1 and RTM5IV3 exhibited antimicrobial activity against Gram positive and negative bacteria, yeast and filamentous fungi, while isolate RTM5IV2 inhibited only the growth of some Gram negative bacteria and filamentous fungi. Isolate RTM5IV3 showed the greatest antimicrobial activity with all tested organisms (Figure 5.1-5.2). *F. oxysporum* f. sp. *vasinfectum* was more susceptible to the volatile metabolites of RTM5IV3 than those of other isolates. *Ps. aeruginosa* ATCC 27859 was the most resistant to the tested microbe, was insensitive to the volatile metabolites produced by all isolate of *Muscodor*. All the *Muscodor* volatiles exhibited significant control of all tested pathogenic fungi of rubber tree (*C. gloeosporioides, P. noxius* and *R. microporus*) compared with other tested fungi.





Figure 5.1 Antifungal ability of VOCs from *Muscodor heveae* (MH) against *Aspergillus flavus, A. niger, Rhizoctonia solani* AG-2 and *Phytophthora parasitica*.



Figure 5.2 Antifungal ability of VOCs from *Muscodor heveae* (MH) against Ganoderma australe, Colletotrichum gloeosporioides, Phellinus noxius and Rigidoporus microsporus.

Test microbes Inhibition activity of microbial growth (viability) after exp					
	ł	y different isolates of <i>Muscodor</i> (%)			
		M. vitigenus	M. equiseti	M. heveae	
Gram positive bacteria					
Bacillus cereus		100 (Dead)	0 (Alive)	100 (Alive)	
Methicillin-resistant		0 (Alive)	0 (Alive)	100 (Dead)	
Staphylococcus aurei	us (MRSA)	10101 9			
Micrococcus luteus	. 91	0 (Alive)	0 (Alive)	100 (Dead)	
Staphylococcus aureus	ATCC 29213	100 (Dead)	0 (Alive)	100 (Dead)	
Gram negative bacteria		She	1.20		
Enterobacter aerogenes	2. /]	100 (Dead)	0 (Alive)	100 (Dead)	
Enterococcus faecalis		0 (Alive)	100 (Dead)	100 (Dead)	
Escherichia coli ATCC	35218	0 (Alive)	0 (Alive)	100 (Dead)	
Klebsiella pneumoniae	(ESBL+)	0 (Alive)	0 (Alive)	100 (Dead)	
Proteus mirabilis		0 (Alive)	100 (Dead)	100 (Dead)	
Pseudomonas aeruginosa ATCC 27859		0 (Alive)	0 (Alive)	0 (Alive)	
Salmonella sp. Group D		0 (Alive)	0 (Alive)	100 (Dead)	
Yeast		CC LANA			
Cryptococcus neoformans		100 (Dead)	0 (Alive)	100 (Dead)	
Filamentous fungi	Nr.		oSY		
Aspergillus flavus	"A	100 (Alive)	98.4±1.3 (Alive)	100 (Dead)	
Aspergillus niger		100 (Dead)	100 (Dead)	100 (Dead)	
Colletotrichum gloeosp	orioides	100 (Dead)	100 (Dead)	100 (Dead)	
Fusarium oxysporum f.	sp. vasinfectum	39.8±2.2 (Alive)	47.4±1.5 (Alive)	62.8±2.2 (Alive)	
Ganoderma australe BO	CC22321	100 (Dead)	100 (Dead)	100 (Dead)	
Phellinus noxius BCC2	6237	100 (Dead)	100 (Dead)	100 (Dead)	
Phytophthora parasitico	a BCC15560	100 (Dead)	97.1±1.3 (Alive)	100 (Dead)	
Rhizoctonia solani AG-	2	100 (Alive)	71.1±2.1 (Alive)	100 (Dead)	
Rigidoporus microporus		100 (Dead)	100 (Dead)	100 (Dead)	

 Table 5.2 Antimicrobial activity of volatile metabolites-producing *Muscodor* against human and plant pathogens.

All tests were done in triplicate and repeated twice, the % inhibition is shown with mean \pm SD while without SD refered to the absence of mean difference. Alive = tested microbe re-growth on fresh media, Dead = tested microbe didn't growth on fresh media. Survival of each tested microbe was evaluated 6 days after exposure.

All active volatile compounds inhibited significantly hyphal growth of *R. microporus* and *P. noxius* (P<0.05). In addition, the major compound; 3-methylbutan-1-ol inhibited hyphal growth of *R. microporus* and *P. noxius* with 100% inhibition, respectively (Figure 5.3-5.4, Table 5.3).



Figure 5.3 Effect of artificial volatile compounds, 2-methylpropanoic acid on the hyphal growth of *Rigidoporus microporus*.



Figure 5.4 Effect of artificial volatile compounds, 3-methylbutyl acetate on the hyphal growth of *Rigidoporus microporus* (A) and *Phellinus noxius* (B).

Chemical agent	Radial growth (cm)				
(µl L ⁻¹ airspace)	Rigidoorus microporus	Phellinus noxius			
Control	3.2a	3.5a			
3-methylbutan-1-ol					
5	3.0±0.1a	3.2±0.1a			
10	2.5±0.3b	1.8±0.1b			
15	1.9±0.3c	1.5±0.9b			
20	Od	0.3±0.1c			
25	Od	0c			
30	Od	0c			
35	Od	Oc			
2-methylpropanoic acid	Style 1	- <u>78</u> 5			
5	1.7±0.1b	3.5±0.1a			
10	1.3±0.1c	3.3±0.1ab			
15	1.0±0.2d	2.8±0.1b			
20	0.2e	2.1±0.1c			
25	Of	0.8±0.1d			
30 ลิขสิทธิ์บเ	หาวิท ⁰ ยาลัยเ	0e			
³⁵ Copyright [©]	by Chiang Mai	0e University			
Allrig	ghts res	erved			

 Table 5.3 Effect of artificial volatile compounds on the hyphal growth of fungal pathogens.



Table 5.3 (continue).

All tests were done in triplicate, the radial growth is shown with mean \pm SD while without SD refers to the absence of mean difference. Means with different letters are significantly different according to Tukey's Test (P < 0.05).

5.3.2 Effect of VOCs from *M. heveae* on root and shoot growth

The volatile metabolites affected root and shoot growth of the tested plant. They inhibited significantly root growth when compared with the control (P<0.05) (Table 5.4). However, shoot length of garden pea and jasmine rice did not different from the control after 7 day and 14 day exposure of VOCs (Table 5.4). Ruzi grass and *A. thaiana* col-0 were used for the artificial volatile compounds bioassay test as monocot and dicot sample, respectively. All volume of three active volatile compounds inhibited significantly root and shoot growth (P<0.05). In addition, 3-methylbutan-1-ol and 2methylpropanoic acid inhibited seed germination of ruzi grass and *A. thaliana* Ceol-0 with 100% inhibition at 15 μ l L⁻¹ airspace (Figure 5.5-5.6, Table 5.5).

Plant	%	Root length (cm)		Shoot length (cm)	
	Germination	Control	M. heveae	Control	M. heveae
Monocotyledon	-10	2101			
Jasmine rice ²	88.3	3.2±0.4*	0.7±0.4*	2.2±0.5	2.1±0.3
Sweet corn ¹	64.0	2.0*	0*	1.3±0.7*	0*
Ruzi grass ²	70.0	1.3±0.4*	0.2±0.1*	2.8±0.5*	0.1±0.1*
Dicotyledon		-	L	1862.	
Arabidopsis thaliana Ceol-0 ¹	95.0	1.0±0.4*	0*	1.2±0.1*	0*
Garden pea ²	83.3	3.1±0.7*	0.8±0.1*	0.6	0.6±0.3
Mung bean ²	88.0	6.3±1.1*	0.4±0.8*	8.4±0.9*	0*
Tomato ¹	85.0	5.7±0.5*	0.1±0.1*	4.9±1.0*	0*

Table 5.4 Effect of VOCs from Muscodor heveae on root and shoot growth.

¹Measured after 7 days of exposure to fungal VOCs and ²measured after 14 days of exposure to fungal VOCs. All tests were done in four replicates, the experimental data was analyzed with T-test (*P<0.05). Each value is the mean \pm SD while without SD refers to the absence of mean difference.

All rights reserved



Figure 5.5 Effect of VOCs from *Muscodor heveae* on the root length and shoot length of garden peas over 7 days (A), tomato over 7 days (B), ruzi grass over 14 days (C), *Arabidopsis thaliana* Ceol-0 over 14 days (D). Effect of artificial volatile compounds on the root and shoot growth of ruzi grass, *A. thaliana* Ceol-0; 3-methylbutan 1-ol (E) and 3-methylbutyl acetate (F-H), respectively.



Figure 5.6 Effect of artificial volatile compounds, 3-methylbutan-1-ol on the root and shoot growth of ruzi grass.

enemieur ugene	Frank Brands				
(µl L ⁻¹ airspace)	Root length Shoot leng		Root length	Shoot length	
	(cm)	(cm)	(cm)	(cm)	
Control	2.6±0.3a	3.8±0.4a	1.8±0.3a	1.0a	
3-methylbutan-1-ol					
5	0.8±0.8b	1.2±1.2b	0b	Ob	
10	0c	0.2±0.3b	Ob	0b	
15	000	Ob Ob	Ob	Ob	
20	°°0c	0,0b	Ob	Ob	
25	0c	Ob Ob	Ob	Ob	
30	0c	©0b	0b	Ob	
35	0c	0b	Ob	Ob	
2-methylpropanoic acid	à		-38	3	
5	0.2b	0.4±0.2b	0b	Ob	
10	0.1b	0.1±0.1b	Ob S	Ob	
15	Ob	ОЬ	Ob	Ob	
20	0b	Ob	Ob Ob	0b	
25	0b	UNOb	Ob	Ob	
30	0b	Ob	Ob	0b	
³⁵ ลิขสิทธิ์	10b13	1180b a	ОБОТО	Ob	
Copyright	t [©] by (Chiang M	lai Unive	ersity	
Allr	ight	s re	serv	e d	

Table 5.5 Effect of artificial volatile compounds on root and shoot growth of tested plant.Chemical agentRuzi grassArabidopsis thaliana Ceol-0

Table 5.5 (continue).					
Chemical agent	Ruzi	grass	Arabidopsis thaliana Ceol-0		
(µl L ⁻¹ airspace)	Root length	Shoot length	Root length	Shoot length	
	(cm)	(cm)	(cm)	(cm)	
3-methylbutyl acetate					
Control	2.6±0.3a	3.8±0.4a	1.8±0.3a	1.0a	
5	1.2±0.3b	2.4±0.2b	0.6b	0.4±0.1b	
10	0.9±0.1bc	2.0±0.4b	0.3bc	0.3bc	
15	0.9bc	1.7b	0.3bc	0.2cd	
20	0.6bc	1.0±0.3c	0.2c	0.2d	
25	0.7±0.3bc	1.0±0.1c	0.2c	0.2d	
30	0.6±0.4bc	1.0±0.1c	0.2c	0.3d	
35	0.4±0.1 c	0.4c	0c	0d	

All tests were done in triplicate, the radial growth is shown with mean \pm SD while without SD refers to the absence of mean difference. Means with different letters are significantly different according to Tukey's Test (P < 0.05).

5.4 Discussion

In nature, *Muscodor* spp. live within host plants under symbiotic condition, where they produce and expose their volatile metabolites into the surrounding environments of the host plants for preventing the invasion of pathogens (Mitchell *et al.*, 2010; Zhang *et al.*, 2010). A number of endophytes produced extremely biologically active compounds against pathogenic microbes (Yang *et al.*, 1994). Thus, *Muscodor* spp. seemed to be beneficial to the host by providing protection from plant pathogens. Most pathogenic fungi in diverse economic plants were inhibited after exposure to the volatile metabolites produced by *Muscodor* isolates. The inhibitory mechanism on fungal growth was likely that volatile metabolites influence protein expression and the function of metabolic enzymes (Humphris *et al.*, 2002; Fialho *et al.*, 2011). *Muscodor heveae* produced active volatile compounds that are lethal to various microbes including rubber tree pathogenic fungi. The previous reports have shown that VOCs from *M. yucatanensis* inhibited seed germination and root elongation of monocots, barnyard grass and dicot,

amaranth and tomatoes (Macías-Rubalcava *et al.*, 2010). However, the possible toxic effects of VOCs on seed germination and vegetative plant growth have received limited research attention (Lee *et al.*, 2014). Volatile metabolites from *M. heveae* inhibited root and shoot growth in almost all tested plant except for jasmine rice and garden peas. The statistical comparison (T-test) demonstrated that the recorded shoot length of jasmine rice and garden peas was not significantly different from that of the controls (P<0.05).

The artificial volatile compounds test confirmed that the major compound, 3methylbutan-1-ol and the minor compounds 3-methylbutyl acetate and 2methylpropanoic acid of *M. heveae* showed antifungal and phytotoxic activity by inhibited significantly mycelial growth of pathogenic fungi and root and shoot growth of ruzi grass and *A. thaliana*, respectively. The volume of artificial volatile compounds to inhibited mycelial growth and showed phytotoxic activity associated with percentage of peak area from the analysis of volatile metabolites by GC/MS. However, the previous study showed that *A. thaliana* grown in the presence of *Trichoderma viride* volatile had more lateral roots (Hung *et al.*, 2013).

Its volatile revealed 51 compounds of which isobutyl alcohol, isopentyl alcohol and 3-methylbutanal were most abundant, including 2-methyl-1-butanol. This bioactive compound showed antifungal (Fialho et al., 2011) and also phytotoxic activity (Macías-Rubalcava et al., 2010; Sánchez-Ortiz et al., 2016). Thus, the phytotoxic activity depended on the mixture of volatile and type of plant. Strobel et al. (2001) found that the effective inhibitory activity of VOCs depended on the mixture of volatile and type of the living organism. Such as, single volatile compound of *M. albus* had low activity in terms of inhibiting fungal and bacterial growth but as a volatile mixture, they inhibited and killed a broad range of microbes. Maintaining a balance between positive and negative effects of endophytic fungi on plant fitness may be important for multi-trophic interactions. Such as, Alternaria sp. CID62 and Epicocum sp. CID66 produce bioactive volatile to prevent Centaurea stoeba from insect herbivore, while Alternaria sp. can reduce flower development and Epicoccum sp. inhibit seed germination (Newcombe et al., 2009). Thus, more studies are necessary for understanding the significant allelochemical effects and mechanisms underlying these mode of action (Zhi-Lin et al., 2012).

Since bioactive volatile metabolites are found from the genus *Muscodor*, they have been considered for uses in agricultural, medical and industrial applications (Strobel, 2006). Biological control with the volatile metabolite-producing fungi could be a sustainable alternative method to prevent plant diseases at the nursery and postharvest stages, with reduced usage of chemical fungicides (Mercier and Manker, 2005). Biological fumigation was a strategy for applying the volatile metabolites-producing fungi, which has been approved for controlling diverse postharvest infections caused by Botrytis cinerea, Penicillium expansum, and Sclerotinia sclerotiorum (Ramin et al., 2005). The applications of biological fumigation using the genus Muscodor for controlling root diseases has also been demonstrated (Worapong and Strobel, 2009; Suwannarach et al., 2012). Based on the potent antimicrobial activity of the volatile metabolite-producing fungi that we observed, the Muscodor isolates could be used effectively for controlling infectious diseases in rubber trees. Further evaluation of the Muscodor isolates for their antimicrobial activity in applied biofumigation are required not only for agricultural aspects but also medical and industrial relevance. In addition, the phytotoxic activity of volatile compounds produced by *M. heveae* on root and shoot elongation of tested plant could be useful in biocontrol of weeds as potential herbicides.

> ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

The MAI U

CHAPTER 6

Mycofumigation with *Muscodor heveae* for controlling of white root rot disease

6.1 Introduction

White root rot disease caused by *Rigidoporus microporus* is the most serious of all problems faced by rubber tree (*Hevea brasiliensis*) plantations and causes slow death at all stages of growth from seedlings to mature trees (Tangonan *et al.*, 2008). However, this disease is present in the Congo, Cote d'Ivorire, India, Malaysia, Nigeria, Sri Lanka and Thailand (Rajalakshmy and Jayarathnam, 2000; Omorusi, 2012; Kaewchai, 2013). It is responsible for yield losses of up to 50% in rubber plantations in West Africa (Ogbebor *et al.*, 2013), 43% in a small-holdings survey in Malaysia and 96% of the results of rubber tree deaths at a rate of up to five trees per hectare per year in planting areas in Nigeria (Oghenekaro *et al.*, 2015).

In Thailand, a previous report has shown that 55% of diseased trees in rubber plantations located in Phangnga Province suffered from white root rot disease (Kaewchai, 2013). *Rigidoporus microporus* syn. *R. lignosus* is a major economically important pathogen of rubber trees acts by decaying lignin and polysaccharides in the root tissues (Geiger *et al.*, 1986). Previous studies have reported on the high *R. microporus* inoculum density in the soil of rubber tree natural forests and plantations (Nandris *et al.*, 1988). The process of disease infection involves 3 stages: penetration, colonization and degradation. Fungal pathogen live and can remain in the soil for a long period of time. It can infect the surface of the root bark and develop in differentiate rhizomorphs. After which time, it releases extracellular enzymes that are able to decay the wood and collar of host plants (Nakaew *et al.*, 2015). It can cause a significant amount of damage over a wide area. Normally, chemical fungicides have been recommended for controlling root rot disease but they are expensive for local farmers and are highly toxic to human health and the environment (Kaewchai and Soytong, 2010). The use of endophytic fungi as a biocontrol

agent can help hosts tolerate harmful abiotic and biotic factors including pathogens (Singh *et al.*, 2011).

An endophytic fungus, *Muscodor* belonged to the family *Xylariaceae*. The hyphal characteristic is a sterile because no reproductive structure have ever been observed (Strobel, 2006). It has properties associated with producing active low-molecular weight volatile compounds with a broad antimicrobial activity to inhibit or kill most species of fungi and bacteria (Strobel et al., 2001; Mercier et al., 2007). In experiments, Muscodor heveae, did not produce spores on any medium and had rope-like mycelium with coiled structures. Its bioactive volatile metabolites included 3-methylbutan-1-ol as a major component, followed by 3-methylbutyl acetate and azulene derivatives. VOCs from M. heveae show strong activity for the biological control of various bacteria, yeast and filamentous fungi, especially rubber tree pathogens such as R. microporus and Phellinus noxius. Recently, volatile-producing endophytes have attracted great attention with their strong antibiotic activity or other fragrant volatiles can be used in the flavoring industries (Zhi-Lin et al., 2012). Mycofumigation involves the use of antimicrobial volatile compounds from fungi such as *Muscodor* that are applied to control other microorganisms (Stinson et al., 2003a). It is an alternative method for controlling soil-borne pathogens. In 2003, M. albus and M. roseus were used as mycofumigants to control Rhizoctonia solani AG-2, Pythium ultimum and Aphanomyces cochlioides (Stinson et al., 2003a). In addition, mycofumigation by *M. albus* was used to control *Verticillium* wilt and the black dot disease of the potato (Solanum tuberosum L.) caused by Verticillium dahlia and Colletotrichum coccodes, respectively (Grimme, 2008). Worapong and Strobel (2009) reported that *P. ultimum* which had caused root rot disease in kale (*Brassica oleracea*) was controlled by M. albus. Furthermore, M. cinnamomi was used as biofumigants to control Rhizoctonia solani AG-2 damping-off disease of plant seedlings (Suwannarach et al., 2012).

6.2 Materials and methods

6.2.1 Fungal cultures

Muscodor heveae was isolated from the leaves of rubber tree plants in Nongbualamphu Province, northeastern Thailand and stored in 20% (v/v) glycerol at -20 °C at the Sustainable Development of Biological Resources (SDBR) Lab, Department of Biology, Faculty of Science, Chiang Mai University, Chiang Mai, Thailand (SDBR-CMU RTM5IV3), and the BIOTEC (National Center for Genetic Engineering and Biotechnology) Culture Collection (BCC), Bangkok, Thailand (BCC 70461). The morphological characteristics and antimicrobial activity of VOCs had been previously described in chapter 4.

A pathogenic *Rigidoporus microporus* was gently provided by Dr. Nareeluk Nakaew at the Department of Microbiology and Parasitology, Faculty of Science, Naresuan University, Phitsanulok, Thailand. The pathoginicity of the fungus was previously confirmed and described (Nakaew *et al.*, 2015). The culture was stored in 20% (v/v) glycerol at -20 °C. The new cultures were transferred to a potato dextrose agar (PDA, Himedia Laboratories, India) plate and grown for 5–7 days at room temperature (25 ± 2 °C).

6.2.2 Preparation of Muscodor heveae and Rigidoporus microporus inocula

A modified method for solid media preparation was applied following that of Suwannarach *et al.* (2012). Rye grain was boiled for 25 min and 40 g of boiled grain were contained in 250 mL glass bottles. The specimens were then autoclaved at 121 °C for 30 min. The mycelium plugs (5 mm diam) of *M. heveae* were inoculated in each glass bottle and were then incubated in the dark at room temperature for 30 days (Figure 6.1). While, *R. microporus* was cultured on sterilized inoculum medium containing: 100 g sawdust, 3 g rice bran and 2 g sucrose moistened with water. The inoculum was contained in plastic bags and incubated at room temperature for 30 days (Kaewchai and Soytong, 2010) (Figure 6.2).



Figure 6.1 Preparation of Muscodor heveae inocula.



Figure 6.2 Preparation of Rigidoporus microporus inocula.

6.2.3 Effect of VOCs from Muscodor heveae on Rigidoporus microporus in

vivo

Five-month healthy rubber tree seedlings (variety RRIM 600) were placed in 6 inch x 12 inch pots containing 3 kg sterilized soil (soil: commercial soil, 3:1). Fourweek-old rye grain inoculum of M. heveae was scraped, cut and minced into the sterilized soil. The experiment was conducted in 7 treatments with 4 replicates under greenhouse conditions. The treatments conducted were as follows: non-infested control pots contained sterilized mixed soil. Pathogenic treatment involved the inoculation with 1 spawn of R. microporus (R). The Muscodor inoculated control was treated with M. heveae inoculum and sterilized mixed soil at a rate of 40 g / 1 kg⁻¹ (MH40) and 80 g / 1 kg⁻¹ (MH80), respectively (Figure 6.3). Biological treatments involved treating the pathogenic fungus with *M. heveae* inoculum and sterilized mixed soil at 40 g / 1 kg⁻¹ (RMH40) and 80 g / 1 kg⁻¹ (RMH80), respectively. Chemical treatment involved treating the pathogenic fungus with fungicide (75% EC tridemorph) every 2 weeks at 20 ml / L⁻¹ pot⁻¹. Data collection as disease score was recorded at 150 days after treatment. The disease score was categorized as follows: level 1 = healthy green leaves, level 2 = 1-25%yellow leaves, level 3 = 26-50% yellow leaves, level 4 = 51-75% yellow leaves and level 5 = 76-100% yellow leaves. Furthermore, the survival rate (%) of rubber tree seedlings was calculated. The infected roots were observed and the presence of rhizomorph was recorded. The experiment was conducted in a completely randomized design (CRD) with four replicates during the period of July to November 2014 (Kaewchai and Soytong, 2010). The temperature was maintained from 28.4 to 33.5 °C and the relative humidity ranged from 50 to 60 %. The daily light intensity ranged from 2,840 to 8,170 lux.

All rights reserved



The *Muscodor* inoculated control was treated with *M. heveae* inoculum and sterilized soil at a rate of 40 g/ 1 kg⁻¹ (MH40) and 80 g / 1 kg⁻¹ (MH80), respectively.

Figure 6.3 Inoculation method for investigation the effect of VOCs from *Muscodor heveae* on *Rigidoporus microporus in vivo*.

6.2.4 Soil analysis

Soil pH was directly measured by pH-meter (Sartorius PB-10, Germany). Organic matter, total inorganic nitrogen and potassium were analyzed following the method described by Sparks *et al.* (1996). The Bray II method (Houba *et al.*, 1988) was used to measured available phosphorus.

6.2.5 Statistical analysis

SPSS program version 17.0 for Windows was used to analyze the experimental data with one-way analysis of variance (ANOVA), along with Turkey's post hoc tests at different significant levels (P).

6.3 Results

6.3.1 Effect of VOCs from Muscodor heveae on Rigidoporus microporus in

vivo

The volatile metabolites produced by *M. heveae* exhibited antimicrobial activity against mycelial growth of *R. microporus* with 100% inhibition on tested plates

according to the results in section 5.3.1, chapter 5. Furthermore, no growth of *R. microporus* was observed when it was transferred to fresh PDA plates after 6 days of exposure to *M. heveae* volatile metabolites. *In vivo*, *R. microporus* was significantly controlled by soil fumigation with VOCs from *M. heveae* (P<0.05). The highest disease score (4.8) was observed in the pathogenic treatment (R) and this was associated with a low survival rate (25%) (Figure 6.4, Table 6.1). In the biocontrol treatments, RMH40 and RMH80 showed the lowest disease score (1.3) and a high survival rate (100%) that was greater than the chemical treatment (RT) and the non-infested control (Figure 6.4, Table 6.1). However, a statistical comparison demonstrated that RMH40 and RMH80 were not found to be significantly difference from RT and the non-infested control but were found to be significantly difference from the untreated inoculated control (P<0.05). In addition, MH40 and MH80 treatments showed no dead rubber trees after fumigated. The untreated inoculated control showed white mycelia of *R. microporus* colonized on infected root (Figure 6.5).



Figure 6.4 Comparison of disease score for different treatments. Data are presented as means of four replicates. The error bar at each point indicated that \pm SD. The different letters above each graph indicated the means are significantly different by Turkey's test (P<0.05). The disease score was categorized as follows: level 1 = healthy green leaves, level 2 = 1-25% yellow leaves, level 3 = 26-50% yellow leaves, level 4 = 51-75% yellow leaves and level 5 = 76-100% yellow leaves.



Figure 6.5 Biofumigation test for suppression of white root rot disease caused by *Rigidoporus microporus* after 150 days. The treatment protocol were as follows; non-infested control, untreated inoculated control (R), fungicide treatment (RT), *Muscodor* treatment with 40 g 1 kg⁻¹ *M. heveae* inoculum (MH40) and biocontrol treatment with 40 g 1 kg⁻¹ *M. heveae* inoculum (MH40), respectively (A) and the present of rhizomorphs on the infected roots displaying white color (B-C) compared with non-infested control (D).

6.3.2 Soil analysis

Soil pH after used in pot culture values ranged from 6.4 to 7.3. Organic matter was reduced from 5.0 g / $100g^{-1}$ into the values between 1.0 to 1.8 g 100 g⁻¹. The MH80 treatment showed increase values of total NPK at 0.3 g 100 g⁻¹, 69.8 mg kg⁻¹ and 371.5 mg kg⁻¹, respectively. However, the highest exchangeable potassium value was 382.3 mg kg⁻¹ in RMH80 treatment (Table 6.1).

Treatment	Survival rate (%)	Soil chemical analysis				
		рН	Organic matter	Ν	Р	K
	ŝ	110-	(g/100 g)	(g/100 g)	(mg/kg)	(mg/kg)
	8	2		31		
*Soil before used in the pot culture	5	7.1	5.0	0.2	68.1	356.0
R	25	6.6	1.0	0.3	67.7	360.1
MH40	100	7.3	1.6	0.1	58.1	304.2
MH80	75	6.4	1.4	0.3	69.8	371.5
RMH40	100	7.3	1.0	0.2	58.5	379.2
RMH80	100	6.8	1.0	0.2	60.8	382.3
RT	75	6.7		0.2	58.5	379.2

Table 6.1 Survival rate (%) of rubber tree after biofumigation and chemical analysis of soil used in the pot culture.

The treatment protocol were as follows; non-infested control, untreated inoculated control (R), *Muscodor* treatment with 40 g 1 kg⁻¹ *M. heveae* inoculum (MH40), *Muscodor* treatment with 80 g 1 kg⁻¹ *M. heveae* inoculum (MH80), biocontrol treatment with 40 g 1 kg⁻¹ *M. heveae* inoculum (RMH40), biocontrol treatment with 80 g 1 kg⁻¹ *M. heveae* inoculum (RMH40) and fungicide treatment (RT).

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

115

6.4 Discussion

Endophytic fungi can provide benefits to the host plant by providing protection from plant pathogens. It has been suggested that applying a mixture of VOCs is more effective than using them individually (Strobel *et al.*, 2001). White root rot disease has been controlled *in vivo* by various biological control methods such as the application of *Chaetomium cupreum* in both powder and oil form (Kaewchai and Soytong, 2010), the utilization of active rhizospheric *Streptomyces sioyaensis* (Nakaew *et al.*, 2015) and the use of antagonistic *Hypocrea virens* and *H. jecorina* (Ogbebor *et al.*, 2015). This is the first report to demonstrate biofumigation for the control of white root rot disease in rubber trees. *Muscodor heveae* showed strong activity of VOCs in inhibiting *R. microporus in vitro* with 100% growth inhibition and 0% recovery of mycelium growth. There was no growth of *R. microporus* during the 6 day exposure to 10-day-old *M. heveae* culture. After transferred to fresh PDA plates, no growth of *R. microporus* was observed indicating that they were killed after the 6 day exposure of *M. heveae*.

Mycelium penetration and colonization are important stages of root disease infection. It was observed as intra and inner interactions of the plant's cell wall and it has produced degrading enzymes, which correspond to the diverse polymers in the cell wall of host plant (Omorusi, 2012). In the white root rot disease cycle, the infection and colonization of *R. microporus* occurred in the root system of young healthy rubber trees via rhizomorphs that grew from spore attacks or rhizomorphs that survive on the root stumps and wood pieces of infected trees (Omorusi, 2012). Soil fumigation could be an alternative method to inhibited rhizomorph penetration and colonization.

This is of particular interest with regard to the non-accumulation of toxic chemicals in soil over a long period of time. Although VOCs of *M. heveae* affected the root and shoot growth in the some test plants, but they had no effect on tropical plants when fumigation was demonstrated *in vivo*. Because, MH40 and MH80 treatments revealed no dead rubber trees after treatment. Furthermore, plant defense was enhanced by volatile compounds such as 1-octen-3-ol, a major fungal VOC with phytotoxic activity (Splivallo *et al.*, 2007). When, *A. thaliana* was exposured to 1-octen-3-ol, the defense genes associated with wounding or ethylene and jasmonic acid signaling in *A. thaliana* were induced expression and inhibited *Botrytis cinerea* growth on infected leaves

(Kishimoto *et al.*, 2007). The slightly off-green leaves were displayed as a symptom. Then, the green leaves in the unaffected branches turned yellow and the whole canopy was destroyed. This was a possible indication of root disease (Tangonan *et al.*, 2008). Thus, the disease score of the yellow leaf was related to disease control values in each treatment. However, a sign of disease is the presence of rhizomorphs on the infected roots with a white color but that becomes yellow or reddish brown, as the roots grow old (Omorusi, 2012). Only untreated inoculated control (R) showed a sign of root disease. Furthermore, soil chemical characteristics supported the claim that disease symptoms that occurred by the pathogen and did not occur by nutrient deficiency in the soil.

Biofumigation using the genus *Muscodor* for controlling root diseases was demonstrated such as biocontrol of damping-off by *M. albus* (Mercier and Jiménez, 2005), biocontrol of root rot of kale by *M. albus* (Worapong and Strobel, 2009) and *M. cinnamomi* (Suwannarach *et al.*, 2012). Thus, *M. heveae* would be used effectively for controlling white root rot disease in rubber trees, while there was a potential to expand these findings toward further agricultural and industrial applications such as using to control post-harvest decay of fruits and vegetables.

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

The MAI
CHAPTER 7

General discussion and conclusion

This study provide the effective endophytic fungi associated with medicinal plants and the rubber tree in Thailand. Non-volatile producing endophytic fungi were isolated and determined their antagonistic activity. Crude extract of endophytes were examined their antifungal activity by TLC-autobiography. Volatile producing endophytic fungi were analyzed their antagonistic volatile metabolites by GC/MS and used as biological control agents. In addition, plant growth promoting properties were also detected. Morphological characteristic and the rDNA sequence analysis were used to identify isolated endophytes. This chapter concluded the results, discusses the implications of the findings and proposes future work.

7.1 Endophytic fungi in medicinal plants and the rubber tree

Endophytic fungi from medicinal plants were attractive for bioactive metabolite or novel compound discover. Their diversity made a new area of research for novel drug or medicine (Mishra *et al.*, 2015). The present study indicated that the fungal endophyte communites in leaves is diverse more than other part of host plant. However, fungal endophyte presented in all part of host plant (Hyde and Soythong, 2008) and could transfer from one plant to other by seeds (Aly *et al.*, 2011). In this study, most fungal endophytes were isolated from *Clinacanthus nutans* in family *Acanthaceae* (Thai name: Phaya Yo). It was known to have various medicinal properties such as antimicrobial, anticancer, antioxidant and antiviral activities (Kunsorn *et al.*, 2013; Yahaya *et al.*, 2015) and colonized with endophytes such as fungi and bacteria (Jinffeng *et al.*, 2014). While, the effective endophytic fungi with antifungal activity were isolated from short stem of *Aloe vera* L. in family *Lilaceae* (Thai name: Wan Hang Chora Khe). This medicinal plant showed anti-inflammatory activity, immune stimulatory activity and cell growth stimulatory activity, including antiviral and antifungal activity (Sitara *et al.*, 2011). According to previous studies, communities of endophytic fungi could be found in *A. vera* with antimicrobial activity (Jalgaonwala *et al.*, 2010; Bara *et al.*, 2013; Yadav *et al.*, 2015), including other endophyte such as bacteria (Akinsanya *et al.*, 2015; Nongkhlaw and Joshi, 2015). The effective endophytic fungi, AL1T1 and AL1T2 belonged to phylum Ascomycota in the order *Sordariales* based on morphological characteristic and the sequence similarities of ITS rRNA gene (99-100% sequence similarity).

Non-volatile and volatile producing endophytic fungi were isolated from rubber tree (*Hevea brasiliensis* Müll.Arg.). According to the previous studies of endophytic fungi such as *Botryosphaeria*, *Colletotrichum*, *Phomopsis*, *Xylaria* (Gamboa *et al.*, 2002; Arnold and Lutzoni, 2007), *Penicillium paxilli*, *Pestalotiopsis* aff. *palmarum*, *Trichoderma harzianum* (Gazis and Chaverri, 2010), *Trichoderma amazonicum* (Chaverri *et al.*, 2011), including various species of arbuscular mycorrhizal fungi (AMF, Pereira *et al.*, 2014) were found in this economic plant. However, it is the first report to isolated volatile-producing endophytic fungi, *Muscodor* spp. from leaves of the rubber tree.

7.2 Antifungal activity and plant growth promoting properties of endophytic fungi

The extracts from many types of local medicinal plants were used in traditional manner for treatments of various disease. The bioactive compounds were produced by the plant but in a mutualistic association with the host plant, endophytes may enhance bioactive metabolites that process bioactivity such as antibacterial and antifungal activity in host plant (Radu and Kqueen, 2002). It was supported by many studies, such as it found that hydroalcoholic, the extract from leaves of *A. vera* could inhibit mycelial growth of various fungi such as *Botrytris gladiolorum*, *Fusarium oxysporum*, *Heterosporium pruneti* and *Penicillium gladioli* (Jasso *et al.*, 2005). While, *A. vera* gel extract was evaluated for their antifungal activity against *Aspergillus niger*, *A. flavus*, *Alternaria alternate*, *Drechslera hawaiensis* and *P. digitatum* (Sitara *et al.*, 2011). According to this study, isolate AL1T1 and AL1T2 showed strong antagonistic activity against several phytopathogenic fungi in the rubber tree, including *F. oxysporum* which have been inhibited by the extract from *A. vera* leaves (Jasso *et al.*, 2005). Furthermore, Bara *et al.* (2013) showed that *Talaromyces worthmanii*, *A. vera* endophytic fungus could produce

bioactive compounds such as atropsiomer and worthmannin derivative with antimicrobial activity. Endophytic fungi, which was isolated from *Bauhinia strychnifolia* (BS1B201), *Clinacanthus nutans* (C4IV301) and *Pseuderanthemum palatiferum* (PS1IV102) exhibited plant growth promote property via produced indole acetic acid (IAA). This result was supported by the previous studies that showed several endophytic fungi, which produced IAA and increased seed germination and plant growth (Khan *et al.*, 2012, 2015, 2016; Zhou *et al.*, 2013; Kedar *et al.*, 2014). Future study on purification of effective metabolites and anti-cancer, anti-oxidant, anti-viral of their compounds will be evaluated.

7.3 Molecular and morphological identification of genus Muscodor

Volatile-producing endophytic fungus, *Muscodor* is classified into the family Xylariaceae and does not produced spores (Kudalkar et al., 2011). It could be isolated from several host plants and their VOCs are active against many pathogenic microorganisms such as bacteria, fungi, nematodes and insects. (Worapong et al., 2001; Suwannarach et al., 2013a; Saxena et al., 2015). Moreover, some species of Muscodor produced volatile metabolites which exhibit phytotoxic activity (Gonzalez et al., 2009; Macías-Rubalcava et al., 2010). It was difficult to classify Muscodor by using only morphological characteristics because it did not produce any reproductive structures. Thus, a more appropriate genotypic classification method was needed using more genes than ITS for identification at the species level and to understand their phylogenetic lineage. Moreover, profiling and comparison of unique volatile metabolites produced by members of the genus Muscodor were important criteria for its classification (Strobel, 2006; González et al., 2009). In this study, M. vitigenus, M. equiseti and a novel species, M. heveae were isolated from leaves of the rubber tree in northeastern Thailand. Based on morphological and physiological characteristics, profile of volatile metabolites produced and a molecular phylogenetic analysis. In addition, M. vitigenus and M. qeuiseti have been isolated from *Paullinia paullinoides* (Daisy et al., 2002) and *Equisetum debile* (Suwannarach et al., 2013a), respectively. Isolation of 3 species of Muscodor from the rubber tree indicates a possible susceptibility of this plant to colonize by members of this genus of mycelia sterilia endophytes.

7.4 Antagonistic volatile compounds of Muscodor heveae and revelant alleopathy

The mixture of volatiles with bioactive activity have been produced by several volatile-producing endophytic fungi such as Botrytis sp. (Ting et al., 2010), Hypoxylon sp. (Tomsheck et al., 2010), Nodulisporium sp. (Mends et al., 2012), Phoma sp. (Strobel et al., 2011), Phomopsis sp. (Singh et al., 2011), including Muscordor spp. (Worapong et al., 2001; Suwannarach et al., 2013a; Saxena et al., 2015). They had potential as harmful pesticides, fungicides and bactericides as well as genetic modification (Kanchiswamy et al., 2015). The present study indicated that VOCs from Muscodor spp. which isolated from leaves of the rubber tree inhibited growth of various fungi and bacteria. F. oxysporum f. sp. vasinfectum was more susceptible to the volatile metabolites of M. heveae than M. vitigenus and M. geuiseti. The artificial volatile compounds test confirmed that the major component, 3-methylbutan-1-ol, and the minor compounds, 3-methylbutyl acetate and 2-methylpropanoic acid exhibited antifungal activity and phytotoxic activity in tested fungal pathogens and plants, according to previous studies of Gonzalez et al. (2009) and Macías-Rubalcava et al. (2010). Future study on the application of volatile metabolites from M. heveae in biocontrol of weeds in vivo as potential herbicides and controlling postharvest disease.

7.5 Controlling of white root rot disease in rubber tree using volatile compounds

UNIV

from Muscodor heveae

Mycofumigation involved the use of antimicrobial volatile compounds from fungi such as *Muscodor* that were applied to control other microorganisms, including soil-borne pathogens (Stinson *et al.*, 2003a). VOCs of *M. albus* and *M. roseus* were used as mycofumigants to control *Rhizoctonia solani* AG-2, *Pythium ultimum* and *Aphanomyces cochlioides* (Stinson *et al.*, 2003a). While, *M. albus* inoculum was used to control *Verticillium* wilt and the black dot disease of the potato (Grimme, 2008), including root rot disease in kale (Worapong and Strobel, 2009). Furthermore, *M. cinnamomi* were used as biofumigants to control *Rhizoctonia solani* AG-2 damping-off disease of plant seedlings (Suwannarach *et al.*, 2012). In this study, a mixture of VOCs from *M. heveae* exhibited antagonistic activity against the mycelial growth of phytogenic fungi, which could be used as a biological control agent. The SPME-GC/MS analysis indicated that the major volatile compound produced by *M. heveae* with the highest percentage peak

area was 3-methylbutan-1-ol. The minor volatile compounds were 3-methylbutyl acetate and 2-methylpropanoic acid, respectively. It was supported by the previous studies, the bioactive volatile components for the genus *Muscodor* are 2-methylpropanoic acid, 3methylbutyl acetate and 3-methyl-1-butanol (3-methylbutan-1-ol) (Strobel, 2006; Mitchell *et al.*, 2010). Variation of volatile metabolites-profile was commonly found across member species of the genus *Muscodor*, but they often share the same major volatile component (Kudalkar *et al.*, 2011; Suwannarach *et al.*, 2013a).

Biofumigation by *M. heveae* showed great suppression of the disease. Biocontrol treatments; RMH40 (40 g 1 kg⁻¹ *M. heveae* inoculum) and RMH80 (80 g 1 kg⁻¹ *M. heveae* inoculum) were not found to be significantly different when compared with fungicide treatment (RT) and the non-infested control, but results were found to be significantly different from those of the untreated inoculated control (R) (P<0.05). According to the biofumigation of mixed volatile, including 2-methylpropanoic acid and 3-methylbutan-1-ol of *M. suthepensis* for controlling postharvest disease (Suwannarach, 2013). Further study on the optimal techniques of inocula production will be studied with the goal of finding low cost and easily maintained production methods and application of mixed culture of volatile-producing microorganism for biofumigants on a commercial agriculture scale in Thailand.

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

MAI UNIVER

REFERENCES

- Abdel-Motall FF, Nassar MSM, El-Zayat SA, El-Sayad MA, Ito S (2010) Antifungal activity of endophytic fungi isolated from Egyptian henbane (*Hyoscyamus muticus* L.). *Pakistan Journal of Botany* 42: 2883-2894.
- Ann PJ, Chang TT, Ko WH (2002) *Phellinus noxius* brown root rot of fruit and ornamental trees in Thaiwan. *Plant disease* 8: 820-826.
- Ahmad F, Ahmad I, Khan MS (2005) Indole acetic acid production by the indigenous isolates of *Azotobacter* and fluorescent *Pseudomonas* in the presence and absence of tryptophan. *Turkish Journal of Biology* 29: 29-34.
- Alvin A, Miller KI, Neilan BA (2014) Exploring the potential of endophytes from medicinal plants as sources of antimycobacterial compounds. *Microbiological Research* 169: 483-495.
- Aly AH, Debbab A, Proksch P (2011) Fungal endophytes: unique plant inhabitants with great promises. *Applied Microbiology and Biotechnology* 90: 1829-1845.
- Akinsanya MA, Goh JK, Lim SP, Ting ASY (2015) Metagenomics study of endophytic bacteria in *Aloe vera* using next-generation technology. *Genomics data* 6: 159-163.
- Arnold AE, Lutzoni F (2007) Diversity and host range of foliar fungal endophytes: are tropical trees biodiversity hotspots? *Ecology* 88: 541-549.
- Atmosukarto I, Castillo U, Hess WM, Sears J, Strobel G (2005) Isolation and characterization of *Muscodor albus* I-41.3s, a volatile antibiotic producing fungus. *Plant Science* 169: 854-861.

- Atugala DM, Deshappriya N (2015) Effect of endophytic fungi on plant growth and blast disease incidence of two traditional rice varieties. *Journal of the National Science Foundation of Sri Lanka* 43: 173-187.
- Bailey BA, Strem MD, Wood D (2009) Trichoderma species form endophytic associations within Theobroma cacao trichomes. Mycological Research 113: 1365-1376.
- Bara R, Aly AH, Pretsch A, Wray V, Wang B, Proksch P, Debbab A (2013) Antibiotically active metabolites from *Talaromyces wortmannii*, an endophyte of *Aloe vera*. *The Journal of Antibiotics* 66: 491-493.
- Backman PA, Sikora RA (2008) Endophytes: an emerging tool for biological control. *Biological Control* 46: 1-3.
- Banerjee D, Strobel GA, Booth E, Geary B, Sears J, Spakoxicz D, Busse S (2010) An endophytic *Myrothecium inundatum* producing volatile organic compounds. *Mycosphere* 1: 229-240.
- Banguela-Castillo A, Ramos-González PL, Penã-Marey M, Tanaka FAO, Blassioli-Moraes MC, Hernández-Rodríguez L, Cabrera RI (2015) The same host but a different *Muscodor*: a new *Muscodor albus* isolate from wild pine apple (*Ananas ananassoides*) with potential application in agriculture. *Crop protection* 78: 284-292.
- Bhavya M, Mohanapriya P, Pazhanimurugan R, Balagurunathan R (2011) Potential bioactive compound from marine actinomycetes against biofouling bacteria. *Indian Journal of Geo-Marine Sciences* 40: 578-582.
- Beena KR, Ananda K, Sridhar KR (2000) Fungal endophytes of three sand dune plant species of west coast of India. *Sydowia* 52: 1-9.
- Benítez T, Rincón MA, Limón MC, Codón CA (2004) Biocontrol mechanisms of *Trichoderma* strains. *International Microbiology* 7: 249-260.

Cannon CD, Simmons C M (2002) Diversity and host preference of leaf endophytic fungi in the Iwokrama forest reserve, Guyana. *Mycologia* 94: 210-220.

Cannon PF, Kirk PM (2007) Fungal Families of The World. CABI, Wallingford, pp.463.

- Cardinal ABB, Goncalves P de S, Martins ALM (2007) Stock-scion interactions on growth and rubber yield of *Hevea brasiliensis*. Scientia Agricola 64: 235-240.
- Chaverri P, Gazis RO (2011) Trichoderma amazonicum, a new endophytic species on Hevea brasiliensis and H. guianensis from the Amazon basin. Mycologia 103: 139-151.
- Clay K (1988) Fungal endophytes of grasses: a defensive mutualism between plants and fungi. *Ecology* 69: 10-16.
- Cícero SM, Marcos-Filho J (2010) Rubber Tree Seed Production. Website: http://seedbiology.osu.edu/HCS630_files/May%2031/Rubber%20seed %20production%20-%20text.pdf, 11 June 2015.
- Cohen SD (2006) Host selectivity and genetic variation of *Discula umbrinella* isolates from two oak species: analyses of intergenie spacer region sequences of ribosomal DNA. *Microbial Ecology* 52: 463-469.
- Daisy B, Strobel G, Ezra D, Castillo U, Baird G, Hess WM (2002) *Muscodor vitigenus* anam. sp. nov., an endophyte from *Paullinia paullinioides*. *Mycotaxon* 84: 39-50.
- Desale MG, Bodhankar MG (2013) Antimicrobial activity of endophytic fungi isolated from Vitex negundo Linn. International Journal of Current Microbiology and Applied Sciences 2: 389-395.
- Dickson IA, Okere A, Elizabeth J, Mary O., Olatunde F, Abiodun S (2011) *In vitro* culture of *Hevea brasiliensis* (rubber tree) embryo. *Journal of Plant Breeding and Crop Science* 3: 185-189.

- Dreyfuss MM, Chapela IH (1994) Potential of fungi in the discovery of novel lowmolecular weight pharmaceuticals. In: Gullo VP (Eds.), *The discovery of natural products with therapeutic potential*, London, UK: Butterworth-Heinemann, pp. 49-80.
- Ehmann A (1977) The van Urk-Salkowski reagent-a sensitive and specific chromogenic reagent for silica gel thin layer chromatographic detection and identification of indole derivatives. *Journal of Chromatography* 132: 267-276.
- Espinel MA, Laszlo A, Simonsen L, Boulahbal F, Kim SJ, Reniero A, Hoffner S, Rieder HL, Binkin N, Dye C, Williams R, Raviglione MC (2001) Global trends in resistance to antituberculosis drugs. *The New England Journal of Medicine* 344:1294-1303.
- Evueh GA, Ogbebor NO (2008) Use of phylloplane fungi as biocontrol agent against *Colletotrichum* leaf disease of rubber (*Hevea brasiliensis* Muell. Arg.). *African Journal of Biotechnology* 7: 2569-2572.
- Ezra D, Hess WM, Strobel GA (2004) New endophytic isolates of *Muscodor albus*, a volatile-antibiotic-producing fungus. *Microbiology* 150: 4023-4031.
- Ezra D, Strobel GA (2003) Effect of substrate on the bioactivity of volatile antimicrobials produced by *Muscodor albus*. *Plant Science* 165: 1229-1238.
- Fialho MB, Duarte de Moraes MH, Tremocoldi AR, Pascholati SF (2011) Potential of antimicrobial volatile organic compounds to control Sclerotinia sclerotiorum in bean seeds. Pesquisa Agropecuária Brasileira 46: 137-142.
- Gamboa MA, Laureano S, Bayman P (2002) Measuring diversity of endophytic fungi in leaf fragments: does size matter? *Mycopathologia* 156: 41-45.
- Garyali S, Kumar A, Reddy MS (2013) Taxol production by an endophytic fungus, *Fusarium redolens*, isolated from Himalayan yew. *Journal of Microbiology and Biotechnology* 23: 1372-1380.

- Gazis RO (2012) Evaluating the Endophytic Fungi Community in Planted and Wild Rubber Trees (*Hevea brasiliensis*). PhD Thesis, University of Maryland.
- Gazis R, Chaverri P (2010) Diversity of fungal endophytes in leaves and stems of wild rubber trees (*Hevea brasiliensis*) in Peru. *Fungal Ecology* 3: 240-254.
- Geiger JP, Rio B, Nicole M, Nandris D (1986) Biodegradation of Hevea brasiliensis wood by Rigidoporus lignosus and Phellinus noxius. European Journal of Plant Pathology 16: 147-159.
- González MC, Anaya AL, Glenn AE, Macías-Rubalcava ML, Hernández-Bautista BE, Hanlin RT (2009) Muscodor yucatanensis, a new endophytic ascomycete from Mexican chakah, Bursera simaruba. Mycotaxon 110: 363-372.
- Grimme E (2008) Mycofumigation with Muscodor albus: Effects on Verticillium wilt and Black Dot Root Rot of Potato, Effects on Glomus intraradices and Ectomycorrhizal Fungi, and M. albus Proliferation in Soil. PhD thesis, Montana State University, U.S.A.
- Guo B, Dai J, Ng S, Huang Y, Leong C, Ong W, Carte BK (2000) Cytonic acids A and B: novel tridepside inhibitors of hCMV protease from the endophytic fungus Cytonaema species. Journal of Natural Products 63: 602-604.
- Guyot J, Flori A (2002) Comparative study for detecting *Rigidoporus lignosus* on rubber trees. *Crop Protection* 21: 461-466.
- Guzmán-Trampe S, Rodríguez-Penã K, Espinosa-Gómez A, Sánchez-Fernández RE, Macías-Rubalcava ML, Flores-Cotera LB, Sánchez S (2015).
 Endophytes as a potential source of new antibiotics. In: Sánchez S, Demain AL (Eds.), *Antibiotics, Current Innovations and Future Trends*, Norfolk: Caister Academic Press., pp.175–204.

- Hamayun M, Khan SA, Khan AL, Rehman G, Kim YH, Iqbal I, Hussain J, Sohn EY, Lee IJ (2010) Gibberellins production and plant growth promotion by pure cultures of *Cladosporium* sp. MH-6 isolated from Cucumber (*Cucumis sativus* L.). *Mycologia* 102: 989-995.
- Harman GE, Howell CR, Viterbo A, Chet I, Lorito M (2004) Trichoderma species opportunistic, avirulent plant symbionts. Natural Reviews Microbiology 2: 43-56.
- Hering O, Nirenberg HI, Köhn S, Deml G (1999) Characterization of isolates of *Fusarium* oxysporum Schlecht. f. sp. vasinfectum (Atk.) Snyd. & Hans., races 1-6, by cellular fatty acid analysis. *Journal of Phytopathology* 147: 509-514.
- Hosseyni-Moghaddam MS, Soltani J (2014) Bioactivity of endophytic Trichoderma fungal species from the plant family Cupressaceae. Annals of Microbiology 64:753-761.
- Houba VJG, van Der Lee JJ, Novozamsky I, Wallinga J (1988) Soil and plant analysis part 5. In: Soil Analysis Procedures, Wageningen: Agricultural University, pp. 154-158.
- Huang Y, Wang J, Li G, Zheng Z, Su W (2001) Antitumor and antifungal activities in endophytic fungi isolated from pharmaceutical plants *Taxus mairei*, *Cephalataxus fortunei* and *Torreya grandis*. *FEMS Immunology and Medical Microbiology* 31: 163-167.
- Huang Z, Cai X, Shao C, She Z, Xia X, Chen Y, Yang J, Zhou S, Lin Y (2008) Chemistry and weak antimicrobial activities of phomopsins produced by mangrove endophytic fungus *Phomopsis* sp. ZSU-H76. *Phytochemistry* 69: 1604-1608.
- Humphris SN, Bruce A, Buultjens E, Wheatley RE (2002) The effects of volatile microbial secondary metabolites on protein synthesis in Serpula lacrymans. FEMS Microbiology Letters 210: 215-219.

- Hung R, Lee S, Bennett JW (2013) Arabidopsis thaliana as a model system for testing the effect of Trichoderma volatile organic compounds. Fungal Ecology 6: 19-26.
- Hyde KD, Soythong K (2008) The fungal endophyte dilemma. *Fungal Diversity* 33: 163-173.
- Idris A, Al-tahir I, Idris E (2013) Antibacterial activity of endophytic fungi extracts from the medicinal plant *Kigelia africana*. *Egyptian Academic Journal of Biological Science* 5: 1-9.
- Jalgaonwala RE, Mohite BV, Mahajan RT (2010) Evaluation of Endophytes for their antimicrobial activity from indigenous medicinal plants belonging to north Maharashtra region India. *International Journal on Pharmaceutical and Biomedical Research* 5: 136-141.
- Jalgaonwala RE, Mohite VB, Mahajan RT (2011) A review: natural products from plant associated endophytic fungi. *Journal of Microbiology and Biotechnology Research* 1: 21-32.
- Jasso de Rodriguez D, Hernandez-Castillo D, Rodriguez-Gracia R, Angulo-Sanchez JL (2005) Antifungal activity *in vitro* of *Aloe vera* pulp and liquid fraction against plant pathogenic fungi. *Industrial Crops and Products* 21: 81-

87.

- Jayasinghe CK (2010) White root disease of rubber tree: an overview. Proceeding of International workshop on white root rot disease of *Hevea* ruber, 14th-16th December, Sri-Lanka, pp. 1-8.
- Jayasuriya KE, Deacon JW (1995) In vitro interactions between Rigidoporus lignosus, the cause of white root disease of rubber and some potentially antagonistic fungi. Journal of the Rubber Research Institute of Sri Lanka 76: 36-54.
- Jayasuriya KE, Thennakoon BI (2007) Biological control of *Rigidoporus microporus*, the cause of white root disease in rubber. *Ceylon Journal of Science* (*Biological Sciences*) 36: 9-16.

- Jinffeng EC, Cheah YK, Hoon KC, Tan SG, Ho KL (2014) Isolation, Diversity and Antibacterial Activity of Endophytes Isolated from *Clinacanthus nutans*, a Well-known Medicinal Plant. Website: https://www.resear chgate.net/publication/305728447_Isolation_Diversity_and_Antibacte rial_Activity_of_Endophytes_Isolated_from_Clinacanthus_nutans_A_ Well-known_Medi cinal_Plant, 11 October 2016.
- Kaewchai S (2013) White root disease of rubber tree and control. *Princess of Naradhiwas* University Journal 5: 118-131.
- Kaewchai S, Soytong K (2010) Application of biofungicides against Rigidoporus microporus causing white root disease of rubber trees. Journal of Agricultural Technology 2: 349-363.
- Kanchiswamy CN, Malnoy M, Maffei ME (2015) Chemical diversity of microbial volatiles and their potential for plant growth and productivity. *Frontiers in Plant Science* 6: 1-23.
- Karunai SB, Balagengatharathilagam P (2014). Isolation and screening of endophytic fungi from medicinal plants of Virudhunagar district for antimicrobial activity. *International Journal of Science and Nature* 5: 147-155.
- Kedar A, Rathod D, Yadav A, Agarkar G, Rai M (2014) Endophytic *Phoma* sp. isolated from medicinal plants promote the growth of *Zea mays*. *Nusantara Bioscience* 6:132-139.
- Khan AL, Gilani SA, Waqas M, Al-Hosni K, Al-Khiziri S, Kim Y, Ali L, Kang S, Asaf S, Shahzad R, Hussain J, Lee I, Al-Harrasi A (2016) Endophytes from medicinal plants and their potential for producing indole-acetic acid, improving seed germination and mitigating oxidative stress. *Journal of Zhejiang University Science B.*, http://dx.doi.org/10.1631/jzus.B15002.

71.

- Khan SA, Hamayun M, Khan AL, Lee I, Shinwari ZK, Kim J (2012) Isolation of plant growth promoting endophytic fungi from dicots inhabiting coastal sand dunes of Korea. *Pakistan Journal of Botany* 44:1453-1460.
- Khan AR, Ullah I, Waqas M, Shahzad R, Hong SJ, Park GS, Jung BK, Lee IJ, Shin JH (2015) Plant growth-promoting potential of endophytic fungi isolated from Solanum nigrum leaves. World Journal of Microbiology and Biotechnology 31: 1461-1466.
- Kirk TK (1971) Effect of microorganisms on ligin. *Annaual Review of Phytopathology* 9: 185-210.
- Kim BS, Hwang BK (2007) Microbial fungicides in the control of plant diseases. *Journal* of Phytopathology 155: 641-653.
- Kishimoto K, Matsui K, Ozawa R, Takabayashi J (2007). Volatile 1-octen-3-ol induces a defensive response in Arabidopsis thaliana. Journal of General Plant Pathology 73: 35-37.
- Ko J, Chow K, Han K (2003) Transcriptome analysis reveals novel features of the molecular events occurring in the laticifers of *Hevea brasiliensis* (para rubber tree). *Plant Molecular Biology* 53: 479-492.
- Kodsueb R, McKenzie EHC, Lumyoung S, Hyde KD (2008) Diversity of saprobic fungi on Magnoliaceae. *Fungal Diversity* 30: 37-53.
- Korpi A, Järnberg J, Pasanen AL (2009) Microbial volatile organic compounds. *Critical Reviews in Toxicology* 39: 139-193.
- Kudalkar P, Strobel G, Riyaz-Ul-Hassan S, Geary B, Sears J (2011) Muscodor sutura, a novel endophytic fungus with volatile antibiotic activities. Mycoscience 53: 319-325.
- Kumar M, Yadav V, Kumar H, Sharma R, Singh A, Tuteja N, Johri AK (2011) *Piriformospora indica* enhances plant growth by transferring phosphate. *Plant Signaling & Behavior* 6: 723-725.

- Kunsorn P, Ruangrungsi N, Lipipun V, Khanboon A, Rungsihirunrat K (2013) The identities and anti-herpes simplex virus activity of *Clinacanthus nutans* and *Clinacanthus siamensis*. Asian Pacific Journal of Tropical Biomedicine 3: 284-290.
- Lee S, Hung R, Schink A, Mauro J., Bennett JW (2014) *Arabidopsis thaliana* for testing the phytotoxicity of volatile organic compounds. *Plant Growth Regulation* 74: 177-186.
- Li E, Jiang L, Guo L, Zhang H, Che Y (2008) Pestalachlorides A-C, antifungal metabolites from the plant endophytic fungus *Pestalotiopsis adusta*. *Bioorganic & Medicinal Chemistry* 16: 7894-7899.
- Li H, Wei J, Pan S, Gao J, Tian J (2014a) Antifungal, phytotoxic and toxic metabolites produced by *Penicillium purpurogenum*. *Natural Product Research* 28: 2358-2361.
- Li H, Xia J, Gao YQ, Tang JJ, Zhang AL, Gao JM (2014b) Chaetoglobosins from *Chaetomium globosum*, an endophytic fungus in *Ginkgo biloba*, and their phytotoxic and cytotoxic activities. *Journal of Agricultural and Food Chemistry* 62: 3734-3741.
- Li X, Zhang Q, Zhang A, Gao J (2012) Metabolites from *Aspergillus fumigatus*, an endophytic fungus associated with Melia *azedarach*, and their antifungal, antifeedant, and toxic activities. *Journal of Agricultural and Food Chemistry* 60: 3424-3431.
- Liu K, Zheng Y, Miao C, Xiong Z, Xu L, Guan H, Yang Y, Zhao L (2014) The antifungal metabolites obtained from the rhizospheric Aspergillus sp. YIM PH30001 against pathogenic fungi of Panax notoginseng. Natural Product Research 28: 2334-2337.
- Lu Y, Chen C, Chen H, Zhang J, Chen W (2012) Isolation and Identification of endophytic fungi from Actinidia macrosperm and investigation of their bioactivities. Evidence Based Complementary and Alternative Medicine 2012: 1-8.

- Macías-Rubalcava ML, Hernández-Bautista BE, Oropeza F, Duarte G, González MC, Glenn AE, Hanlin RT, Anaya AL (2010) Allelochemical effects of volatile compounds and organic extracts from *Muscodor yucatanensis*, a tropical endophytic fungus from *Bursera simaruba*. Journal of Chemical Ecology 36: 1122-1131.
- Martins MBG, Zieri R (2003) Leaf anatomy of rubber-tree clones. *Scientia Agricola* 70: 709-713.
- Mejia LC, Rojas EI, Maynard Z, Arnold AE, Kyllo D, Robbins N, Herre EA (2003) Inoculation of beneficial endophytic fungi into *Theobroma cacao* tissues. Proceedings of the 14th International Cocoa Research Conference, Accra-Ghana.
- Mends MT, Yu E, Strobel GA, Riyaz-Ul-Hassan S, Booth E, Geary B, Sears J, Taatjes CA, Hadi MZ (2012) An Endophytic Nodulisporium sp. producing volatile organic compounds having bioactivity and fuel potential. Journal of Petroleum & Environmental Biotechnology 33: 1-7.
- Mercier J, Jiménez JI (2004) Control of fungal decay of apples and peaches by the biofumigant fungus *Muscodor albus*. *Postharvest Biology and Technology* 31: 1-8.
- Mercier J, Jiménez JI (2007) Potential of the volatile-producing fungus *Muscodor albus* for control of building molds. *Canadian Journal of Microbiology* 53: 404-410.
- Mercier J, Jiménez-Santamaría JI, Tamez-Guerra P (2007) Development of the volatileproducing fungus *Muscodor albus* Worapong, Strobel, and Hess as a novel antimicrobial biofumigant. *Revista Mexicana de Fitopatología* 25: 173-179.
- Mercier J, Manker DC (2005) Biological control of soil-borne diseases and plant growth enhancement in greenhouse soilless mix by the volatile-producing fungus *Muscodor albus*. *Crop Protection* 24: 355-362.

- Meshram V, Kapoor N, Saxena S (2013) *Muscodor kashayum* sp. nov. a new volatile antimicrobial producing endophytic fungus. *Mycology* 4: 196-204.
- Meshram V, Saxena S, Kapoor N (2014) *Muscodor strobelii*, a new endophytic species from South India. *Mycotaxon* 128: 93-104.
- Meshram V, Gupta M, Saxena S (2015) Muscodor ghoomensis and Muscodor indica: new endophytic species based on morphological features and molecular and volatile organic analysis from Northeast India. Sydowia 67: 133-146.
- Mishra M, Prasad R, Varma A (2015) Endophytic fungi: biodiversity and functions. International Journal of Pharma and Bio Sciences 1: 18-36.
- Mitchell AM, Strobel GA, Hess WM, Vargas PN, Ezra D (2008) Muscodor crispans, a novel endophyte from Ananas ananassoides in the Bolivian Amazon. Fungal Diversity 31: 37-43.
- Mitchell AM, Strobel GA, Moore E, Robison R, Sears J (2010) Volatile antimicrobials from *Muscodor crispans*, a novel endophytic fungus. *Microbiology* 156: 270-277.
- Monkai J, Hyde KD, Xu J, Mortimer PE (2016) Diversity and ecology of soil fungal communities in rubber plantations. *Fungal Biology Reviews*, http://dx.doi.org/10.1016/j.fbr.2016.08.003.
- Morath SU, Hung R, Bennett JW (2012) Fungal volatile organic compounds: a review with emphasis on their biotechnological potential. *Fungal Biology Reviews* 26: 73-83.
- Moricca S, Ragazzi A, Kasuga T, Mitchelson KR (1998) Detection of Fusarium oxysporum f. sp. vasinfectum in cotton tissue by polymerase chain reaction. Plant Pathology 47: 486-495.

- Moreno E, Varughese T, Spadafora C, Arnold E, Coley PD, Kursar TA, Gerwick WH, Cubilla-Riosa L (2011) Chemical constituents of the new endophytic fungus *Mycosphaerella* sp. nov. and their anti-parasitic activity. *Natural Product Communications* 6: 835-840.
- Mulaw TB, Druzhinina IS, Kubicek CP, Atanasova Lea (2013) Novel endophytic *Trichoderma* spp. isolated from healthy *Coffea arabica* roots are capable of controlling coffee tracheomycosis. *Diversity* 5: 750-766.
- Nakaew N, Rangjaroen C, Sungthong R (2015) Utilization of rhizospheric Streptomyces for biological control of Rigidoporus sp. causing white root disease in rubber tree. European Journal of Plant Pathology 142: 93-105.
- Nandris D, Nicole M, Geiger JP (1988) Root rot diseases of rubber tree. I. Severity, dynamics an characterization of epidemics. *Canadian Journal of Forest Research* 18: 1248-1254.
- Naznin HA, Kiyohara D, Kimura M, Miyazawa M, Shimizu M, Hyakumachi M (2014) Systemic resistance induced by volatile organic compounds emitted by plant growth promoting fungi in *Arabidopsis thaliana*. *PLoS One* 9: 1-10.
- Newcombe G, Shipunov A, Eigenbrode S, Raghavendra AKh, Ding H, Anderson CL, Menjivar R, Crawford M, Schwarzländer M. (2009) Endophytes influence protection and growth of an invasive plant. *Communitative & Integrative Biology* 2: 29-31.
- Nongkhlaw FMW, Joshi SR (2015) Investigation on the bioactivity of culturable endophytic and epithytic bacteria associated with ethnomedicinal plants. *The Journal of Infection in Developing Countries* 9: 954-961.
- Ogbebor NO, Adekunle AT, Eghafona ON, Ogboghodo AI (2013) Incidence of *Rigidoporus lignosus* (Klotzsch) Imaz. of para rubber in Nigeria. *Researcher* 5: 181-184.

- Ogbebor NO, Adekunle AT, Eghafona ON, Ogboghodo AI (2015) Biological control of Rigidoporus lignosus in Hevea brasiliensis in Nigeria. Fungal Biology 119: 1-6.
- Oghenekaro AO, Daniel G, Asiegbu FO (2015) The saprotrophic wood-degrading abilities of Rigidoporus microporus. Silva Fennica 49: 1-10.
- Omorusi VI (2012) Effects of white root rot disease on Hevea brasiliensis (Muell. Arg.)challenges and control approach. In: Nabin KD and Sudam CS (Eds.), Plant Science, Croatia: InTech.
- Ortiz-Castro R, Contreras-Cornejo HA, Macías-Rodríguez L, López-Bucio J (2009) The role of microbial signals in plant growth and development. Plant Signaling & Behavior 4: 701-712.
- Pagans E, Font X, Sánchez A (2006) Emission of volatile organic compounds from composting of different solid wastes: abatement by biofiltration. Journal of Hazardous Materials 131: 179-186.
- Panthong A, Kanjanapothi D, Taesotikul T, Taylor WC (1991) Ethnobotanical review of medicinal plants from Thai traditional books, Part II: Plants with antidiarrheal, laxative and carminative properties. Journal of Ethnopharmacology 31: 121-156.
- Pereira CMR, ds Silva DKA, Ferreira ACdeA, Goto BT, Maia LC (2014) Diversity of arbuscilar mycorrhizal fungi in Atlantic forest areas under different CODland uses. Agriculture, Ecosystems & Environment 185: 245-252.
- FASAFVAA Prakash V (2015) Endophytic fungi as resource of bioactive compounds. International Journal of Phama and Bio Sciences 6: 887-898.

righte

Prakash S, Ramasubburayan R, Iyapparaj P, Kumar C, Mary CJ, Palavesam A, Immanuel G (2013) Screening and partial purification of antifungal metabolite from Streptomyces rochei MSA 14: an isolate from marine mining soil of Southwest coast India. Indian Journal of Geo-Marine Sciences 42: 888-897.

- Powthong P, Jantrapanukorn B, Thongmee A, Suntornthiticharoen P (2013) Screening of antimicrobial activities of the endophytic fungi isolated from *Sesbania* grandiflora (L.) Pers. Journal of Agricultural Science and Technology 15: 1513-1522.
- Radu S, Kqueen CY (2002) Preliminary screening of endophytic fungi from medicinal plants in Malaysia for antimicrobial and antitumor activity. *Malaysian Journal of Medical Sciences* 9: 23-33.
- Rajalakshmy VK, Jayarathnam K (2000) Root diseases and non- microbial maladies. In: George PJ and Kuruvilla Jacob C (Eds.), Natural Rubber: Agromanagement and Crop Processing, Kerala: Rubber Research Institute of India.
- Ramin AA, Braun PG, Prange RK, DeLong JM (2005) In vitro effects of Muscodor albus and three volatile components on growth of selected postharvest microorganisms. HortScience 40: 2109-2114.
- Rebecca AIN, Kumar DJM, Srimathi S, Muthumary J, Kalaichelvan PT (2011) Isolation of *Phoma* species from *Aloe vera*: an endophyte and screening the fungus for taxol production. *World Journal of Science & Technology* 1: 23-31.
- Rocha ACS, Garcia D, Uetanabaro APT, Carneiro RTO, Araújo IS, Mattos CRR, Góes-Neto A (2010) Foliar endophytic fungi from *Hevea brasiliensis* and their antagonism on *Microcyclus ulei*. *Fungal Diversity* 47: 75-84.
- Rodriguez RJ, Henson J, van Volkenburgh E, Hoy M, Wright L, Beckwith F, Kim Y, Redman RS (2008) Stress tolerance in plants via habitat-adapted symbiosis. *The ISME Journal* 2: 404-416.
- Rodriguez RJ, White Jr JF, Arnold AE, Redman RS (2009) Fungal endophytes: diversity and functional roles. *New Phytologist* 182: 314-330.

- Rosa LH, Tanabca N, Techen N, Pan z, Wedge DE (2012) Antifungal activity of extracts from endophytic fungi associated with *Smallanthus maintained in vitro* as autotrophic cultures and as pot plants in the greenhouse. *Canadian Journal of Microbiology* 58: 1202-1211.
- Ryan RP, Germaine K, Franks A, Ryan DJ, Dowling DN (2008) Bacterial endophytes:
 recent developments and applications. *FEMS Microbiol Letters* 278: 19.
- Sadrati N, Daoud H, Zerroug A, Dahamna S, Bouharati S (2013) Screening of antimicrobial and antioxidant secondary metabolites from endophytic fungi isolated from wheat (*Triticum durum*). Journal of Plant Protection Research 53: 127-136.
- Saengruksawong C, Khamyoun S, Anongrak N, Pinthong J (2012) Growths and carbon stocks of para rubber plantations on phonpisai soil series in northeastern Thailand. *Rubber Thai Journal* 1: 1-18.
- Sahashi N (2013) Brown Root Rot Caused by Phellinus noxius in Subtropical Areas of Japan. International symposium on forest health management, Seoul, South Korea. p.102.
- Saikkonen K, Wäli P, Helander M, Faeth SH (2004) Evolution of endophyte-plant symbioses. *Trends in Plant Science* 9: 275–280.
- Sánchez-Ortiz BL, Sánchez-Fernández RE, Duarte G, Lappe-Oliveras P, Macías-Rubalcava ML (2016). Antifungal, anti-oomycete and phytotoxic effects of volatile organic compounds from the endophytic fungus *Xylaria* sp. strain PB3fb isolated from *Haematoxylon brasiletto*. *Journal of Applied Microbiology* 120: 1313-1325.
- Santamaria J, Bayman P (2005) Fungal epiphytes and endophytes of coffee leaves (*Coffea arabica*). *Microbial Ecology* 50: 1-8.

- Sappapan R, Sommit D, Ngamrojanavanich N, Pengpreecha S, Wiyakrutta S, Sriubolmas N, Pudhom K (2008) 11-Hydroxymonocerin from the plant endophytic fungus *Exserohilum rostratum*. *Journal of Natural Products* 71: 1657-1659.
- Saraf M, Pandya U, Thakkar A (2014) Role of allelochemicals in plant growth promoting rhizobacteria for biocontrol of phytopathogens. *Microbiological Research* 169: 18-29.
- Saxena S, Meshram V, Kapoor N (2014) Muscodor darjeelingensis, a new endophytic fungus of Cinnamomum camphora collected from northeastern Himalayas. Sydowia 1: 55-67.
- Saxena S, Meshram V, Kapoor N (2015) *Muscodor tigerii* sp. nov. volatile antibiotic producing endophytic fungus from the northeastern Himalayas. *Annals of Microbiology* 65: 47-57.
- Sharmin T, Rahman MR, Anisuzzaman ASM, Anwar-Ul Islam M (2013) Antimicrobial and cytotoxic activities of secondary metabolites obtained from a novel species of *Streptomyces*. *Bangladesh Pharmaceutical Journal* 16:15-19.
- Silva GH, Teles HL, Trevisan HC, Bolzani VS, Young MCM, Pfenning LH, Eberlin MN, Haddad R, Costa-Neto CM, Araújo AR (2005) New bioactive metabolites produced by *Phomopsis cassiae*, an endophytic fungus in *Cassia spectabilis*. Journal of the Brazilian Chemical Society 16: 1463-1466.
- Singh SK, Strobel GA, Knighton B, Geary B, Sears J, Ezra D. (2011). An endophytic *Phomopsis* sp. possessing bioactivity and fuel potential with its volatile organic compounds. *Microbial Ecology* 61: 729-739.
- Sitara U, Hassan N, Naseem J (2011) Antifungal activity of *Aloe vera* gel against plant pathogenic fungi. *Pakistan Journal of Botany* 43: 2231-2233.

- Sopalun K, Strobel GA, Hess WM, Worapong J (2003) A record of *Muscodor albus*, an endophyte from *Myristica fragrans* in Thailand. *Mycotaxon* 88: 239-247.
- Sparks DL, Page AL, Helmke PA, Loeppert RH, Soltanpour PN, Tabatabai, MA, Johston, CT, Sumner, ME (1996) *Methods of Soil Analysis. Part 3. Chemical Methods.* Soil Science Society of America, Inc., Wisconsin.
- Splivallo R, Fischer U, Gobel C, Feussner I, Karlovsky P (2009) Truffles regulate plant root morphogenesis via the production of auxin and ethylene. *Plant Physiology* 150: 2018-2029.
- Splivallo R, Novero M, Bertea MC, Bossi S, Bonfante P (2007). Truffle volatiles inhibit and induce an oxidative burst in *Arabidopsis thaliana*. *The New Phytologist* 175: 417-424.
- Stierle A, Strobel G, Stierle D (1993) Taxol and taxane production by *Taxomyces andreanae*, an endophytic fungus of Pacific yew. *Science* 260: 214-216.
- Stinson AM, Zidack NK, Strobel GA, Jacobsen BJ (2003a) Mycofumigation with Muscodor albus and Muscodor roseus for control of seedling diseases of sugar beet and Verticillium wilt of eggplant. Plant Disease 87: 1349-1354.
- Stinson M, Ezra D, Hess WM, Sears J, Strobel G (2003b) An endophytic *Gliocladium* sp. of *Eucryphia cordifolia* producing selective volatile antimicrobial compounds. *Plant Science* 165: 913-922.
- Stone JK, Polishook JD, White Jr JF (2004) Endophytic fungi. In *Biodiversity of Fungi:* Inventory and Monitoring Methods. Amsterdam: Elsevier Academic Press.
- Strobel GA (2003) Endophytes as sources of bioactive products. *Microbs and Infect*ion 5: 535–544.
- Strobel G (2006) Harnessing endophytes for industrial microbiology. *Current Opinion in Microbiology* 9: 240-244.

- Strobel G (2011) *Muscodor* species-endophytes with biological promise. *Phytochemistry Reviews* 10: 165-172.
- Strobel G, Daisy B (2003) Bioprospecting for microbial endophytes and their natural products. *Microbiology and Molecular Biology Reviews* 4: 491-502.
- Strobel GA, Dirkse E, Sears J, Markworth C (2001) Volatile antimicrobials from Muscodor albus, a novel endophytic fungus. Microbiology 147: 2943-2950.
- Strobel GA, Kluck K, Hess WM, Sears J, Ezra D, Vargas PN (2007) Muscodor albus E-6, an endophyte of Guazuma ulmifolia making volatile antibiotics: isolation, characterization and experimental establishment in host plant. Microbiology 153: 2613-2620.
- Strobel GA, Singh SK, Riyaz-Ul-Hassan S, Mitchell AM, Geary B, Sears J (2011) An endophytic/pathogenic *Phoma* sp. from creosote bush producing biologically active volatile compounds having fuel potential. *FEMS Microbioly Letters* 320: 87-94.
- Suwannarah N (2013) Biodiversity and secondary metabolites of endophytic fungi isolated from some medicinal plants in Northern Thailand. PhD Thesis, Chiang Mai University, Thailand.
- Suwannarach N, Bussaban B, Hyde KD, Lumyong S (2010) *Muscodor cinnamomi*, a new endophytic species from *Cinnamomum bejolghota*. *Mycotaxon* 114: 15-23.
- Suwannarach N, Bussaban B, Nuangmek W, Pithakpol W, Jirawattanakul B, Matsui K, Lumyong S (2016) Evaluation of *Muscodor suthepensis* strain CMU-Cib462 as a postharvest biofumigant for tangerine fruit rot caused by *Penicillium digitatum. Journal of the Science of Food and Agriculture* 96: 339-345.

- Suwannarach N, Kumla J, Bussaban B, Hyde KD, Matsui K, Lumyong S (2013a) Molecular and morphological evidence support four new species in the genus *Muscodor* from northern Thailand. *Annals of Microbiology* 63: 1341-1351.
- Suwannarach N, Kumla J, Bussaban B, Lumyong S (2012) Biocontrol of *Rhizoctonia* solani AG-2, the causal agent of damping-off by *Muscodor cinnamomi* CMU-Cib 461. World Journal of Microbiology and Biotechnology 28: 3171-3177.
- Suwannarach N, Kumla J, Bussaban B, Nuangmek W, Matsui K, Lumyong S (2013b) Biofumigation with the endophytic fungus *Nodulisporium* spp. CMU-UPE34 to control postharvest decay of citrus fruit. *Crop Protection* 45: 63 -70.
- Tamura K, Stecher G, Peterson D, Filipski A, Kumar S (2013) MEGA6: Molecular evolutionary genetics analysis version 6.0. *Molecular Biology Evolution* 30: 2725-2729.
- Tan RX, Zou WX (2001) Endophytes: a rich source of functional metabolites. *Natural Product Reports* 18: 448-459.
- Tangonan N, Pecho JA, Butardo EGG (2008) Technoguide on Disease of Rubber and Their Management. Department of Agriculture - Bureau of Agricultural Research, Philippines.
- Tejesvi MV, Nalini MS, Mahesh B, Prakash HS, Kini KR, Shetty HS, Subbiah V (2007) New hopes from endophytic fungal secondary metabolites. *Boletín de la Sociedad Química de México* 1: 19-26.
- Theantana T, Hyde KD, Lumyong S (2009) Asparaginase production by endophytic fungi from Thai medicinal plants: cytoxicity properties. *International Journal of Intergrative Biology* 7: 1-8.

- Ting ASY, Mah SW, Tee CS (2010) Identification of volatile metabolites from fungal endophytes with biocontrol potential towards *Fusarium oxysporum* f. sp. cubense Race 4. American Journal of Agricultural and Biological Sciences 5: 177-182.
- Tomsheck AR, Strobel GA, Booth E, Geary B, Spakowicz D, Knighton B, Floerchinger C, Sears J, Liarzi O, Ezra D (2010) *Hypoxylon* sp., an endophyte of *Persea indica*, producing 1, 8-cineole and other bioactive volatiles with fuel potential. *Microbial Ecology* 60: 903-914.
- Usuki F, Narisawa K (2007) A mutualistic symbiosis between a dark, septate endophytic fungus, *Heteroconium chaetospira* and a nonmycorrhizal plant, Chinese cabbage. *Mycologia* 99:175-184.
- van Beilen JB, Poirier Y (2007) Establishment of new crops for the production of natural rubber. *Trends in Biotechnology* 25: 522-529.
- Verma VC, Gond SK, Kumara A, Kharwar RN, Strobel GA (2007) Endophytic mycoflora from leaf, bark, and stem of *Azadirachta indica* A. Juss. from Varanasi, India. *Microbial Ecology* 54: 119-125.
- Viterbo A, Inbar J, Hadar Y, Chet I (2007) Plant disease biocontrol and induced resistance via fungal mycoparasites. In *Environmental and Microbial Relationships*, 2nd Edition The Mycota IV. Berlin Heidelberg: Springer-Verlag.
- von der Schulenburg JH, Hancock JM, Pagnamenta A, Sloggett JJ, Majerus ME, Hurst, GD (2001) Extreme length and length variation in the first ribosomal internal transcribed spacer of ladybird beetles (Coleoptera; Coccinellidae). *Molecular Biology Evolution* 18: 648-660.
- Wang L, Xu B, Wang J, Su Z, Lin F, Zhang C, Kubicek CP (2012) Bioactive metabolites from *Phoma* species, an endophytic fungus from the Chinese medicinal plant *Arisaema erubescens*. *Applied Microbiology and Biotechnology* 93: 1231-1239.

- Waqas M, Khan AL, Kamran M, Hamayun M, Kang SM, Kim YH, Lee IJ (2012) Endophytic fungi produce gibberellins and indole acetic acid and promotes host-plant growth during stress. *Molecules* 17: 10754-10773.
- Wiyakrutta S, Sriubolmas N, Thongon N, Danwisetkanjana K, Ruangrungsi N, Meevootisom V (2004) Endophytic fungi with antimicrobial, anticancer and antimalarial activities isolated from Thai medicinal plants. World Journal of Microbiology and Biotechnology 20: 265-272.
- Worapong J, Strobel GA (2009) Biocontrol of a root rot of kale by *Muscodor albus* strain MFC2. *BioControl* 54: 301-306.
- Worapong J, Strobel GA, Daisy B, Castillo UF, Baird G, Hess WM (2002) Muscodor roseus anam. sp. nov., an endophyte from Grevillea pteridifolia. Mycotaxon 81: 463-475.
- Worapong J, Strobel GA, Ford EJ, Li JY, Baird G, Hess WM (2001) Muscodor albus anam. nov., an endophyte from Cinnamomum zeylanicum. Mycotaxon 79: 67-79.
- Xiong Z, Yang Y, Zhao N, Wang Y (2013) Diversity of endophytic fungi and screening of fungal paclitaxel producer from Anglojap yew, *Taxus x media*. BMC Microbiology 13: 71-80.
- Yadav R, Singh AV, Joshi S, Kumar M (2015) Antifungal and enzyme activity of endophytic fungi isolated from Ocimum sanctum and Aloe vera. African Journal of Microbiology Research 9: 1783-1788.
- Yahaya R, Dash GK, Abdullah MS, Mathews A (2015) Clinacanthus nutans (burm. F.) Lindau: an useful medicinal plant of south-east asia. *International Journal of Phamacognosy and Phytochemical Research* 7:1244-1250.
- Yang X, Strobel GA, Stierle A, Hess WM, Lee J, Clardy J (1994) A fungal endophytes tree relationship: *Phoma* sp. in *Taxus wallachiana*. *Plant Science* 102: 1-9.

- Zhang CL, Wang GP, Mao LJ, Komon-Zelazowska M, Yuan ZL, Lin FC, Druzhinina IS, Kubicek CP (2010) *Muscodor fengyangensis* sp. nov. from southeast China: morphology, physiology and production of volatile compounds. *Fungal Biology* 114: 797-808.
- Zhang G, Sun S, Zhu T, Lin Z, Gu J, Li D, Gu D (2011) Antiviral isoindolone derivatives from an endophytic fungus *Emericella* sp. associated with *Aegiceras corniculatum*. *Phytochemistry* 72: 1436-1442.
- Zhang G, Zhang Y, Qin J, Qu X, Liu J, Li X, Pan H (2013) Antifungal metabolites produced by *Chaetomium globosum* no.04, an endophytic fungus isolated from *Ginkgo biloba*. *Indian Journal of Microbiology* 53: 175-180.
- Zhi-Lin Y, Yi-Cun C, Bai-Ge X, Chu-Long Z (2012) Current perspective on the volatileproducing fungal endophytes. *Critritical Reviews in Biotechnology* 32: 363-373.
- Zhou Z, Zhang C, Zhou W, Li W, Chu L, Yan J, Li H (2013) Diversity and plant growthpromoting ability of endophytic fungi from the five flower plant species collected from Yunnan, Southwest Chaina. *Journal of Plant Interactions* 9: 585-591.

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

LIST OF PUBLICATIONS

 Siri-Udom S, Suwannarach N, Lumyong S (2015) Existence of *Muscodor* vitigenus, M. equiseti and M. heveae sp. nov. in leaves of the rubber tree (*Hevea* brasiliensis Müll.Arg.), and their biocontrol potential. Annals of Microbiology 66: 437-448.



ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

APPENDIX A

Media

Various media were used in this study as isolation medium, storage medium or test medium and prepared as followed formula (per liter), dispensed into containers and sterilized by autoclaving at 121°C for 15 min. 2/2

1. Corn Meal Agar (CM	IA) (Himedia Laboratories, India)
Corn meal	50 g
Agar	15 g, pH 6.0
2. Malt extract agar (M	(A)
Malt extract	30 g
Agar	15 g, pH 5.6
3. 2% Malt extract aga	
Malt extract	20 g
Agar	15 g, pH 5.5
4. Nutrient agar (NA)	
Beef extract	ธิมหาวูติทยาลัยเชียงไหม
Peptone Copyrig	ht [©] bys Chiang Mai University
Agar A	righ ¹⁵ g, pH7 reserved
5. Potato dextrose agar	(PDA)

200 g

20 g

15 g. pH 5.6

147

Potatoes

Glucose

Agar

11

100

6. Potato dextrose broth (PDB)

Potatoes	200 g
Glucose	20 g, pH 6.0

7. Water agar (WA)

Agar

15 g, pH 7.0

8. Yeast extract-malt extract agar (YMA)

Peptone

Yeast extract

Malt extract

Dextrose

Agar



Copyright[©] by Chiang Mai University All rights reserved

APPENDIX B

Standard curve of Indole acetic acid (IAA) preparation

1. Reagent

Salkowski's reagent (1 ml of 0.5 M FeCl₃; 50 ml of 35% HClO₄).

2. Method

Mixed IAA standard (1 ml) with Salkowski's reagent and incubated at room temperature $(25\pm2 \ ^{\circ}C)$ in the dark for 30 min. The appearance of a pink color indicated IAA production and was measured at OD 530 nm, uninoculated media mixed with reagent used as a blank.

งหยนุติ

IAA concentration (µg/ml)	OD 530 nm
0	
10	0.036
20 20	0.048
ลิขสิ ³⁰ ธิ์มหาวิทย	0.086
Copy ⁴⁰ ht [©] by Chia	ng Mai 0.116 ersity
All 50 rights	res (0.150 v e d
100	0.316
150	0.498
200	0.068
250	0.900



APPENDIX C

Extraction of crude IAA and Thin layer chromatography (TLC)

Method

The crude extracts of endophytic fungi were dissolved in 1 ml of methanol and spotted on TLC sheets (TLC aluminium sheets, silica gel 60GF254, Merck, Germany). Then, developed in ethyl acetate:chloroform:formic acid (55:35:10, v/v/v). Spots with Rf values identical to authentic IAA were identified under UV light at wave length of 254 nm by spraying with Ehmann's reagent and Salkowski's reagent (Ehmann, 1977).

 Table C Rf values of crude IAA after developed by TLC

No.	Sample	Rf	NO84
	121 -	Solawaski's reagent	Ehmann's reagent
	131	MANS	2/
	IAA	0.68	0.73
	IBA	0.70	0.77
		UNIVE	
4	C4IV301	0.58, 0.61, 0.74	0.64, 0.69, 0.73
580	C4V202-1	0.52, 0.61	0.63, 0.67, 0.73
₆ Co	RTM5IV1	0.52, 0.61	niversity
8 A	PS1IV102	t s _{0.57, 0.75} s e	0.61, 0.67, 0.75
14	BS1B201	0.57, 0.73	0.67, 0.78

(-) = no data



Salkowski's reagent Ehmann's reagent

Figure C Thin layer chromatography (TLC) of IAA crude extract such as IAA standard, IBA standard, no. 4 = C4IV301, 5 = C4V202-1, 6 = RTM5I-V1, 8 = PS1IV102, 14 = BS1B201 and PDB (negative control).



APPENDIX D

Fungal preparation for scanning electron microscopy

1. Agar plug of 7-day-old fungal endophytes were placed into 2% glutaraldehyde in 0.1 M sodium carcodylate buffer (pH 7.2-7.4) with TritonX and left overnight.

HI MO

2. All agar plugs were moved into dehydrating solutions. They were washed in seven 15 min changes of water-buffer, followed by a 15 min change in 5%, 10%, 15%, 20%, 40%, 50% and 70% ethanol. They were left overnight or longer in 70% ethanol.

3. They were rinsed for 15 min in 80%, 95% and 100% ethanol, followed by nine 15 min changes in acetone, followed by a 15 min change in 10%, 15%, 20%, 40%, 50%, 70%, 80%, 95% and 100% acetone.

The dehydration process was done slowly. For scanning electron microscopy the fungal material was critical-point dried, gold spotter-coated and images were recorded with a scanning electron microscope (JEOL JSM-5910LV, Japan).

IINT

ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved
APPENDIX E

Nucleotide sequence and GenBank accession number of ITS

rRNA gene of *Muscodor vitigenus* (RTM5IV1)

LOCUS DEFINITION	KR076793 579 bp DNA linear PLN 23-SEP-2015 Rigidoporus microporus strain SDBR-CMU R1 internal Transcribed spacer 1, partial sequence; 5.8S ribosomal RNA gene, complete sequence; and internal transcribed spacer 2, partial sequence.
ACCESSION VERSION KEYWORDS	KR076793 KR076793.1
SOURCE ORGANISM	Rigidoporus microporus (Fomes lignosus) Rigidoporus microporus
REFERENCE	Eukaryota; Fungi; Dikarya; Basidiomycota; Agaricomycotina; Agaricomycetes; Polyporales; Rigidoporus.
AUTHORS TITLE	Nakaew,N., Rangjaroen,C. and Sungthong,R. Utilization of rhizosperic Streptomyces for biological control of <i>Rigidoporus</i> sp. causing white root disease in rubber tree
JOURNAL REFERENCE	Eur. J. Plant Pathol. 142, 93-105 (2015) 2 (bases 1 to 579)
AUTHORS TITLE	Siri-Udom,S. Direct Submission
JOURNAL Chiangmai	Submitted (07-APR-2015) Biology, Faculty of Science,
	University, 239 Huay Kaew Road, Muang, Chiangmai 50200, Thailand
COMMENT	##Assembly-Data-START## Sequencing Technology :: Sanger dideoxy sequencing
223	##Assembly-Data-END##
source	Location/Qualifiers 1579
	/organism="Rigidoporus microporus" /mol_type="genomic DNA"
	/strain="SDBR-CMU R1" /isolation_source="infected roots of diseased tree"
	<pre>/host="rubber tree; cultivar: RRIM600; age: 8-10 months"</pre>
	/country="Thailand: Wangthong District, Phitsanulok"
misc_R	/lat_lon=" <u>16.8242 N 100.4286 E</u> " NA <1>579
	/note="contains internal transcribed spacer 1, 5.8S

ribosomal RNA, and internal transcribed spacer 2" ORIGIN 1 cattaacgaa ttgcgttcgg ggttgttgct ggttttcttt ttaacaggag agaacatgtg 61 cacgcctcgc aatccatttc aaaccacact tgtgcacttc agaggggggg cctctcttgg 121 cctctccttc tttcatcact acaaaccact ttaaagtctt ttgtatttgt tggttaacta 181 taatgttaaa tacaactttc aacaacggat ctcttggctc tcgcatcgat gaagaacgca 241 gcgaaatgcg ataagtaatg tgaattgcag aattcagtga atcatcgaat ctttgaacgc 301 accttgcgct ccttggtatt ccgaggagca tgcctgtttg agtgtcgtgt aattctcaat 361 ctcaactttt tttgttgtgg attggatttg ggagcttgtc gtgtctcttt ctataatgaa 421 agaggttaga ctctccttga atgcattagc tcggtcacgt agtttgcccg acggttcacg 481 gtgtgatagt ctcacttcat cgccgttcta actgttggtg cctgtgtttt tqccqqcttc 541 taatctctqq cctctttttc aaaqtqqcct ttacacttt



ลิขสิทธิ์มหาวิทยาลัยเชียงใหม่ Copyright[©] by Chiang Mai University All rights reserved

APPENDIX F

Nucleotide sequence and GenBank accession number of ITS1 rRNA gene of *Muscodor equiseti* (RTM5IV2)

```
KF850711 611 bp DNA linear PLN 17-MAY-2016
LOCUS
           Muscodor sp. RTM5-IV2 18S ribosomal RNA gene, partial
DEFINITION
           sequence; internal transcribed spacer 1, 5.8S ribosomal RNA
           gene, and internal transcribed spacer 2, complete sequence;
           and 28S ribosomal
           RNA gene, partial sequence.
ACCESSION
           KF850711
VERSION
           KF850711.1
KEYWORDS
SOURCE
           Muscodor sp. RTM5-IV2
  ORGANISM Muscodor sp. RTM5-IV2
           Eukaryota; Fungi; Dikarya; Ascomycota; Pezizomycotina;
           Sordariomycetes; Xylariomycetidae; Xylariales; mitosporic
           Xylariales; Muscodor.
           1 (bases 1 to 611)
REFERENCE
           Siri-udom, S., Suwannarach, N. and Lumyong, S.
 AUTHORS
           Existence of Muscodor vitigenus, M. equiseti and M. heveae
 TITLE
           sp. nov. in leaves of the rubber tree (Hevea brasiliensis
           Müll.Arg.), and their biocontrol potential
 JOURNAL
           Ann. Microbiol. 66 (1), 437-448 (2016)
REFERENCE
           2 (bases 1 to 611)
 AUTHORS
          Siri-Udom, S. and Lumyong, S.
 TITLE
           Muscodor, an endophytic fungus of Hevea brasiliensis (Para
           Rubber tree) from northeast of Thailand with volatile
           Antibiotic activities
 JOURNAL
          Unpublished
           3 (bases 1 to 611)
REFERENCE
 AUTHORS Siri-Udom, S. and Lumyong, S.
 TITLE Direct Submission
  JOURNAL
           Submitted (08-NOV-2013) Biology, Faculty of Science,
           ChiangMai University, 239 Huay Kaew Road, Muang, ChiangMai
           50200, Thailand
FEATURES
                    Location/Qualifiers
                    1..611
  source
                     /organism="Muscodor sp. RTM5-IV2"
                     /mol_type="genomic DNA"
                     /isolate="RTM5-IV2"
                     /isolation_source="healthy leaves"
                     /host="Hevea brasiliensis"
                     /db_xref="taxon:1457325"
                     /country="Thailand"
                     /lat lon="17.2039 N 102.4406 E"
                     /collection date="May-2011/Dec-2012"
                     /collected by="Siri-Udom, S"
```

/note="PCR_primers=fwd_name: its5, rev_name: its4" misc RNA <1..>611 /note="contains 18S ribosomal RNA, internal Transcribed spacer 1, 5.8S ribosomal RNA, internal transcribed spacer 2, and 28S ribosomal RNA" ORIGIN 1 gtcgtaacaa ggtctccgtt ggtgaaccag cggagggatc attacagagt tttctaaact 61 cccaacccta tgtgaactta cctttgttgc ttcggcggcg gaggctaccc tgcgggagaa 121 taccacttag tggttaccct gtagtttcag gtacatcagc tacccggtag tcatccccgc 181 cggcggccaa ctaaactctg ttttctttgg aattctgaat aacaaactta ataagttaaa 241 actttcaaca acggatctct tggttctggc atcgatgaag aacgcagcga aatqcqataa 301 gtaatgtgaa ttgcagaatt cagtgaatca tcgaatcttt gaacgcacat tgcgcccatt 361 agcattetag tgggcatgee tgttegageg teattteace acttaageet tgttgcttag 421 cgttgggagc ctacggcacg gcccgtagct ccttaaagtg attggcggag ttqqttctca 481 ctctaagcgt agtaattata tctcgcttct gtagtggtcc cggcccctgc cotaaaaccc 541 cttatacaaa ggttgacctc ggatcaggta ggaatacccg ctgaacttaa gcatatcaat 601 aagccggagg a

ออกส์บหาวิทยาลัยเชียงให Copyright[©] by Chiang Mai Universit

APPENDIX G

Nucleotide sequence and GenBank accession number of ITS1 rRNA gene of *Muscodor heveae* (RTM5IV3)

LOCUS	KF850712 593 bp DNA linear PLN 17-MAY-2016	
DEFINITION	Muscodor sp. RTM5-IV3 internal transcribed spacer 1,	
	Partial sequence; 5.8S ribosomal RNA gene, complete	
	sequence; and internaltranscribed spacer 2, partial	
	sequence.	
ACCESSION	KF850712	
VERSION	KF850712.1	
KEYWORDS	the the state	
SOURCE	Muscodor sp. RTM5-IV3	
ORGANISM	Muscodor sp. RTM5-IV3	
	Eukaryota; Fungi; Dikarya; Ascomycota; Pezizomycotina;	
	Sordariomycetes; Xylariomycetidae; Xylariales; mitosporic	
	Xylariales; Muscodor.	
REFERENCE	1(bases 1 to 593)	
AUTHORS	Siri-udom, S., Suwannarach, N. and Lumyong, S.	
TITLE	Existence of Muscodor vitigenus, M. equiseti and M. heveae	
	sp. nov.in leaves of the rubber tree (Hevea brasiliensis	
	Müll.Arg.), and their biocontrol potential	
JOURNAL	Ann. Microbiol. 66 (1), 437-448 (2016)	
REFERENCE	2(bases 1 to 593)	
AUTHORS	Siri-Udom, S. and Lumyong, S.	
TITLE	Muscodor, an endophytic fungus of Hevea brasiliensis (Para	
	Rubber tree) from northeast of Thailand with volatile	
	Antibiotic activities	
JOURNAL	Unpublished	
REFERENCE	3 bases 1 to 593)	
AUTHORS	Siri-Udom,S. and Lumyong,S.	
TITLE	Direct Submission	
JOURNAL	Submitted (08-NOV-2013) Biology, Faculty of Science,	
	Chiang Mai University, 239 Huay Kaew Road, Muang, Chiang	
	Mai 50200, Thailand	
FEATURES	Location/Qualifiers	

```
/organism="Muscodor sp. RTM5-IV3"
/mol_type="genomic DNA"
/isolate="RTM5-IV3"
/isolation_source="healthy leaves"
/host="Hevea brasiliensis"
/db_xref="taxon:<u>1457326</u>"
/country="Thailand"
/lat lon="17.2039 N 102.4406 E"
```

1..593

/collection_date="May-2011/Dec-2012"

/collected_by="Siri-Udom, S"

/note="PCR_primers=fwd_name: its5, rev_name: its4"

misc RNA

source

<1..>593

/note="contains internal transcribed spacer 1, 5.8S ribosomal RNA, and internal transcribed spacer 2"

ORIGIN

1 gcagaggaca ggttacagag ttttctaaac tcccaacccc atgtgaaatt acctttggtt

61 cccaaccggc ggcggaggct acccctgtgg gggataccac cggaggggtt accccggaag

121 ccccgagggc cataaatggc ttgccccgac atcttatccc cttacgacta gctacccggt

181 gggcctcccc tgcagggggc caaataaaat ctgttttat tggaattctg aattataaaa

241 ttaataagtt aaaactttca aaaagggatc tcttgattct ggcatcgatg aagaaggcaa

301 cgaaatggct aaaactaatg agaattgcaa caattcagtg gaaccatcaa atctttgaac

361 gcacattgcg cccattagca ttttcactgg gcatgcctgt tgctaccgtc atttcaccac

421 ttaagccctg ttgcttagcg ccggaaacct acggcactgg cccggttctc ccttaaagtg

481 attggaacaa ttggttctca ctctaggggt accaaatcca tactctgcct ccccctggta

541 ggtttccagg cccctgccct aaaaccccct atatataaag gttgacctcg gat

APPENDIX H

Nucleotide sequence and GenBank accession number of ITS1 rRNA gene of *Rigidoporus microporus* strain SDBR-CMU R1

LOCUS	KR076793 579 bp DNA linear PLN 23-SEP-2015
DEFINITION	Rigidoporus microporus strain SDBR-CMU RI internal transcribed
	spacer 1, partial sequence; 5.8S ribosomal RNA gene,
	complete
	sequence.
ACCESSION	KR076793
VERSION	KR076793.1
KEYWORDS	·
ORGANISM	Rigidoporus microporus
	Eukaryota; Fungi; Dikarya; Basidiomycota; Agaricomycotina;
	Agaricomycetes; Polyporales; Rigidoporus.
REFERENCE	1 (bases 1 to 579)
AUTHORS TITLE	Nakaew, N., Rangjaroen, C. and Sungthong, R.
	Utilization of rhizosperic Streptomyces for biological
	rubber tree
JOURNAL	Eur. J. Plant Pathol. 142, 93-105 (2015)
REFERENCE	2 (bases 1 to 579)
AUTHORS	Siri-Udom,S.
TITLE JOURNAL	Direct Submission
	Submitted (U/-APR-2015) Biology, Faculty of Science, Chiangmai University 239 Huay Kaew Road Muang Chiangmai
	50200, Thailand
COMMENT	##Assembly-Data-START##
	Sequencing Technology :: Sanger dideoxy sequencing
CIUCI	##Assembly-Data-END##
FEATURES	Location/Qualifiers
SOULCE	/organism="Rigidoporus microporus"
	/mol_type="genomic DNA"
	/strain="SDBR-CMU R1"
	/isolation_source="infected roots of diseased
	tree"
	months"
	/db xref="taxon:219653"
	/country="Thailand: Wangthong District,
	Phitsanulok"
	/lat_lon=" <u>16.8242 N 100.4286 E</u> "
misc_R	NA <1>5/9 /note="contains internal transcribed spacer 1
	, mote- contrating interinat transcribed spacer 1,

5.8S ribosomal RNA, and internal transcribed spacer 2"

ORIGIN 1 cattaacgaa ttgcgttcgg ggttgttgct ggttttcttt ttaacaggag agaacatgtg 61 cacgcctcgc aatccatttc aaaccacact tgtgcacttc agagggggag cctctcttgg 121 cctctccttc tttcatcact acaaaccact ttaaagtctt ttgtatttgt tggttaacta 181 taatgttaaa tacaactttc aacaacggat ctcttggctc tcgcatcgat gaagaacgca 241 gcgaaatgcg ataagtaatg tgaattgcag aattcagtga atcatcgaat ctttgaacgc 301 accttgcgct ccttggtatt ccgaggagca tgcctgtttg agtgtcgtgt aattctcaat 361 ctcaactttt tttgttgtgg attggatttg ggagcttgtc gtgtctcttt ctataatgaa 421 agaggttaga ctctccttga atgcattagc tcggtcacgt agtttgcccg acggttcacg 481 gtgtgatagt ctcacttcat cgccgttcta actgttggtg cctgtgtttt tqccqqcttc 541 taatctctgg cctctttttc aaagtggcct ttacacttt



ลิขสิทธิ์มหาวิทยาลัยเชียงใหม Copyright[©] by Chiang Mai University All rights reserved

CURRICULUM VITAE

Author's name	Ms.Sakuntala Siri-Udom
Date/Year of Birth	14 June 1978
Place of Birth	Nonsang District, Nongbualamphu province
Education	 1977 Bachelor's Degree of Science, Major Microbiology, Chiang Mai University 2001 Master's Degree of Science, Major Biology, Chiang Mai University 2011 Doctor of Philosophy, Major Applied Microbiology, Chiang Mai University
Scholarship	 2012 The Graduate School of Chiang Mai University, The Higher Education Commission, Thailand, under the National Research University (A1) program, Chiang Mai University 2013 Thailand Research Fund (RTA5580007)
Publication	Siri-Udom S, Suwannarach N, Lumyong S (2015) Existence of <i>Muscodor vitigenus</i> , <i>M. equiseti</i> and <i>M. heveae</i> sp. nov. in leaves of the rubber tree (<i>Hevea brasiliensis</i> Müll.Arg.), and their biocontrol potential. <i>Annals of Microbiology</i> 66: 437-448.
A	rights reserved

